CWI Tracts

Managing Editors

J.W. de Bakker (CWI, Amsterdam) M. Hazewinkel (CWI, Amsterdam) J.K. Lenstra (CWI, Amsterdam)

Editorial Board

W. Albers (Maastricht)
P.C. Baayen (Amsterdam)
R.T. Boute (Nijmegen)
E.M. de Jager (Amsterdam)
M.A. Kaashoek (Amsterdam)
M.S. Keane (Delft)
J.P.C. Kleijnen (Tilburg)
H. Kwakernaak (Enschede)
J. van Leeuwen (Utrecht)
P.W.H. Lemmens (Utrecht)
M. van der Put (Groningen)
M. Rem (Eindhoven)
A.H.G. Rinnooy Kan (Rotterdam)
M.N. Spijker (Leiden)

Centrum voor Wiskunde en Informatica

Centre for Mathematics and Computer Science P.O. Box 4079, 1009 AB Amsterdam, The Netherlands

The CWI is a research institute of the Stichting Mathematisch Centrum, which was founded on February 11, 1946, as a nonprofit institution aiming at the promotion of mathematics, computer science, and their applications. It is sponsored by the Dutch Government through the Netherlands Organization for the Advancement of Pure Research (Z.W.O.).

الدوم روم الدين المرومة رومة ومحتورة مروم معرومة مروم ومر<mark>م ر</mark>و

CWI Tract

Homogeneous zero-dimensional absolute Borel sets

A.J.M. van Engelen



Centrum voor Wiskunde en Informatica Centre for Mathematics and Computer Science

1980 Mathematics Subject Classification: Primary: 54H05. Secondary: 03E15. ISBN 90 6196 303 6

ı

Copyright $\ensuremath{^\odot}$ 1986, Mathematisch Centrum, Amsterdam Printed in the Netherlands

PREFACE

This book was written while I was holding a position at the Mathematical Institute of the University of Amsterdam. My main debt is to my supervisors, J. van Mill and A.B. Paalman-de Miranda, for their contribution to my work, their constant encouragement, and their careful reading of the manuscript. Thanks are also due to A.W. Miller, who pointed my attention to a paper of J. Steel that led me to the Wadge hierarchy and the results of Chapters 4 and 5 of this book; and to A. Louveau, for some helpful comments. I thank the Centre for Mathematics and Computer Science for the opportunity to publish this book in their series CWI Tracts.

A.J.M. van Engelen

CONTENTS

Introduction	1
<u>Chapter 1</u> : Preliminaries	5
Chapter 2: Borel sets of class 1	11
2.1 The multiplicative class 0	11
2.2 The ambiguous class l	14
2.3 The multiplicative class 1	15
2.4 The additive class l	17
Chapter 3: Borel sets of ambiguous class 2	22
3.1 Small Borel classes	23
3.2 Knaster-Reichbach coverings and their applications	29
3.3 Hurewicz-type theorems	37
3.4 The small Borel classes $\mathbb{D}_{n}(\Sigma^{0}_{2})$ for $\mathfrak{n} \in N$	44
3.5 The small Borel classes $D_{\alpha}(\Sigma_2^{0})$ for $\alpha \in [\omega, \omega_1)$	57
3.6 The main theorem, part 1	69
Chapter 4: Borel sets of higher class	73
4.1 The Wadge hierarchy of Borel sets	74
4.2 An inductive description of the Borel Wadge classes	81
4.3 Existence of homogeneous Borel sets	93
4.4 The main theorem, part 2	97
4.5 Some applications	102
4.6 The ambiguous class 2 revisited	105

Chapter 5: Rigid Borel sets	110
<u>Appendix</u> : The countable infinite product of rationals A.1 The first characterization of Q^{ω} A.2 The second characterization of Q^{ω}	113 113 121
References	127
Special symbols	131
Author index	132
Subject index	133

iii

INTRODUCTION

All spaces under discussion are separable and metrizable.

This monograph is an investigation into the internal topological structure of zero-dimensional absolute Borel sets. The problem of finding topological characterizations of zero-dimensional absolute Borel sets was first stated explicitly in 1928, by Alexandroff and Urysohn [1]. Before this, Brouwer [4] and Sierpiński [50] had obtained the following characterizations of, respectively, the Cantor set C, and the space of rational numbers Q:

1

- up to homeomorphism, C is the only (non-empty) zero-dimensional compact space without isolated points;
- up to homeomorphism, Q is the only (non-empty) countable space without isolated points.

Alexandroff and Urysohn added to these: characterizations of $C \setminus \{p\}$ (i.e. the Cantor set minus one point), the space of irrationals P, and the product of the rationals with the Cantor set $Q \times C$:

- up to homeomorphism, C \{p} is the only zero-dimensional locally compact, non-compact space without isolated points;
- up to homeomorphism, *P* is the only (non-empty) zero-dimensional topologically complete space which is nowhere compact;
- up to homeomorphism, $Q \times C$ is the only (non-empty) zero-dimensional σ compact space which is nowhere compact and nowhere countable.

These three spaces, together with the Cantor set, were called *irreducible* Borel sets by Alexandroff and Urysohn. More precisely, they defined a zerodimensional space to be irreducible if it is of one of the following types:

l. compact, dense in itself;

2. locally compact, non-compact, dense in itself;

3. σ -compact, nowhere compact, nowhere countable;

4. topologically complete, nowhere compact.

Thus, from the above characterizations they concluded that of each of the types I through 4 there is exactly one space, up to homeomorphism, and that this space is homogeneous. Although Alexandroff and Urysohn also defined irreducibility for arbitrary Borel sets, this notion seems too general to allow characterizations of all such zero-dimensional spaces; in particular, there are many *non-homogeneous* spaces which are irreducible in that sense.

Let us now have a look at the above results against the framework of the Borel hierarchy in the Cantor set. The above spaces are all Borel sets of class 1, i.e. they are σ -compact or complete (\equiv topologically complete). And from the characterizations it is easily deduced that, apart from the discrete spaces, they are the *only* homogeneous zero-dimensional absolute Borel sets of class 1; this can be read from the following diagram, bearing in mind that a homogeneous space which is not locally compact (not complete, not σ -compact) is also nowhere compact (nowhere complete, nowhere σ -compact):

compact no isolated points: *C*



In Chapter 2 of this monograph, proofs of all these old characterizations will be given.

We now turn our attention to zero-dimensional homogeneous absolute Borel sets which are of ambiguous class 2, but not of class 1; in other words, we consider homogeneous spaces which are both an $F_{\sigma\delta}$ and a $G_{\delta\sigma}$ in the Cantor set, but which are neither σ -compact nor complete. Of these homogeneous Borel sets of exactly ambiguous class 2, the first to be characterized was $Q \times P$, by van Mill [38], 53 years (!) after Alexandroff and Urysohn's paper. If we embed $Q \times P$ densely in C, then its complement will be a nowhere complete, nowhere σ -compact Borel set of ambiguous class 2; there are exactly two such complements that are homogeneous, and one of them, denoted by S, was also characterized by van Mill [39], the other, denoted by T, was characterized by van Douwen [7].

In this monograph, we continue the investigation into the structure of the zero-dimensional absolute Borel sets of ambiguous class 2. In Chapter 3, we will prove that there are precisely ω_1 homogeneous elements in this class, and we will obtain internal, topological characterizations of each of them. To describe these spaces, we will define topological properties that are closely related to the so-called "small Borel classes" (see Kuratowski [28] \$33.IV).

Then, in Chapter 4, we consider Borel subsets of the Cantor set that are not of ambiguous class 2; we will find characterizations of all homogeneous spaces at these levels, thus obtaining a complete picture of the class of homogeneous zero-dimensional absolute Borel sets. Whereas the properties describing the spaces at the lower levels are rather perspicuous, those needed in this chapter are in general not so easy to grasp. The characterizations we give involve the concept of the Wadge class of a space, a concept which is related to game theory; we heavily rely on results of Steel [56] and Louveau [32].

A somewhat unexpected corollary to our classification of the *homogeneous* Borel sets in the Cantor set is, that non-trivial *rigid* Borel sets in Cdo not exist; this answers a question of van Douwen [8]. The proof is given in Chapter 5.

Remark: Originally, my research was aimed at finding an internal, topological characterization of the countable infinite product of rationals, Q^{ω} . After I had succeeded in this, I learned of Steel's results which could easily be used to prove my theorem; my subsequent investigation into the theory of Wadge classes led to the results of Chapters 4 and 5. In an appendix to

this monograph, I have nevertheless included my proof of the characterization of Q^{ω} , in the first place because it is purely topological and does not involve the deep results from game theory used in Chapter 4, and in the second place since the techniques used seem to be interesting in their own right (see e.g. van Mill [41]).

CHAPTER 1: PRELIMINARIES

In this chapter, we establish some notational conventions, and we state various results that will be used throughout this monograph, often without further reference.

1.1 As far as standard notions from general topology are concerned, we mostly follow Engelking [15].

1.2 A \approx B means that A and B are homeomorphic; h: A \approx B means that h: A \rightarrow B is a homeomorphism.

1.3 An ordinal number is the set of all smaller ordinal numbers. If α is an ordinal, and $\alpha = \beta+1$, then α is a successor ordinal, and we define $\alpha-1 = \beta$. If $\alpha \neq 0$ and α is not a successor, then α is a limit ordinal; we write $\lim(\alpha)$. The first infinite ordinal is denoted by ω , $N = \omega \setminus 1 = \omega \setminus \{0\}$, and ω_1 is the first uncountable ordinal.

If α is an ordinal, then α can be represented as $\gamma+n$, where $n < \omega$, and $\gamma = 0$ or γ is a limit; α is said to be *even*, resp. *odd*, if n is even, resp. odd.

The cofinality of a limit ordinal α is the least ordinal γ such that $\alpha = \sup\{\alpha_{\beta}: \beta < \gamma\}$ for certain ordinals $\alpha_{\beta} < \alpha$.

The cardinality of a set A is denoted by |A|; we identify cardinal numbers with *initial* ordinals, where α is initial if $|\beta| < |\alpha|$ for all ordinals $\beta < \alpha$.

1.4 The countable infinite product of a space X is denoted by X^{ω} ; for n < ω , $\pi_n: X^{\omega} \to X$ is the projection onto the nth coordinate (starting with

the 0th coordinate), and $\pi_{\leq n}: X^{\omega} \to X^{n+1}$ is the projection onto the first n+1 coordinates. Usually, X will be ω or $2 = \{0,1\}$. By *M*, we denote the set $\omega^{\leq \omega}$ of all finite sequences of elements from ω , including the empty sequence \emptyset (i.e. $M = \{\pi_{\leq n}(\sigma): \sigma \in \omega^{\omega}, n < \omega\} \cup \{\emptyset\}$). Fix $s = (i_0, \dots, i_{k-1}) \in M$, $k \in N$; then -|s| = k, $|\emptyset| = 0$; $-v(s) = i_0 + i_1 + \dots + i_{k-1} + |s|, v(\emptyset) = 0$; $-s^i = (i_0, \dots, i_{k-1}, i), \ \emptyset^i = (i)$; $-f(s) = i_{k-1}$; $-s|\ell = (i_0, \dots, i_{\ell-1})$ if $1 \leq \ell \leq k$, $s|0 = \emptyset$; $-\hat{s} = s|k-1$ (thus, $s = \hat{s}^f(s)$). Furthermore, if $s, t \in M$, then we write $s \leq t$ (resp. s < t) if s = t|kfor some $k \leq |t|$ (resp. k < |t|). Finally, if $s \in M$, and $\sigma = (i_n)_{n \leq \omega} \in \omega^{\omega}$, then for $k < \omega$, $\sigma|k = \pi_{\leq k-1}(\sigma)$ if k > 0, $\sigma|0 = \emptyset$; and $s < \sigma$ if $s = \sigma|n$ for some $n < \omega$.

1.5 All spaces in this monograph are separable and metrizable. Metrics are denoted by d, and assumed to be bounded by 1; almost invariably, the choice of metric is irrelevant, otherwise it is obvious. If f,g: $X \rightarrow Y$ are continuous fuctions, and X is compact, then d(f,g) = $\sup\{d(f(x),g(x)): x \in X\}$.

1.6 A subset of a space X is *clopen* if it is both open and closed in X; thus, a *zero-dimensional* space is a space with a (countable) basis consisting of clopen sets.

1.7 If Y is a subset of a space X, then U is a covering of Y if UU = Y. U is said to be open (closed, clopen) if it consists of open (closed, clopen) subsets of Y; if the elements of U are actually open (closed, clopen) in X, then we say that U is a covering of Y by open (closed, clopen) subsets of X.

U is said to be *disjoint* if it consists of pairwise disjoint sets; if we say that $U = \{U_i: i \in I\}$ is disjoint, then this includes the statement that $U_i \neq U_i$ if $i \neq j$.

1.8 Let P be a topological property. A space X is nowhere P if no nonempty open subset of X is P, and if in addition <u>X is non-empty</u>. We also exclude \emptyset from the spaces said to be 'dense in itself' or 'without isola-

ted points' (hence, those are the spaces which are 'nowhere of cardinality 1').

1.8.1 Let P be closed-hereditary, let X be non-empty and zero-dimensional, and let B be a basis for X consisting of clopen sets. Then X is nowhere P if and only if no non-empty $B \in B$ is P.

Proof: "Only if" is trivial. Conversely, if U is a non-empty open subset of X, then U contains a non-empty $B \in B$; since B is closed in U, and P is closed-hereditary, U is not P.

If Q is also a topological property, then we say that "X is $P \cup Q$ " if X can be written as the union of two subspaces, one of which is P, the other Q.

1.9 A space X is *rigid* if the identity is the only autohomeomorphism of X. X is *homogeneous* if for each x, y \in X, there exists a homeomorphism h: X \Rightarrow X such that h(x) = y.

X is homogeneous with respect to dense copies of A if for all dense subspaces A_0, A_1 of X such that $A_0 \approx A \approx A_1$, there exists a homeomorphism h: X \Rightarrow X such that $h[A_0] = A_1$.

X is strongly homogeneous if $U\approx X$ for each non-empty clopen subspace U of X.

Of course, the notion of strong homogeneity is only of interest in the realm of zero-dimensional spaces.

1.9.1 A strongly homogeneous zero-dimensional space X is homogeneous.

Proof: Let $x, y \in X$. If $\{x\}$ or $\{y\}$ is open, then |X| = 1, and we are done. Otherwise, there exist strictly decreasing sequences $\langle U_n: n < \omega \rangle$, $\langle V_n: n < \omega \rangle$ of clopen subsets of X such that $\bigcap_{n=0}^{\infty} U_n = \{x\}, \bigcap_{n=0}^{\infty} V_n = \{y\}$; of course we may assume that $U_0 = X = V_0$. Now for $n < \omega$, put $U'_n = U_n \setminus U_{n+1}, V'_n = V_n \setminus V_{n+1}$; then U'_n, V'_n are non-empty clopen subsets of X, and hence there is a homeomorphism $h_n: U'_n \rightarrow V'_n$. Now define $h: X \rightarrow Y$ by $h|U'_n = h_n$ for each $n < \omega$, and h(x) = y. It is easily verified that h is a homeomorphism.

1.9.2 Let X be a non-compact zero-dimensional strongly homogeneous space; then $U \approx X$ for each non-empty open subspace of X, and $X \approx \omega \times X$.

Proof: Let U be a non-empty open subspace of X. Then U is not compact (otherwise, U is clopen in X, so $U \approx X$; but X is non-compact), so there

is an infinite disjoint covering $\{U_n : n < \omega\}$ of U by clopen subsets of X. Then each $U_n \approx X$, so $U \approx \omega \times X$.

1.10 The reader is assumed to be familiar with the main facts about absolute Borel sets, which can be found in Kuratowski [28]. However, in this monograph, it turns out that it is often convenient to use the *notation* from Moschovakis [43]:

Let X be a space. We inductively define the Borel classes $\Sigma_{\alpha}^{0}, \Pi_{\alpha}^{0}$ in X, for $\alpha \in [1, \omega_{1})$, by: A is a Σ_{1}^{0} -set in X if A is open in X; A is a Π_{1}^{0} -set in X if A is closed in X. If $\alpha > 1$, then: A is a Σ_{α}^{0} -set in X if $A = \bigcup_{i=0}^{\infty} A_{i}$, where each A_{i} is a $\Pi_{\beta_{i}}^{0}$ -set in X for some $\beta_{i} < \alpha$; A is a Π_{α}^{0} -set in X if $A = \bigcap_{i=0}^{\infty} A_{i}$, where each A_{i} is a $\Sigma_{\beta_{i}}^{0}$ -set in X for some $\beta_{i} < \alpha$; A is a Π_{α}^{0} -set in X if $A = \bigcap_{i=0}^{\infty} A_{i}$, where each A_{i} is a $\Sigma_{\beta_{i}}^{0}$ -set in X for some $\beta_{i} < \alpha$ (i.e. if X\A is a Σ_{α}^{0} -set in X). For each $\xi \in [1, \omega_{1})$, we put $\Delta_{\ell}^{0} \in \Pi_{\ell}^{0} \cap \Sigma_{\ell}^{0}$.

Comparing this definition with Kuratowski's indexing of the Borel hierarchy, we see that unfortunately, there is a discrepancy at the finite levels: if $n < \omega$, then A is of additive (resp. multiplicative, resp. ambiguous) class n in X if and only if A is a \sum_{n+1}^{0} -set (resp. \prod_{n+1}^{0} -set, resp. Δ_{n+1}^{0} -set) in X. If $\alpha \ge \omega$, then there is no such discrepancy; thus, for all $\alpha < \omega_1$, we have:

A is of additive (resp. multiplicative, resp. ambiguous) class α in X if and only if A is a $\Sigma_{1+\alpha}^0$ -set (resp. $\Pi_{1+\alpha}^0$ -set, resp. $\Delta_{1+\alpha}^0$ -set) in X. If we say that A is a Σ_{ξ}^0 -set (resp. Π_{ξ}^0 -set, resp. Δ_{ξ}^0 -set) without mentioning a space X, then we mean that A is a Σ_{ξ}^0 -set, etc. absolutely, i.e. A is a Σ_{ξ}^0 -set, etc., in X for every space X, whenever A is embedded in X; in Kuratowski's terminology, A is absolutely of additive (resp. multiplicative, resp. ambiguous) class α , where $\xi = 1+\alpha$. The notion of an absolute Borel set is now clear.

It is easy to see that

A is Π_1^0 if and only if A is compact; A is Σ_2^0 if and only if A is σ -compact; and it is well-known that A is Π_2^0 if and only if A is complete ($\stackrel{\text{def}}{=}$ Čech-complete \equiv completely metrizable);

if $\xi \ge 3$ then A is Π^0_{ξ} (resp. Σ^0_{ξ}) if and only if A is Π^0_{ξ} (resp. Σ^0_{ξ}) in some completely metrizable space.

The last two statements can be deduced from Lavrentieff's theorem ([29]; see also Engelking [15] 4.3.21). Thus, for all classes $\Sigma_{\xi}^{0}, \Pi_{\xi}^{0}, \Delta_{\xi}^{0}$, excluding $\Sigma_{1}^{0}, \Delta_{1}^{0}$, a space is absolutely of that class if and only if it is of that class when embedded in some compact space; since zero-dimensional spaces can be embedded in the Cantor set, this is the reason that we mainly consider Borel subsets of the Cantor set in this monograph.

A space A is *exactly of additive class* α if it is absolutely of additive class α , but not absolutely of multiplicative class α , and A is *exactly of multiplicative class* α if it is absolutely of multiplicative class α but not absolutely of additive class α ; A is *of exact class* α if it is exactly of additive class α , or exactly of multiplicative class α . A is *exactly of ambiguous class* α if it is absolutely of ambiguous class α , but not absolutely of additive or multiplicative class β for any $\beta < \alpha$.

1.11 A class $\Gamma \subset P(X)$ (the power set of X) is said to satisfy the *reduc*tion property if for each $A_0, A_1 \in \Gamma$, there exist $B_0, B_1 \in \Gamma$ such that

> (i) $B_0 \subset A_0$, $B_1 \subset A_1$; (ii) $A_0 \cup A_1 = B_0 \cup B_1$; (iii) $B_0 \cap B_1 = \emptyset$.

It is well-known (Kuratowski [28] §21.II and §27.VII) that the reduction property holds for the class of Σ_{ξ}^{0} -sets in X, if $\xi \ge 2$; if X is zero-dimensional, then the reduction property also holds for the class of Σ_{1}^{0} -sets in X. In fact, for these classes, we can also reduce *infinite* sequences: if $\langle A_{n}: n < \omega \rangle$ is a sequence of Σ_{ξ}^{0} -sets in X (where $\xi \ge 2$, and $\xi \ge 1$ if X is zero-dimensional), then there exists a sequence $\langle B_{n}: n < \omega \rangle$ of Σ_{ξ}^{0} -sets in X reducing $\langle A_{n}: n < \omega \rangle$, i.e. for all $n, B_{n} \subset A_{n}$; for all $n \neq m$, $B_{n} \cap B_{m} = \emptyset$; and $\bigcup_{n=0}^{\infty} A_{n} = \bigcup_{n=0}^{\infty} B_{n}$.

1.12 A space X is *first category* if $X = \bigcup_{i=0}^{\infty} X_i$, where each X_i is nowhere dense in X; X is *Baire* if the intersection of countably many dense open subsets of X is dense in X.

1.12.1 A homogeneous space X is either Baire or first category.

Proof: If X is not Baire, then there exists a family $\{U_n: n < \omega\}$ of dense open subsets of X, and a non-empty open subset V of X, such that $V \subset X \setminus (\bigcap_{n=0}^{\infty} U_n) = \bigcup_{n=0}^{\infty} (X \setminus U_n)$; note that each $X \setminus U_n$ is nowhere dense in X. Let $x \in V$; by homogeneity of X, we can find, for each $y \in X$, a homeomorphism $h_y: X \to X$ such that $h_y(x) = y$. Now if $\{h_{y_n}[V]: n < \omega\}$ is a countable subcovering of $\{h_y[V]: y \in X\}$, then $X = \bigcup_{n=0}^{\infty} \bigcup_{m=0}^{\infty} h_{y_n}[X \setminus U_m]$, so X is first category.

The next statement is a special case of a result of Levi [31].

1.12.2 Let A be an absolute Borel set. Then A is Baire if and only if A contains a dense complete subspace.

Proof: Suppose A is Baire, and embed A densely in a compact space X. By Kuratowski [28]\$35.II, there exist an open subset U of X, and nowhere dense subsets P_i, R_i of X such that

$$A = (U \setminus U_{i=0}^{\infty} P_i) \cup U_{i=0}^{\infty} R_i.$$

Then $U \setminus (U_{i=0}^{\infty} \overline{P}_i)$ is dense in A, and a G_{δ} in X, whence complete. The converse is trivial.

1.12.3 Let A be a dense and co-dense Borel set in a compact space X. Then A is Baire if and only if $X \setminus A$ is first category.

Proof: If A is Baire, then A contains a dense complete subspace G by 1.12.2; then G is a G_{δ} in X, say $G = \bigcap_{n=0}^{\infty} U_n$, where each U_n is open in X. Since A is dense in X, $X \setminus U_n$ is nowhere dense in X, and since X\A is dense in X, $(X \setminus A) \cap (X \setminus U_n)$ is nowhere dense in X\A; thus,

$$X \setminus A = \bigcup_{n=0}^{\infty} ((X \setminus A) \cap (X \setminus U_n))$$

is first category.

Conversely, if $X \setminus A = \bigcup_{n=0}^{\infty} F_n$, where F_n is closed and nowhere dense in $X \setminus A$, then each \overline{F}_n is nowhere dense in X; since X is compact, $\bigcup_{n=0}^{\infty} \overline{F}_n$ has empty interior in X, so $\bigcap_{n=0}^{\infty} (X \setminus \overline{F}_n)$ is a dense subset of A which is a G_{δ} in X, whence complete.

CHAPTER 2: BOREL SETS OF CLASS 1

In this chapter, we consider the characterizations of zero-dimensional absolute Borel sets that were obtained between 1910 and 1930. In section 2.1, we prove Brouwer's characterization of the Cantor set. From this, Alexandroff and Urysohn's characterization of $C \setminus \{p\}$ follows rather easily (section 2.2). The space of irrationals P, also characterized by Alexandroff and Urysohn, is discussed in section 2.3, and in section 2.4, we obtain characterizations of the homogeneous σ -compact spaces Q (due to Sierpiński) and $Q \times C$ (again due to Alexandroff and Urysohn).

The proofs that we give are essentially the same as the original ones, although of course, terminology has somewhat changed.

2.1 The multiplicative class 0

The only space which really deserves to be called *the* Cantor set is the set defined by Cantor in 1883 ([5]), consisting of all points in the closed unit interval [0,1] that do not require the use of "1" in their ternary expansion; this space, which is denoted by C, is compact, zero-dimensional, and dense in itself. In this monograph, by the name Cantor set we usually denote *any* compact zero-dimensional space without isolated points; by the following theorem, this is topologically justified.

2.1.1 THEOREM (Brouwer [4]): Up to homeomorphism, the Cantor set C is the only zero-dimensional compact space without isolated points.

Proof: Let X and Y be such spaces. First note that if $\ell \in N$, and A is an arbitrary non-empty clopen subset of X or Y, then since A is dense in itself, we can write A as a disjoint union of ℓ non-empty clopen sets. To prove the theorem, we will inductively construct finite indexing sets E_i , and non-empty clopen subsets $x_{i_0...i_n}$ of X, $y_{i_0...i_n}$ of Y, for each $(i_0, \dots, i_n) \in \prod_{j=0}^n E_j$, and each $n < \omega$, such that

- (1) $X_0 = X$, $Y_0 = Y$, $E_0 = \{0\}$;
- (2) $X_{i_0...i_n} = U_{i \in E_{n+1}} X_{i_0...i_n}$; $Y_{i_0...i_n} = U_{i \in E_{n+1}} Y_{i_0...i_n}$; (3) $diam(X_{i_0...i_n}) \le 1/(n+1)$, $diam(Y_{i_0...i_n}) \le 1/(n+1)$;
- (4) $X_{i_0...i_n} \cap X_{j_0...j_n} = \emptyset = Y_{i_0...i_n} \cap Y_{j_0...j_n}$ if $(i_0,...,i_n)$ ≠ (j₀,...,j_n).

Define X_0 , Y_0 , and E_0 as in (1); then suppose that E_n , X_{i_0} ... i_n , $Y_{i_0...i_n}$ have been defined for each $n \leq k$, and each $(i_0,...,i_n) \in \prod_{j=0}^n E_j$. Since the non-empty clopen subsets of X of diameter $\leq 1/(k+2)$ form a basis for X, we can write the compact set $X_{i_0\cdots i_k}$ as a finite disjoint union of such sets, say

$$X_{i_0\cdots i_k} = U\{A_{i_0\cdots i_k} : i \leq s(i_0\cdots i_k)\};$$

similarly, write

$$Y_{i_0\cdots i_k} = \bigcup \{B_{i_0\cdots i_k} : i \leq t(i_0\cdots i_k)\}.$$

Put

$$\mathbf{m} = \max\{\mathbf{s}(\mathbf{i}_0 \dots \mathbf{i}_k), \mathbf{t}(\mathbf{i}_0 \dots \mathbf{i}_k): (\mathbf{i}_0, \dots, \mathbf{i}_k) \in \prod_{j=0}^k \mathbb{E}_j\}$$

and $E_{k+1} = \{0, ..., m\}$.

By the above remark, we can write each $A_{i_0...i_k}s(i_0...i_k)$ as a disjoint union of m+l-s($i_0 \dots i_k$) non-empty disjoint clopen sets, say

$${}^{A_{i_0\cdots i_k}s(i_0\cdots i_k)} = \bigcup \{ X_{i_0\cdots i_k}i : i = s(i_0\cdots i_k), \dots, m \};$$

similarly,

$$\dots_{i_k} t(i_0 \dots i_k) = U\{Y_{i_0 \dots i_k} : i = t(i_0 \dots i_k), \dots, m\}.$$

Finally, for $i < s(i_0 \dots i_k)$, put $X_{i_0 \dots i_k} i = A_{i_0 \dots i_k} i$, and for $i < j_1 < j_2 \dots j_k i$ $t(i_0...i_k)$, put $Y_{i_0...i_k}i = B_{i_0...i_k}i$. Then (2) -(4) are satisfied for n = k+1; this completes the induction. Now by compactness of X and Y, for each $\sigma \in \Pi_{j=0}^{\infty} E_j$, $\bigcap_{s < \sigma} X_s \neq \emptyset \neq \bigcap_{s < \sigma} Y_s$, hence by condition (3), both $\bigcap_{s < \sigma} X_s$ and $\bigcap_{s < \sigma} Y_s$ consist of exactly one point, say x_{σ} resp. y_{σ} . Define f: $X \rightarrow Y$ by $f(x_{\sigma}) = y_{\sigma}$, for each $\sigma \in \Pi_{j=0}^{\infty} E_j$; it is easily verified that f is a homeomorphism.

2.1.2 COROLLARY: C is strongly homogeneous, hence homogeneous.

Proof: A non-empty clopen subset of C is compact, zero-dimensional, and dense in itself, hence homeomorphic to C by the theorem.

Clearly, it also follows from the theorem that C and the finite spaces are the only zero-dimensional homogeneous compact spaces, i.e. the only zero-dimensional homogeneous absolute Borel sets of multiplicative class 0.

Since every zero-dimensional space can be embedded in C, the *compact* zerodimensional spaces are exactly the *closed* subsets of C; this can also be regarded as a consequence of the following:

2.1.3 COROLLARY: Let X be a non-empty compact zero-dimensional space. Then $X \times C \approx C$.

Proof: Apply the theorem.

Another corollary, which we will frequently use, is:

2.1.4 COROLLARY: If X is a zero-dimensional space without isolated points, then X can be densely embedded in C; if X is also nowhere compact, then any such embedding is co-dense.

Proof: Embed X in C, and let $K = \overline{X}$. Then K is compact zero-dimensional, and since X contains no isolated points, neither does K; so $K \approx C$ by theorem 2.1.1. If X is nowhere compact, and U is a non-empty clopen subset of K, then $U \notin X$, so $U \cap K \setminus X \neq \emptyset$, i.e. X is co-dense in K.

As yet another example of the strength of the above characterization, we give a short proof of the following special case of a theorem of Suslin:

2.1.5 COROLLARY: Let X be an uncountable zero-dimensional σ -compact space; then X contains a copy of the Cantor set.

Proof: Write $X = \bigcup_{i=0}^{\infty} X_i$, with each X_i compact. Then some X_j is uncountable. By the Cantor-Bendixson theorem (see e.g. Engelking [15] 1.7.11), X_j contains a closed subset K which is either empty or dense in itself, and a countable subset L, such that $X_j = K \cup L$. Since X_j is uncountable, K is non-empty, and hence $K \approx C$ by theorem 2.1.1.

2.2 The ambiguous class 1

Absolute Borel sets of ambiguous class 1 are those sets which are both σ compact and complete. The following theorem gives an alternative description; it also appears in Kuratowski [28](§12.IX), but the proof there is rather indirect.

2.2.1 THEOREM: A space X is both σ -compact and complete if and only if each non-empty closed subset of X contains a non-empty open locally compact subspace.

Proof: If X is σ -compact and complete, and A is non-empty and closed in X, then A is also σ -compact, say $A = \bigcup_{i=0}^{\infty} A_i$ with A_i compact. A is also complete, hence Baire, so $\operatorname{Int}_A(A_j) \neq \emptyset$ for some j; then $\operatorname{Int}_A(A_j)$ is a non-empty locally compact open subset of A.

For the converse, first suppose that $\ X$ is not $\sigma\text{-compact;}$ put

 $V = U{U: U \text{ is open in } X, \text{ and } \sigma\text{-compact}}.$

Then V is also σ -compact, so $A = X \setminus V \neq \emptyset$. Let W be a non-empty open subset of A, say $W = W' \cap A$, with W' open in X. If W were locally compact, then it would be σ -compact. Since $W' \cap V$ is also σ -compact, W' would be σ -compact, contradicting the fact that $W' \notin V$. If X is not complete, then the proof is wordly the same once every occur-

rence of " σ -compact" is replaced by "complete"; the fact that V is complete is a consequence of Montgomery's theorem that a locally complete space is complete ([42]; see also Michael [37] and Stone [57]).

2.2.2 COROLLARY: A homogeneous space X is both σ -compact and complete if and only if it is locally compact.

Proof: Sufficiency is clear. If X is σ -compact and complete, then by theorem 2.2.1, X contains some non-empty open locally compact subset; hence by homogeneity, X itself is locally compact.

Remark: Without the homogeneity condition, the corollary is false: the space $[0,1]\setminus\{1/n: n \in N\}$ is a counterexample.

From corollary 2.2.2, we see that if we want to characterize homogeneous zero-dimensional absolute Borel sets of ambiguous class 1, then we have to consider locally compact homogeneous subsets of C. Since compact spaces

were discussed in section 2.1, we will now look at the non-compact sets. The homogeneity requirement yields that the space either consists exclusively of isolated points, and thus is infinite and discrete, or is dense in itself. The following theorem shows that the latter condition determines precisely one topological type.

2.2.3 THEOREM (Alexandroff and Urysohn [1]): Up to homeomorphism, the Cantor set minus one point $C \setminus \{p\}$ is the only zero-dimensional locally compact non-compact space without isolated points.

Proof: Clearly, if $p \in C$, then $C \setminus \{p\}$ is a zero-dimensional locally compact non-compact space without isolated points. If X is any such space, then $\widetilde{X} = X \cup \{\infty\}$, the Alexandroff one-point compactification of X, is a compact zero-dimensional space without isolated points, hence $X \approx C$ by theorem 2.1.1. Since C is homogeneous (corollary 2.1.2), there exists a homeomorphism h: $\widetilde{X} \to C$ such that $h(\infty) = p$; then $h[X] = C \setminus \{p\}$.

Remark: The above proof is the original one, as given by Alexandroff and Urysohn. We feel that the following proof is slightly more elementary, since it does not mention the one-point compactification: Cover X by disjoint non-empty compact open sets; then this covering is infinite, and the compact open sets are Cantor sets by theorem 2.1.1, so $X \approx \omega \times C$.

Since $C \setminus \{p\} \approx \omega \times C$, it is homogeneous. However, it is *not* strongly homogeneous, since it contains non-empty open compact subsets; of all the spaces discussed in this monograph, apart from the discrete spaces, $C \setminus \{p\}$ is the only one which is homogeneous, but not strongly homogeneous (see corollary 4.4.6).

2.2.4 COROLLARY: Let X be a non-empty locally compact zero-dimensional space. Then $X \times C \setminus \{p\} \approx C \setminus \{p\}$; thus, X can be embedded as a closed subspace of $C \setminus \{p\}$.

Proof: Apply theorem 2.2.3.

2.3 The multiplicative class 1

By corollary 2.2.2, a homogeneous complete space is σ -compact if and only if it is locally compact. Thus, the homogeneous zero-dimensional absolute Borel sets that are exactly of multiplicative class 1 are those homogeneous subsets of C that are complete and nowhere compact.

2.3.1 THEOREM (Alexandroff and Urysohn [1]): Up to homeomorphism, the space of irrationals P is the only zero-dimensional complete space which is no-where compact.

Proof: It is well-known, and easy to show, that P is complete and nowhere compact. So suppose that X and Y are two zero-dimensional, complete, nowhere compact spaces. Using an argument much like that in the proof of theorem 2.1.1, we will show that $X \approx Y$.

Fix complete metrics d and d' for X and Y, respectively. We will construct non-empty clopen subsets X_s of X, Y_s of Y, for each $s \in M$ (= $\omega^{<\omega}$, see 1.4), such that

(1) $X_{\emptyset} = X, Y_{\emptyset} = Y;$ (2) $X_{s} = \bigcup_{i=0}^{\infty} X_{s^{i}}, Y_{s} = \bigcup_{i=0}^{\infty} Y_{s^{i}};$ (3) $d-diam(X_{s}) \le 1/(|s|+1), d'-diam(Y_{s}) \le 1/(|s|+1);$ (4) $X_{s} \cap X_{t} = \emptyset = Y_{s} \cap Y_{t}$ if $|s| = |t|, s \ne t.$

The construction is a triviality once it is noted that, since X and Y are nowhere compact, any non-empty clopen subset of X or Y can be written as a disjoint union of *infinitely* many non-empty clopen sets (of arbitrarily small diameter).

By completeness of d and d', and by condition (3), for each $\sigma \in \omega^{\omega}$, the sets $\bigcap_{s < \sigma} X_s$, $\bigcap_{s < \sigma} Y_s$ contain exactly one point, say x_{σ} resp. y_{σ} . If we put $f(x_{\sigma}) = y_{\sigma}$, then f: $X \rightarrow Y$ is easily seen to be a homeomorphism.

Remark: If X is just a zero-dimensional nowhere compact space, then the above proof shows that X can be densely embedded in Y: indeed, if $x \in X$, then for some $\sigma \in \omega^{\omega}$ we have $x \in \bigcap_{s < \sigma} X_s$; now put $f(x) = y_{\sigma}$, then f is a dense embedding.

Remark: The above characterization of P was also obtained by Hausdorff [22], much later, in 1937.

2.3.2 COROLLARY: P is strongly homogeneous, hence homogeneous.

Thus, P is the only zero-dimensional homogeneous absolute Borel set which is exactly of multiplicative class 1. Note that from theorem 2.3.1 it follows, among others, that P is homeomorphic to Baire space ω^{ω} .

Similar to 2.1.3 and 2.2.4, we have:

2.3.3 COROLLARY: Let X be a non-empty zero-dimensional complete space. Then $X \times P \approx P$; thus, X can be embedded as a closed subspace of P.

2.4 The additive class 1

By corollary 2.2.2, a homogeneous σ -compact space is complete if and only if it is locally compact. Thus, the homogeneous zero-dimensional absolute Borel sets that are exactly of additive class 1 are those homogeneous subsets of C that are σ -compact and nowhere compact. Such a space is either countable or nowhere countable; we will show that each possibility yields exactly one topological type.

2.4.1 THEOREM (Sierpiński [50]): Up to homeomorphism, the space of rationals Q is the only countable space without isolated points.

Proof: Let X and Y be countable spaces without isolated points. Then X and Y are zero-dimensional, so X and Y can be considered as dense subsets of $\{0,1\}^{\omega} \approx C$ (corollary 2.1.4). Enumerate X,Y as $\{x_n: n < \omega\}, \{y_n: n < \omega\}$, respectively; we write $x_n \leq x_m$, $y_n \leq y_m$ if $n \leq m$. For each $n < \omega$, and each $(\alpha_0, \dots, \alpha_n) \in \{0,1\}^{n+1}$, define

(i)
$$p_0 = \min\{x \in X: \pi_0(x) = 0\}, p_1 = \min\{x \in X: \pi_0(x) = 1\};$$

(ii) $p_{\alpha_0...\alpha_n} = p_{\alpha_0...\alpha_n}$
(iii) $p_{\alpha_0...\alpha_n} = \min\{x \in X: \pi_{\leq n}(x) = \pi_{\leq n}(p_{\alpha_0...\alpha_n}), \pi_{n+1}(x) = 1 - \pi_{n+1}(p_{\alpha_0...\alpha_n})\}$

Observe that

(1)
$$p_{\alpha_0...\alpha_n} = \min(X \cap \pi_{\leq n}^{-1} \pi_{\leq n}(p_{\alpha_0...\alpha_n}));$$

(2) if $p_{\alpha_0...\alpha_n} = p_{\beta_0...\beta_n}$, then $\alpha_i = \beta_i$ for each $i \leq n$.

Also, note that

(3) each point of X is some $p_{\alpha_0...\alpha_n}$.

Indeed, let $x \in X$. Put $\alpha_0 = \pi_0(x)$, and if $\alpha_0, \ldots, \alpha_k$ have been defined, put $\alpha_{k+1} = 0$ if $\pi_{k+1}(x) = \pi_{k+1}(p\alpha_0 \ldots \alpha_k)$, $\alpha_{k+1} = 1$ otherwise. It is easily verified that $\pi_{\leq n}(x) = \pi_{\leq n}(p\alpha_0 \ldots \alpha_n)$ for each n. Hence by (1), $p\alpha_0 \ldots \alpha_n$ $\leq x$ for each n, and thus for some $y \leq x$, $y = p\alpha_0 \ldots \alpha_n$ for infinitely many n. Since $\pi_{\leq k}^{-1} \pi_{\leq k}(p\alpha_0 \ldots \alpha_k) \supset \pi_{\leq k+1}^{-1} \pi_{\leq k+1}(p\alpha_0 \ldots \alpha_{k+1})$ for each k, we have $y \in \bigcap_{n=0}^{\infty} \pi_{\leq n}^{-1} \pi_{\leq n}(p\alpha_0 \ldots \alpha_n)$, a set which consists of exactly one point. But also

$$x \in \bigcap_{n=0}^{\infty} \pi_{\leq n}^{-1} \pi_{\leq n}(p_{\alpha_0}, \ldots, \alpha_n),$$

so y = x, whence $x = p_{\alpha_0...\alpha_n}$ for some n. Define points $q_{\alpha_0...\alpha_n}$ of Y in the same way, with similar properties. We claim that $f: X \to Y$ defined by $f(p_{\alpha_0...\alpha_n}) = q_{\alpha_0...\alpha_n}$ is a homeomorphism. By (2) and (3), f is a bijection. We will show that f is continuous; the continuity of f^{-1} can be proved in exactly the same way. Let $q_{\alpha_0...\alpha_n} \in U = \pi_{\leq m}^{-1} \pi_{\leq m}(q_{\alpha_0...\alpha_n})$. Of course we may assume that $m \ge n$, and by (ii) we can in fact take m = n. We will show that $f[X \cap \pi_{\leq n}^{-1} \pi_{\leq n}(p_{\alpha_0...\alpha_n})] \subset U$. Indeed, take $p_{\beta_0...\beta_k} \in \pi_{\leq n}^{-1} \pi_{\leq n}(p_{\alpha_0...\alpha_n})$. Again by (ii), we may assume that k > n. Then

$$\pi_{\leq n}(p_{\alpha_0\ldots\alpha_n}) = \pi_{\leq n}(p_{\beta_0\ldots\beta_k}) = \pi_{\leq n}(p_{\beta_0\ldots\beta_n}),$$

so $p_{\alpha_0...\alpha_n} = p_{\beta_0...\beta_n}$ by (1), whence $\alpha_i = \beta_i$ for each $i \le n$ by (2). Thus,

$$f(p_{\beta_0...\beta_k}) = q_{\beta_0...\beta_k} = q_{\alpha_0...\alpha_n\beta_{n+1}...\beta_k},$$

and since $\pi_{\leq n}(q_{\alpha_0...\alpha_n\beta_n+1...\beta_k}) = \pi_{\leq n}(q_{\alpha_0...\alpha_n})$, we find that $f(p_{\beta_0...\beta_k}) \in U$.

Remark: Sierpiński, in his original proof, considered subsets X,Y of some \mathbb{R}^n , but the technique described here is the same. The proof presented here is certainly not the most elegant one possible, e.g. consider the following argument: since X and Y are countable and dense in itself, they are zero-dimensional and nowhere compact; hence by the remark following theorem 2.3.1, each can be densely embedded in the irrationals considered as a subset of \mathbb{R} , thus X and Y are order-isomorphic, and hence they are homeomorphic. Yet another proof will be given in Chapter 3 (remark following theorem 3.2.4).

2.4.2 COROLLARY: ${\it Q}$ is strongly homogeneous, hence homogeneous.

2.4.3 COROLLARY: Let X be a non-empty countable space. Then $X \times Q \approx Q$; thus, X can be embedded as a closed subspace of Q.

We now turn to the "nowhere countable" case; we start with a lemma (here, "unbounded" means: unbounded left and right).

2.4.4 LEMMA:(a) Let X_0, X_1 be Cantor sets in **R**. Then there exists an orderpreserving homeomorphism h: $X_0 \rightarrow X_1$.

(b) Let X_0, X_1 be closed, nowhere dense, unbounded subsets of R without isolated points. Then there exists an order-preserving homeomorphism h: $X_0 \rightarrow X_1$.

Proof: (a) For $i \in \{0,1\}$, let $\mathcal{U}_i = \{(a_n^i, b_n^i): n < \omega\}$ be the collection of components of $[\min X_i, \max X_i] \setminus X_i$. Since X_i is dense in itself, $b_n^i \neq a_m^i$ for each n,m; thus, we can linearly order \mathcal{U}_i by: $(a_n^i, b_n^i) < (a_m^i, b_m^i)$ if and only if $b_n^i < a_m^i$. If $b_n^i < a_m^i$, then since X_i is nowhere dense in R, we have $(a_k^i, b_k^i) < (b_n^i, a_m^i)$ for some k, i.e. we have $(a_n^i, b_n^i) < (a_k^i, b_k^i) < (a_m^i, b_m^i)$. Furthermore, for each n, we have $b_k^i < a_n^i$ and $b_n^1 < a_\ell^i$ for some k, ℓ , i.e. we have $(a_k^i, b_k^i) < (a_k^i, b_k^i) < (a_n^i, b_n^i) < (a_\ell^i, b_\ell^i)$. Thus, $(\mathcal{U}_i, <)$ is order-isomorphic to Q (see Sierpiński [53] XI.8, theorem 1), and we can define an order-preserving homeomorphism f: $(\mathcal{U}_0, <) \rightarrow (\mathcal{U}_1, <)$; it is trivial to define an order-preserving homeomorphism g: $\mathcal{U}\mathcal{U}_0 \rightarrow \mathcal{U}\mathcal{U}_1$, induced by f. Now define h: $X_0 \rightarrow X_1$ by

$$\begin{split} h(\min X_0) &= \min X_1; \\ h(x) &= \sup\{g(y): y < x, y \in UU_0\} \text{ if } x \neq \min X_0. \end{split}$$

It is routine to verify that h is as required. The proof of (b) is similar.

Remark: In fact, the homeomorphisms h in the above lemma can be defined as restrictions of order-preserving autohomeomorphisms of R.

2.4.5 THEOREM (Alexandroff and Urysohn [1]): Up to homeomorphism, the product of the rationals with the Cantor set, $Q \times C$, is the only zero-dimensional σ -compact space which is nowhere compact and nowhere countable.

Proof: It is easily seen that $Q \times C$ is σ -compact, nowhere compact, and nowhere countable.

In Alexandroff and Urysohn's paper, the space $Q \times C$ is not mentioned at all; they use a "test space" Ψ which is defined as follows. If $U = (a,b) \subset R$, denote by P(U) the "real" Cantor set in [a,b], obtained by subsequently deleting open middle third intervals. Put $P_0 = \bigcup_{n \in \mathbb{Z}} P((n,n+1))$; note that P_0 is closed in R. If the closed subset P_n of R has been defined, then we put $P_{n+1} = P_n \cup U\{P(U): U$ is a component of $R \setminus P_n\}$. Finally, define $\Psi = \bigcup_{n=0}^{\infty} P_n$. The sets P_n have the following properties:

- P_n is a closed, nowhere dense, unbounded subset of R without isolated points;
- (2) $P_n \subset P_{n+1};$
- (3) if (a,b) is a component of $R \setminus P_n$, then $a = \inf(P_{n+1} \cap (a,b))$, $b = \sup(P_{n+1} \cap (a,b))$.

Now let X be any zero-dimensional, σ -compact, nowhere countable, nowhere compact space. By the remark following the proof of theorem 2.3.1, we can consider X as a dense subset of $P \subset R$. We will show that we can write X = $\bigcup_{n=0}^{\infty} X_n$, where the sets X_n satisfy properties similar to those of the sets P_n .

Let $X = \bigcup_{n=0}^{\infty} K_n$, with K_n compact, and let $A \in X$ be discrete, and closed unbounded in R. Then $Y_n = K_n \cup A$ is closed, unbounded, and nowhere dense in R. Fix $n < \omega$, and let $\{z_i: i < \omega\}$ be the set of isolated points of Y_n . Let $\{U_i: i < \omega\}$ be a disjoint collection of open intervals such that $U_i \cap Y_n = \{z_i\}$ for $i < \omega$. Since X is σ -compact, so is each open subset of X; hence, since X is nowhere countable, we can apply corollary 2.1.5 to obtain, for each $k \in N$, a Cantor set C_k^i in $U_i \cap (z_i - \frac{1}{k}, z_i + \frac{1}{k}) \cap X$. Define

$$Z_n = Y_n \cup \bigcup \{C_k^i: k \in N, i < \omega\};$$

then $Z_n \subset X$ is a closed unbounded subset of R without isolated points. Finally, put $X_0 = Z_0$, and suppose that X_n has been defined for each $n \leq k$, satisfying the obvious analogues to properties (1), (2), and (3) above, and such that $Z_n \subset X_n$. Now if $\{(a_i, b_i): i < \omega\}$ is the family of components of $R \setminus X_n$, then as above, we can find Cantor sets $C_k^i(0)$ in $(a_i, b_i) \cap (a_i, a_i + \frac{1}{k}) \cap X$, $C_k^i(1)$ in $(a_i, b_i) \cap (b_i - \frac{1}{k}, b_i) \cap X$. Defining

$$X_{n+1} = X_n \cup Z_n \cup \bigcup \{C_k^1(j): j \in \{0,1\}, i < \omega, k \in N\},$$

(1)-(3) are satisfied.

Applying lemma 2.4.4(b) to P_0 and X_0 , we obtain an order-preserving homeomorphism $h_0: P_0 \rightarrow X_0$, by (1). Let (a,b) be a component of $R \setminus P_0$; by the construction of Ψ , [a,b] $\cap P_1 \approx C$. Since h_0 is order-preserving, $(h_0(a), h_0(b))$ is a component of $R \setminus X_0$, so by (1), (3), and theorem 2.1.1, we find that $[h_0(a), h_0(b)] \cap X_1 \approx C$. By lemma 2.4.4(a), there exists an order-preserving homeomorphism $h_1(a,b): [a,b] \cap P_1 \rightarrow [h_0(a), h_0(b)] \cap X_1$, and by (3),

$$h_1 = h_0 \cup \bigcup \{h_1(a,b): (a,b) \text{ is a component of } R \setminus P_0\}$$

extends h_0 . It is now easy to construct subsequently order-preserving homeomorphisms $h_n: P_n \to X_n$. Then $\bigcup_{n=0}^{\infty} h_n = h: \Psi \to X$ is a homeomorphism. \Box

Remark: In [38], van Mill gave a new proof of this theorem. Again, our remark following theorem 3.2.4 will provide yet another proof (cf. remark following theorem 2.4.1).

We have the usual corollaries:

2.4.6 COROLLARY: $Q \times C$ is strongly homogeneous, hence homogeneous.

2.4.7 COROLLARY: Let X be a non-empty zero-dimensional σ -compact space. Then X × Q × C \approx Q × C; thus, X can be embedded as a closed subspace of Q × C.

21

CHAPTER 3: BOREL SETS OF AMBIGUOUS CLASS 2

Consider the following picture of the lower levels of the Borel hierarchy in the Cantor set:



The results in the preceding chapter indicate, that each of the spaces in the diagram is almost determined by its level in the Borel hierarchy; the most conspicuous example is P, a space which is completely characterized by its being Π_2^0 and nowhere Σ_2^0 . When looking for characterizations of spaces at level Δ_3^0 , i.e. the ambiguous class 2, the first idea that comes to mind might be that perhaps we are so lucky that there is precisely one zerodimensional space which is Δ_3^0 , nowhere Σ_2^0 , and nowhere Π_2^0 . However, results of van Mill ([38] and [39]) show that there are at least two homogeneous spaces with these properties, and, as we shall see, there are in fact ω_1 of them.

Thus, if we want to characterize spaces by way of their level in the Borel hierarchy, then we will have to refine this hierarchy at level Δ_3^0 , i.e. create a subhierarchy in Δ_3^0 , in such a way that there will be only one space (or maybe finitely many) at a certain level which is nowhere at a lower level. Fortunately, such a hierarchy is already available, viz. the hierarchy of "small Borel classes", which has length ω_1 ; in section 3.1, we

discuss this hierarchy, and change it slightly to make it better suit our needs. This will give the reader a rough idea of how the collection of homogeneous zero-dimensional absolute Borel sets of ambiguous class 2 is built up. Sections 3.2 and 3.3 contain the technical tools necessary to obtain the main results of this chapter, which are presented in sections 3.4 - 3.6. In section 3.4, we describe and characterize the spaces at the finite levels of our subhierarchy, and in section 3.5, the spaces at the infinite levels of this hierarchy; section 3.6 contains the main theorem, stating that the results of sections 3.4 and 3.5, together with those of Chapter 2, yield a complete picture of the homogeneous zero-dimensional absolute Borel sets of ambiguous class 2.

The main results of this chapter are taken from van Engelen [10].

3.1 Small Borel classes

3.1.1 DEFINITION: (a) Let X be a space, $A \subset X$, let $\eta < \omega_1$, and let $(A_{\zeta}: \zeta < \eta)$ be an increasing sequence of subsets of X. Then $A = D_{\eta}((A_{\zeta}: \zeta < \eta))$ if and only if $A = U\{A_{\zeta} \setminus (U_{\beta < \zeta} A_{\beta}): \zeta \text{ odd } < \eta\}$ when η is even,

resp. = $\bigcup \{A_{\zeta} \setminus (\bigcup_{\beta < \zeta} A_{\beta}): \zeta \text{ even } < \eta \}$ when η is odd. (b) Let X be a space, $A \subset X$, and let $\eta, \xi \in [1, \omega_1)$. Then $A \in D_{\eta}^{X}(\Sigma_{\xi}^{0})$ if and only if $A = D_{\eta}(\langle A_{\zeta}: \zeta < \eta \rangle)$ for some increasing sequence $\langle A_{\zeta}: \zeta < \eta \rangle$ of Σ_{ξ}^{0} -sets in X.

(c) Let $n, \xi \in [1, \omega_1)$. Then $A \in D_{\eta}(\Sigma_{\xi}^0)$ if and only if $A \in D_{\eta}^X(\Sigma_{\xi}^0)$ for every space X, whenever A is embedded in X.

Part (c) says that A is $D_{\eta}(\Sigma_{\xi}^{0})$ absolutely (note that $D_{1}(\Sigma_{\xi}^{0}) = \Sigma_{\xi}^{0}$). It easily follows from Lavrentieff's theorem that if $\xi \ge 2$, then this is equivalent to $A \in D_{\eta}^{X}(\Sigma_{\xi}^{0})$ for some compact space X; thus, if A is zero-dimensional, then $A \in D_{\eta}(\Sigma_{\xi}^{0})$ if and only if $A \in D_{\eta}^{C}(\Sigma_{\xi}^{0})$ for some embedding of A in C.

The above definition is basically due to Kuratowski ([28] §33.IV; for $\xi = 1$, Hausdorff [21]); he called the $D_{\eta}(\Sigma_{\xi}^{0})$ the "small Borel classes". Here, we have more or less adopted the notation from Louveau [32] since this definition and other ones from that paper will be needed anyway in Chapters 4 and 5.

3.1.2 THEOREM (Kuratowski [28]; for $\xi = 1$, Hausdorff [21]): Let A be a subset of a compact space X, and let $\xi \in [1, \omega_1)$. Then $A \in \Delta^0_{\xi+1}$ if and only if $A \in D^X_n(\Sigma^0_{\xi})$ for some $\eta \in [1, \omega_1)$.

Of course, in this chapter we will be interested in the case $\xi = 2$, X = Cof the above theorem; the sets A_{ζ} of definition 3.1.1(a) will then be σ compact subsets of C. Equivalently, we will work with zero-dimensional elements of the absolute classes $D_{\eta}(\Sigma_2^0)$. Clearly, $D_{\alpha}(\Sigma_2^0) \in D_{\beta}(\Sigma_2^0)$ if $\alpha \leq \beta$ (and in fact, $D_{\alpha}(\Sigma_2^0) \neq D_{\beta}(\Sigma_2^0)$ if $\alpha < \beta$, see Lavrentieff [30], and also Luzin [34], Sierpiński [52]). Thus, the classes $D_{\eta}(\Sigma_2^0)$ form a hierarchy, which refines Δ_3^0 .

For even n, we are now going to give an alternative description of the classes $D_n(\Sigma_2^0)$, which is often easier to work with; it has the additional advantage over the original definition that it describes the elements of the classes *internally*.

First consider the "building blocks" of the elements of $D_{\eta}(\Sigma_2^0)$; they are sets of the form $F_1 \setminus F_0$, where F_0, F_1 are σ -compact, i.e. they are elements of $D_2(\Sigma_2^0)$.

3.1.3 LEMMA: $A \in D_2(\Sigma_2^0)$ if and only if there exist closed, complete subsets A_i of A such that $A = \bigcup_{i=0}^{\infty} A_i$.

Proof: Embed A in a compact space X, and first suppose that $A \in D_2(\Sigma_2^0)$, say $A = F_1 \setminus F_0$, with F_0, F_1 σ -compact; then $G = X \setminus F_0$ is complete, and $F_1 = \bigcup_{i=0}^{\infty} K_i$, with K_i compact. Since $K_i \cap G$ is closed in G, it is complete, and closed in A. But $A = \bigcup_{i=0}^{\infty} (K_i \cap G)$.

Conversely, let $A = \bigcup_{i=0}^{\infty} A_i$, where A_i is complete and closed in A; then both $F_0 = \bigcup_{i=0}^{\infty} (\overline{A}_i \setminus A_i)$ and $F_1 = \bigcup_{i=0}^{\infty} \overline{A}_i$ are σ -compact, and $A = F_1 \setminus F_0$.

Because of the above lemma, elements of $D_2(\Sigma_2^0)$ will also be called *strong-ly* σ -complete spaces (see also definition 3.3.2(a)).

Note that the lemma internally describes the elements of $D_2(\Sigma_2^0)$; we will now do the same for $D_{2n}(\Sigma_2^0)$, where $n \in N$.

3.1.4 LEMMA: Let $n \in N$. Then $A \in D_{2n}(\Sigma_2^0)$ if and only if A is the union of n strongly σ -complete subspaces.

Proof: The "only if" part follows from lemma 3.1.3. For the converse, assume that A is embedded in a compact space X. We first establish the following CLAIM: If $B \subset X$, $m < \omega$, and $B = \bigcup_{i=0}^{m} (X_i \setminus Y_i)$, where X_i and Y_i are σ -

compact subsets of X, then there exist σ -compact \widetilde{X}_i and \widetilde{Y}_i in X such that $B = \bigcup_{i=0}^{m} (\widetilde{X}_i \setminus \widetilde{Y}_i)$, and $\bigcup_{i < m} (\widetilde{X}_i \setminus \widetilde{Y}_i) \subset \widetilde{Y}_m$.

Indeed, for i < m, put

$$\begin{split} \widetilde{\mathbf{X}}_{i} &= \mathsf{U}_{j=0}^{1}(\mathbf{Y}_{j} \cap \mathbf{X}_{i+1}) \cup \mathsf{U}_{j=0}^{1}(\mathbf{X}_{j} \cap \mathbf{Y}_{i+1}), \\ \widetilde{\mathbf{Y}}_{i} &= \mathsf{U}_{j=0}^{i}(\mathbf{Y}_{j} \cap \mathbf{Y}_{i+1}), \end{split}$$

and let

$$\begin{split} \widetilde{X}_{m} &= \bigcup_{j=0}^{m} X_{j}, \ \widetilde{Y}_{m} = \bigcup_{j=0}^{m} Y_{j}. \\ & \text{Clearly, } \bigcup_{i < m} (\widetilde{X}_{i} \setminus \widetilde{Y}_{i}) < \widetilde{Y}_{m}, \text{ and } \bigcup_{i=0}^{m} (\widetilde{X}_{i} \setminus \widetilde{Y}_{i}) < \bigcup_{i=0}^{m} (X_{i} \setminus Y_{i}). \text{ So let } n \\ &\leq m, \text{ and } x \in X_{n} \setminus Y_{n}. \text{ If } x \notin Y_{i} \text{ for all } i, \text{ then } x \in X_{n} \setminus (\bigcup_{j=0}^{m} Y_{j}) < \\ & \widetilde{X}_{m} \setminus \widetilde{Y}_{m}. \text{ Otherwise, there is a minimal } i \text{ with } x \in Y_{i}. \text{ If } i > n, \\ & \text{ then } i-1 \geq n \geq 0, \text{ and we claim that } x \in \widetilde{X}_{i-1} \setminus \widetilde{Y}_{i-1}; \text{ indeed, } x \in \\ & \bigcup_{j=0}^{i-1} (X_{j} \cap Y_{(i-1)+1}), \text{ and } x \notin \bigcup_{j=0}^{i-1} Y_{j} \text{ by minimality of } i. \text{ Since } x \\ & \notin Y_{n}, \text{ the last possibility is } i < n; \text{ but then } x \in \bigcup_{j=0}^{n-1} (Y_{j} \cap X_{(n-1)+1}), \\ & \text{ and } x \notin Y_{(n-1)+1}, \text{ whence } x \in \widetilde{X}_{n-1} \setminus \widetilde{Y}_{n-1}. \text{ This proves the claim.} \\ & \text{Now suppose that } A = \bigcup_{i=0}^{n-1} A_{i}, \text{ where } A_{i} \text{ is strongly } \sigma\text{-complete; by lemma} \end{split}$$

Now suppose that $A = \bigcup_{i=0}^{n-1} A_i$, where A_i is strongly σ -complete; by lemma 3.1.3, and by successive applications of the claim, we can write $A = \bigcup_{i=0}^{n-1} (X_i \setminus Y_i)$, where X_i and Y_i are σ -compact subsets of X, and $\bigcup_{i < m} (X_i \setminus Y_i)$ $\subset Y_m$ for each m < n. Now if i < n, put

$$\widetilde{C}_{2i} = \bigcap_{j \ge i} Y_j, \ \widetilde{C}_{2i+1} = X_i \cap \bigcap_{j > i} Y_j;$$

then $\widetilde{C}_{2i+1} \setminus \widetilde{C}_{2i} = X_i \setminus Y_i$. Finally, for each $\ell < 2n$, put

$$C_{\ell} = U_{m \leq \ell} \widetilde{C}_{m};$$

then $B = D_{2n}(\langle C_{\ell}: \ell < 2n \rangle)$. Indeed, the C_{ℓ} are clearly increasing; and if i < n, then $C_{2i+1} \setminus C_{2i} \subset \widetilde{C}_{2i+1} \setminus \widetilde{C}_{2i}$. Furthermore, $U_{j<2i} \subset \widetilde{C}_{j} \subset Y_{i}$, so $\widetilde{C}_{2i+1} \setminus \widetilde{C}_{2i}$.

The following lemma covers the limit case.

3.1.5 LEMMA: Let $\alpha < \omega_1$ be a limit. Then $A \in D_{\alpha}(\Sigma_2^0)$ if and only if there exist closed subsets A_i of A, and ordinals $\alpha_i \in [1, \alpha)$, such that $A_i \in D_{\alpha_i}(\Sigma_2^0)$ and $A = \bigcup_{i=0}^{\infty} A_i$.

Proof: Embed A in a compact space X. If $A = D_{\alpha}(\langle C_{\zeta} : \zeta < \alpha \rangle)$, with each C_{ζ} a σ -compact subset of X, and if $\langle \alpha_i : i < \omega \rangle$ is a sequence of even ordinals in [1, α) such that $\sup_{i < \omega} \alpha_i = \alpha$, then $A = \bigcup_{i=0}^{\infty} D_{\alpha_i}(\langle C_{\zeta} : \zeta < \alpha_i \rangle)$. Since $\bigcup_{\zeta < \alpha_i} C_{\zeta}$ is σ -compact,

$$D_{\alpha_{i}}(\langle C_{\zeta}: \zeta < \alpha_{i} \rangle) = \bigcup_{j=0}^{\infty} D_{\alpha_{i}}(\langle C_{\zeta} \cap K_{j}^{i}: \zeta < \alpha_{i} \rangle)$$

for certain compact subsets K_{j}^{i} of $\bigcup_{\zeta < \alpha_{i}} C_{\zeta}$. Since $D_{\alpha_{i}}(\langle C_{\zeta} \cap K_{j}^{i}: \zeta < \alpha_{i} \rangle)$

= $K_i^i \cap A$ is closed in A, A has the required form. Conversely, suppose that $A = \bigcup_{i=0}^{\infty} A_i$, with A_i closed in A, and $A_i \in$ $\mathbb{D}_{\alpha_i}(\Sigma_2^0)$ for certain $\alpha_i \in [1,\alpha]$; then also $A_i \in D_{\alpha}(\Sigma_2^0)$, say $A_i = D_{\alpha}(\langle A_{\zeta}^i \rangle)$ $\zeta < \alpha$). For each i < ω , let K be a compact subset of X such that A i = $K_i \cap A$, and put $L_i = K_i \setminus (U_{j \le i} K_j)$; note that L_i is σ -compact, and $A = K_i \setminus (U_{j \le i} K_j)$ $\bigcup_{i=0}^{\infty} (L_i \cap A_i).$

Now for $\zeta < \alpha$, put

$$A_{\zeta} = U_{i=0}^{\infty} (L_{i} \cap A_{\zeta}^{i}).$$

Clearly, the sets $\mbox{ A}_{\zeta}$ are $\sigma\mbox{-compact},$ and increasing. Furthermore,

$$D_{\alpha}(\langle A_{\zeta}: \zeta < \alpha \rangle) = U\{A_{\zeta+1} \setminus A_{\zeta}: \zeta \text{ even } < \alpha\}$$

$$= U\{U_{i=0}^{\infty}(L_{i} \cap A_{\zeta+1}^{i}) \setminus U_{i=0}^{\infty}(L_{i} \cap A_{\zeta}^{i}): \zeta \text{ even } < \alpha\}$$

$$= U\{U_{i=0}^{\infty}((L_{i} \cap A_{\zeta+1}^{i}) \setminus (L_{i} \cap A_{\zeta}^{i})): \zeta \text{ even } < \alpha\}$$

$$= U_{i=0}^{\infty}(L_{i} \cap U\{A_{\zeta+1}^{i} \setminus A_{\zeta}^{i}: \zeta \text{ even } < \alpha\})$$

$$= U_{i=0}^{\infty}(L_{i} \cap A_{i}) = A.$$

The third equality follows from the observation that $L_i \cap L_i = \emptyset$ if $i \neq j$.

We are now ready for the alternative description of the $D_{\alpha}(\Sigma_2^0)$.

3.1.6 THEOREM: Let $\alpha \in [1, \omega_1)$ be even, say $\alpha = \beta + 2n$, where $\beta = 0$ or $\lim(\beta), \text{ and } n < \omega. \text{ Then } A \in D_{\alpha}(\Sigma_{2}^{0}) \text{ if and only if } A = \bigcup_{i=0}^{\infty} A_{i} \cup \bigcup_{i < n} B_{i},$ where A_i is closed in A, $A_i \in D_{\beta_i}(\Sigma_2^0)$ for some $\beta_i \in [1,\beta)$, and B_i is strongly o-complete.

Proof: Embed A as a subset of a compact space X. If A $\in D_{\alpha}(\Sigma_2^0)$, then we can find the sets A_i as in the first part of the proof of lemma 3.1.5; the sets B. are obtained from lemma 3.1.3. For the converse, let F be a σ compact subset of X satisfying $F \cap A = \bigcup_{i=0}^{\infty} A_i$. By lemma 3.1.5, there exist σ -compact subsets $\{\widetilde{A}_{\zeta}: \zeta < \beta\}$ of X such that $\bigcup_{i=0}^{\infty} A_i = D_{\beta}(\langle \widetilde{A}_{\zeta}: \zeta < \beta\})$ $\zeta < \beta$), and by lemma 3.1.4, there exist σ -compact subsets $\{\widetilde{A}_{\beta+k}: k < 2n\}$ such that $\bigcup_{i < n} B_i = D_{2n}((\widetilde{A}_{\beta+k}: k < 2n))$. Now for $\zeta < \beta$, put

$$A_{\zeta} = \widetilde{A}_{\zeta} \cap F,$$

and for k < 2n, put

$$A_{\beta+k} = F \cup A_{\beta+k};$$

then $A = D_{\alpha}(\langle A_{\zeta} : \zeta < \alpha \rangle).$

Since $D_{2n}(\Sigma_2^0)$ is described internally by lemma 3.1.4, and since the β_i in the above theorem can be chosen to be even, we indeed have found (inductively) an *internal* description of the classes $D_{\alpha+2n}(\Sigma_2^0)$.

We will now describe the subhierarchy of Δ_3^0 that we will need. It is obtained from the classes $D_{\alpha}(\Sigma_2^0)$, for even α , by making a distinction between different types of "building blocks". These building blocks are just the strongly σ -complete spaces; however, a strongly σ -complete space might have the stronger property of being complete, or σ -compact, or even countable. Combining these topological properties, we find that the union of two building blocks can be of any of the following types (note that complete \cup complete \equiv complete, countable \cup countable \equiv countable, and σ -compact $\cup \sigma$ compact $\equiv \sigma$ -compact):

```
complete U countable;
strongly σ-complete U countable;
complete U σ-compact;
strongly σ-complete U σ-compact;
strongly σ-complete U complete;
strongly σ-complete U strongly σ-complete.
```

Adding another building block again yields six new topological properties; and proceeding in this way, we obtain topological properties describing elements of $\bigcup_{n=1}^{\infty} D_{2n}(\Sigma_2^0)$ as in the following definition; here, as in the remainder of this chapter,

if $n < \omega$, then a space X is said to have property S_n if it is the union of n strongly σ -complete subsets (i.e. X ϵ $D_{2n}(\Sigma_2^0)$ if n > 0, $X = \emptyset$ if n = 0).

3.1.7 DEFINITION: Let X be a topological space. Then for each $k < \omega$, X has property P_{4k} if and only if X is $S_k \cup complete$; X has property P_{4k+1} if and only if X is S_{k+1} ; X has property P_{4k+2}^1 if and only if X is $S_k \cup complete \cup countable$; X has property P_{4k+3}^1 if and only if X is $S_{k+1} \cup countable$ (also for k = -1); X has property P_{4k+2}^2 if and only if X is $S_k \cup complete \cup \sigma$ -compact; X has property P_{4k+3}^2 if and only if X is $S_k \cup complete \cup \sigma$ -compact; X has property P_{4k+3}^2 if and only if X is $S_{k+1} \cup \sigma$ -compact (also for k = -1). For sake of simplicity, we will write "X is $P_n^{(i)}$ " if X has one of the properties defined above, meaning that the index "i" may or may not be there.

Now suppose that the properties $P_n^{(i)}$ are ordered as in the definition, i.e.

$$P_{4k} < P_{4k+1} < P_{4k+2}^1 < P_{4k+3}^1 < P_{4k+2}^2 < P_{4k+3}^2 < P_{4k+3}^2 < P_{4(k+1)}.$$

In section 4 of this chapter, we will show that for each $n < \omega$, $i \in \{1,2\}$, there exists, up to homeomorphism, exactly one zero-dimensional space $\chi_{n}^{(i)}$ which is $P_{n}^{(i)}$ and nowhere $P_{m}^{(j)}$ for each $P_{m}^{(j)} < P_{n}^{(i)}$. The spaces $\chi_{n}^{(i)}$ are roughly described by the following diagram:



Here, the spaces S and T are the homogeneous complements of $Q \times P$ in the Cantor set (see the introduction of this monograph), X_4 is obtained as the homogeneous complement of $Q \times S$ or $Q \times T$ in the Cantor set, $X_5 = Q \times X_4$, and X_6^1 and X_6^2 are the only homogeneous complements of X_5 in the Cantor set. Then $X_7^1 = Q \times X_6^1$, $X_7^2 = Q \times X_6^2$, etc..

The spaces described above turn out to be the only zero-dimensional homogeneous (non σ -compact) elements of $\bigcup_{n=1}^{\infty} D_{2n}(\Sigma_2^0)$. We now turn to the classes $D_{\alpha+2n}(\Sigma_2^0)$, where $n < \omega$, and $\alpha < \omega_1$ is a limit

We now turn to the classes $D_{\alpha+2n}(\Sigma_2^0)$, where $n < \omega$, and $\alpha < \omega_1$ is a limit ordinal. Let us consider the class $D_{\omega+2}(\Sigma_2^0)$; each member X of this class is, by theorem 3.1.6, of the form $\bigcup_{i=0}^{\infty} A_i \cup B$, where A_i is closed in X, $A_i \in \bigcup_{n=1}^{\infty} D_{2n}(\Sigma_2^0)$, and B is strongly σ -complete. If B is σ -compact, then $X = \bigcup_{i=0}^{\infty} A_i \cup \bigcup_{j=0}^{\infty} B_j$, with B_j compact; hence X is already in $D_{\omega}(\Sigma_2^0)$:
the σ -compact part of X is "absorbed" by the part $\bigcup_{i=0}^{\infty} A_i$. Hence, if we want to define topological properties as in definition 3.1.7, then we need only distinguish between "B is complete" and "B is strongly σ -complete".

3.1.8 DEFINITION: Let X be a topological space, and let $\alpha < \omega_1$ be a limit ordinal. Then for each $n < \omega_s$

X has property $P_{\alpha+2n}$ if and only if $X \in D_{\alpha+2n}(\Sigma_2^0)$; X has property $P_{\alpha+2n+1}$ if and only if X is $P_{\alpha+2n} \cup$ complete.

In section 5, we will show that for each $m < \omega$, i $\in \{1,2\}$, and limit ordinal $\alpha < \omega_1$, there exists, up to homeomorphism, exactly one zero-dimensional space $X^i_{\alpha+m}$, such that

The spaces can be described as follows:

$$\mathbf{x}_{\alpha} - \mathbf{x}_{\alpha+1}^{1} - \mathbf{x}_{\alpha+2}^{1} = \mathbf{Q} \times \mathbf{x}_{\alpha+1}^{1} - \mathbf{x}_{\alpha+1}^{2} - \mathbf{x}_{\alpha+2}^{2} = \mathbf{Q} \times \mathbf{x}_{\alpha+1}^{2} - \mathbf{x}_{\alpha+1}^{2} - \mathbf{x}_{\alpha+2}^{2} = \mathbf{Q} \times \mathbf{x}_{\alpha+1}^{2} - \mathbf{x}_{\alpha+1}^{2} - \mathbf{x}_{\alpha+2}^{2} - \mathbf{x}_{$$

Here, $X_{\alpha+1}^1$ is a complement of X_{α} in the Cantor set, $X_{\alpha+1}^2$ is a complement of $X_{\alpha+2}^1$, $X_{\alpha+3}^1$ is a complement of $X_{\alpha+2}^2$, and so on. Furthermore, $X_{\alpha+4}^1 = Q \times X_{\alpha+3}^1$, $X_{\alpha+4}^2 = Q \times X_{\alpha+3}^2$, etc..

 $\overset{X}{}$ is obtained as a countable union of closed nowhere dense copies of previously defined spaces.

In this way, we obtain all homogeneous Borel sets of ambiguous class 2 in the Cantor set.

3.2 Knaster-Reichbach coverings and their applications

From section 3.1, we can see that, to obtain "new spaces from old ones", we frequently apply the following operations:

- (1) taking the product with Q;
- (2) taking complements in the Cantor set.

In this section, we will establish two very general theorems. The first (theorem 3.2.4) is due to Ostrovski^Y [44], and will enable us to handle (1); since Ostrovski^Y's paper is in Russian, we feel that it might be useful to include a proof of this theorem. The second (theorem 3.2.6) will be used to show that, in all cases under consideration in this chapter, if we have two subsets X_0, X_1 of the Cantor set which have as their complements the copies Y_0, Y_1 of the previously characterized space $Q \times X$, then there exists an autohomeomorphism h of the Cantor set such that $h[Y_0] = Y_1$, and hence $X_0 \approx X_1$; this will take care of (2) above.

3.2.1 DEFINITION: Let X and Y be zero-dimensional spaces, and let A be a closed nowhere dense subset of X and B a closed nowhere dense subset of Y. Suppose that h: $A \rightarrow B$ is a homeomorphism, $U = \{U_n: n < \omega\}$ is a covering of X\A by disjoint non-empty clopen subsets of X, and V = $\{V_n: n < \omega\}$ is a covering of Y\B by disjoint non-empty clopen subsets of Y. Then $\{(U_n, V_n): n < \omega\}$ is a Knaster-Reichbach covering, or KR-covering, for $(X \land A, Y \land B, h)$ if, whenever $h_n: U_n \rightarrow V_n$ is a bijection for each $n < \omega$, the combination mapping $\tilde{h} = h \cup \bigcup_{n=0}^{\infty} h_n: X \rightarrow Y$ is continuous in points of A, and \tilde{h}^{-1} is continuous in points of B.

Note that we do not require the bijections h_n to exist. Objects similar to those defined above were used by Knaster and Reichbach in [25] to prove some theorems on extensions of homeomorphisms; their technique has afterwards been used by various authors, e.g. Pollard [46], Ravdin [47], and Gutek [20]. The following lemma is a slight generalization of Knaster and Reichbach's theorem.

3.2.2 LEMMA: Let X and Y be zero-dimensional spaces, and let A and B be non-empty closed nowhere dense subspaces of X and Y, respectively. If h: $A \rightarrow B$ is a homeomorphism, then there exists a KR-covering for (X\A,Y\B, h).

Proof: Embed X densely in a compact zero-dimensional space K. For each $x \in K \setminus \overline{A}$, let D_x be a clopen neighborhood of x such that $diam(D_x) < \frac{1}{2}d(x,\overline{A})$; let $\{D_i: i < \omega\}$ be a countable subcovering of $\{D_x: x \in K \setminus \overline{A}\}$, and enumerate the non-empty elements of $\{D_i \setminus (U_{j \le i} D_j): i < \omega\}$ as $\{\widetilde{U}_i: i < \omega\}$ (indeed, this set is infinite since \overline{A} is nowhere dense in K). Note that $diam(\widetilde{U}_i) < d(\widetilde{U}_i, \overline{A})$ for each $i < \omega$. Also, since $\{x \in K: d(x, \overline{A}) \ge \varepsilon\}$ is compact,

30

 $d(\widetilde{U}_{i},\overline{A}) \rightarrow 0$ (i $\rightarrow \infty$). Put $U_{i} = \widetilde{U}_{i} \cap X$, then $\{U_{i}: i < \omega\}$ is a covering of X\A by non-empty pairwise disjoint clopen subsets of X satisfying diam(U_{i}) $< d(U_{i},A) \rightarrow 0$ (i $\rightarrow \infty$). Similarly, let $\{V_{i}: i < \omega\}$ be a covering of Y\B by non-empty pairwise disjoint clopen subsets of Y satisfying diam(V_{i}) $< d(V_{i},B) \rightarrow 0$ (i $\rightarrow \infty$). Now define bijections $\rho,\sigma: \omega \rightarrow \omega$, and points $a_{n} \in A$, $b_{n} = h(a_{n}) \in B$, such that

$$\begin{array}{rll} \text{if } n & \text{is even: } d(\textbf{U}_{\rho(n)},\textbf{a}_n) < 2d(\textbf{U}_{\rho(n)},\textbf{A}); \\ & \textbf{V}_{\sigma(n)} \subset B(\textbf{b}_n,d(\textbf{U}_{\rho(n)},\textbf{A})); \\ \text{if } n & \text{is odd: } d(\textbf{V}_{\sigma(n)},\textbf{b}_n) < 2d(\textbf{V}_{\sigma(n)},\textbf{B}); \\ & \textbf{U}_{\rho(n)} \subset B(\textbf{a}_n,d(\textbf{V}_{\sigma(n)},\textbf{B})), \end{array}$$

as follows:

Suppose a_k , b_k , $\rho(k)$, $\sigma(k)$ have been defined for k < n; if n is even, let $\rho(n) = \min \omega \setminus \{\rho(k): k < n\}$; let $a_n \in A$ be such that $d(U_{\rho(n)}, a_n) < 2d(U_{\rho(n)}, A)$, and let $b_n = h(a_n)$. Put $\varepsilon = \frac{1}{2}d(U_{\rho(n)}, A)$. Since $d(V_i, B) > 0$ for each $i < \omega$, and since B is nowhere dense in Y, $B(b_n, \varepsilon) \cap V_i \neq \emptyset$ for infinitely many $i < \omega$. Let $\sigma(n) \in \omega \setminus \{\sigma(k): k < n\}$ be such that $V_{\sigma(n)} \cap B(b_n, \varepsilon) \neq \emptyset$ and $diam(V_{\sigma(n)}) < \varepsilon$. Then $V_{\sigma(n)} \subset B(b_n, 2\varepsilon)$. If n is odd, let $\sigma(n) = \min \omega \setminus \{\sigma(k): k < n\}$, and proceed as above. We claim that $\{(U_{\rho(n)}, V_{\sigma(n)}): n < \omega\}$ is a KR-covering for $(X \setminus A, Y \setminus B, h)$. Indeed, let $h_n: U_n \rightarrow V_n$ be a bijection, and let $\tilde{h} = h \cup \bigcup_{n=0}^{\infty} h_n$. Suppose that $a \in A$, and $x_n \rightarrow a$. Since $\tilde{h} \mid A: A \approx B$, we may assume that $x_n \in U_{\rho(i_n)}$ for some $i_n < \omega$, for all $n < \omega$. Since each U_i is clopen in X, we have $\rho(i_n) \rightarrow \infty$, hence $diam(U_{\rho(i_n)}), d(U_{\rho(i_n)}, A), diam(V_{\sigma(i_n)}), d(V_{\sigma(i_n)}, B) \rightarrow 0$. Now for each $n < \omega$,

$$d(a_{i_n},a) \leq d(a_{i_n},x_n) + d(x_n,a) \leq d(a_{i_n},U_{\rho(i_n)}) + diam(U_{\rho(i_n)}) + d(x_n,a)$$

 $\leq \max[d(V_{\sigma(i_n)}, B), 2d(U_{\rho(i_n)}, A)] + \operatorname{diam}(U_{\rho(i_n)}) + d(x_n, a) \rightarrow 0.$

So
$$a_{i_n} \rightarrow a$$
, hence $\tilde{h}(a_{i_n}) = b_{i_n} \rightarrow \tilde{h}(a)$. Hence,

$$\begin{aligned} d(\widetilde{h}(x_{n}),\widetilde{h}(a)) &\leq d(\widetilde{h}(x_{n}),b_{i_{n}}) + d(b_{i_{n}},\widetilde{h}(a)) \\ &\leq d(b_{i_{n}},V_{\sigma(i_{n})}) + diam(V_{\sigma(i_{n})}) + d(b_{i_{n}},\widetilde{h}(a)) \\ &\leq \max[d(U_{\rho(i_{n})},A),2d(V_{\sigma(i_{n})},B)] + diam(V_{\sigma(i_{n})}) + \\ &+ d(b_{i_{n}},\widetilde{h}(a)) \rightarrow 0. \end{aligned}$$

So $\widetilde{h}(x_n) \rightarrow \widetilde{h}(a)$. Similarly, if $b \in B$, $y_n \rightarrow b$, then $\widetilde{h}^{-1}(y_n) \rightarrow \widetilde{h}^{-1}(b)$.

For the sake of completeness, we mention the following corollary:

3.2.3 COROLLARY (van Mill [38]): Let X be strongly homogeneous and zerodimensional, and let A and B be closed nowhere dense subspaces of X. If h: $A \rightarrow B$ is a homeomorphism, then there exists a homeomorphism \tilde{h} : X \rightarrow X extending h.

Proof: If $A = \emptyset$, then there is nothing to prove. If $A \neq \emptyset$, then by the lemma, there is a KR-covering $\{(U_n, V_n): n < \omega\}$ for $(X \setminus A, X \setminus B, h)$, and by strong homogeneity, there is a homeomorphism $h_n: U_n \rightarrow V_n$ for each $n < \omega$. Now $\tilde{h} = h \cup \bigcup_{n=0}^{\infty} h_n$ is as required.

Remark: Taking A to be a one-point set, we obtain the result of 1.9.1. In fact, in that proof, $\{(U'_n, V'_n): n < \omega\}$ is a KR-covering for $(X \setminus \{x\}, X \setminus \{y\}, g)$, where g: $\{x\} \rightarrow \{y\}$.

We now come to the first of the main results of this section.

3.2.4 THEOREM (Ostrovskii [44]): Let A be a strongly homogeneous, zerodimensional space, and suppose that $X = \bigcup_{i=0}^{\infty} X_i$, where each X_i is closed and nowhere dense in X, and $X_i \approx A$ for each i. Then $X \approx Q \times A$.

Proof: If $A = \emptyset$, then there is nothing to prove, so suppose that $A \neq \emptyset$. Let $Y = Q \times A$; put $Q = \{q_i: i < \omega\}$, and let $Y_i = \{q_i\} \times A$. Note that for each i, Y_i is closed and nowhere dense in Y, and that $Y_i \approx A$. Recall that $M = \omega^{<\omega}$, see 1.4. For each $s \in M$, we will define collections $U(s) = \{U(s^n): n < \omega\}$ of disjoint clopen subsets of X, $V(s) = \{V(s^n):$ $n < \omega\}$ of disjoint clopen subsets of Y, closed nowhere dense subsets D(s) of U(s), E(s) of V(s), and homeomorphisms $h(s): D(s) \rightarrow E(s)$, such that

- (1) $D(\emptyset) = X_0$, $E(\emptyset) = Y_0$, $U(\emptyset) = X$, $V(\emptyset) = Y$;
- (2) {(U(s^n),V(s^n)): n < ω} is a KR-covering for (U(s)\D(s), V(s)\E(s),h(s));

(3)
$$X_k \subset U_{|s| \le k} D(s), Y_k \subset U_{|s| \le k} E(s).$$

Define $D(\emptyset)$, $E(\emptyset)$, $U(\emptyset)$, and $V(\emptyset)$ as in (1); then there exists by lemma 3.2.2 a KR-covering $\{(U(n), V(n)): n < \omega\}$ for $(U(\emptyset) \setminus D(\emptyset), V(\emptyset) \setminus E(\emptyset), h(\emptyset))$, where $h(\emptyset): D(\emptyset) \rightarrow E(\emptyset)$ is an arbitrary homeomorphism. Now suppose that U(s), V(s), D(s), E(s), and h(s) have been defined for $|s| \le k$, and fix $s \in M$ with |s| = k. Let

$$k_n = \min\{j: U(s^n) \cap X_j \neq \emptyset\}, \ \ell_n = \min\{j: V(s^n) \cap Y_j \neq \emptyset\},$$

and put

$D(s^n) = U(s^n) \cap X_{k_n}, E(s^n) = V(s^n) \cap Y_{\ell_n}.$

Since A is strongly homogeneous, there exists a homeomorphism $h(s^n)$: $D(s^n) \rightarrow E(s^n)$. Since $D(s^n)$ (resp. $E(s^n)$) is closed and nowhere dense in $U(s^n)$ (resp. $V(s^n)$), and non-empty, there exist by lemma 3.2.2 collections $U(s^n) = \{U(s^n^i): i < \omega\}$ and $V(s^n) = \{V(s^n^i): i < \omega\}$ such that $\{(U(s^n^i), V(s^n^i)): i < \omega\}$ is a KR-covering for $(U(s^n) \setminus D(s^n), V(s^n) \setminus E(s^n), h(s^n))$. It is easily seen that (3) is satisfied. This completes the induction. We claim that $h = \bigcup_{s \in M} h_s: X \approx Y$. Since $D(s_1)$ $\cap D(s_2) = \emptyset$ if $s_1 \neq s_2$, h is well-defined, and by (3), the domain of h is all of X, and the range is all of Y. Since h is clearly bijective, it suffices to show that h and h^{-1} are continuous. So let $x \in X$, say $x \in$ D(s). By (2), it suffices to show that $h[U(s^n)] = V(s^n)$ for each $n < \omega$; but this follows immediately from the observation that $U(s^n) = \bigcup_{t \in M} D(s^n^t)$, $V(s^n) = \bigcup_{t \in M} E(s^n^t)$, and $h[D(s^n^t)] = E(s^n^t)$ for each $t \in M$. The continuity of h^{-1} follows in exactly the same way.

Remark: If the strongly homogeneous space A in the above theorem contains a closed nowhere dense copy of itself, then the result can also be deduced from a theorem of van Mill [38].

Remark: If $X = \{x_i: i < \omega\}$ has no isolated points, then taking $X_i = \{x_i\}$ in the above theorem yields $X \approx Q$, substantiating the claim made in section 2.4. To extract the characterization of $Q \times C$ from the above theorem, let X be zero-dimensional, σ -compact, nowhere countable, and nowhere compact, say $X = \bigcup_{n=0}^{\infty} Y_n$, with Y_n compact and non-empty. Fix $n < \omega$. Since X is nowhere compact, Y_n is nowhere dense in X, so as in the proof of lemma 3.2.2, we can construct a covering $\{U_i: i < \omega\}$ of $X \setminus Y_n$ by non-empty clopen subsets of X satisfying diam $(U_i) < d(U_i, Y_n) \rightarrow 0$ $(i \rightarrow \infty)$. Since X is nowhere countable, each U_i is uncountable, and hence it contains a Cantor set C_i (corollary 2.1.5). Using theorem 2.1.1, it is easily verified that

$$X_n = Y_n \cup \bigcup_{i=0}^{\infty} C_i \approx C_i$$

and that X_n is nowhere dense in X. Thus, $X = \bigcup_{n=0}^{\infty} X_n \approx Q \times C$ by the theorem.

In the proof of our next result, we will use the following modification of a convergence criterion of Anderson [2].

3.2.5 LEMMA: Let X be compact, and for each $n \in N$, let $h_n: X \to X$ be a homeomorphism such that $d(h_{n+1},h_n) < \varepsilon_n = \min\{2^{-n},3^{-n},\min\{\min\{d(h_i(x),h_i(y))\}\}$ $d(x,y) \ge 1/n$: $1 \le i \le n$ }. Then $h = \lim_{n \to \infty} h_n$ is an autohomeomorphism of X.

Proof: Since $d(h_{n+1},h_n) < 2^{-n}$, for each $x \in X$ we have that $(h_n(x))_{n \in N}$ is a Cauchy sequence in X, and thus it converges to h(x) ϵ X. Since the convergence is uniform, h is continuous, and hence it is closed since X is compact. So it suffices to show that h is a bijection. Assume that h is not surjective, say $y \in X \setminus h[X]$. Since h[X] is closed in X, d(y,h[X]) = $\varepsilon > 0$. Let $k \in N$ be such that $2^{-k} < \frac{1}{2}\varepsilon$, and put $x = h_{t_{c}}^{-1}(y)$; then

a contradiction. So h is onto. Now suppose that x,y \in X, x \neq y, and let $n \in N$ be such that $d(x,y) \ge 1/n$. Let $\varepsilon = d(h_n(x),h_n(y)) > 0$. Then for each $k \ge n$, $d(h_{k+1}, h_k) < 3^{-k} \cdot \varepsilon$ since $1 \le n \le k$ and $d(x, y) \ge 1/n \ge 1/k$; thus, $d(h_n(x), h(x)) < \sum_{k=n}^{\infty} 3^{-k} \cdot \varepsilon \le \frac{1}{2}\varepsilon$, and similarly $d(h_n(y), h(y)) < \frac{1}{2}\varepsilon$. So

$$d(h(x),h(y)) \ge d(h_n(x),h_n(y)) - d(h_n(x),h(x)) - d(h_n(y),h(y)) > 0,$$

i.e. $h(x) \neq h(y)$.

3

υ c

3.2.6 THEOREM: Let A be a zero-dimensional, strongly homogeneous space,
and suppose that the Cantor set is homogeneous with respect to dense copies
of A if
$$|A| > 1$$
. Let X and Y be dense subsets of C such that X =
 $U_{i=0}^{\infty} X_i, Y = U_{i=0}^{\infty} Y_i, X_i$ is closed and nowhere dense in X, and Y_i is
closed and nowhere dense in Y. If

(1)
$$X_i \approx A \approx Y_i$$
 for each $i < \omega$,
or (2) $\overline{X}_i \setminus X_i \approx A \approx \overline{Y}_i \setminus Y_i$ and $\overline{X}_i \setminus X_i$ is dense in $\overline{X}_i, \overline{Y}_i \setminus Y_i$ is
dense in \overline{Y}_i for each $i < \omega$,

then there exists a homeomorphism $h: C \rightarrow C$ such that h[X] = Y.

Proof: Let U be a clopen subset of C such that $U \cap X_i \neq \emptyset$. Then in case (1), $\mathbb{U} \cap \mathbb{X}_i \approx \mathbb{A}$, and in case (2), $(\overline{\mathbb{U} \cap \mathbb{X}_i}) \setminus (\mathbb{U} \cap \mathbb{X}_i) = \mathbb{U} \cap (\overline{\mathbb{X}}_i \setminus \mathbb{X}_i) \approx \mathbb{A}$, since A is strongly homogeneous. In case (2), also $(\overline{(U \cap X_i)})(U \cap X_i) = (U \cap X_i)$ $(\overline{X}_{i} \setminus X_{i}))^{-} = \overline{U \cap X}_{i}$ since $\overline{X}_{i} \setminus X_{i}$ is dense in \overline{X}_{i} . Similarly, if V is a clopen subset of C such that $V \cap Y_i \neq \emptyset$, then in case (1), $V \cap Y_i \approx A$,

and in case (2), $(\overline{V \cap Y}_{j}) \setminus (V \cap Y_{j}) \approx A$ is dense in $\overline{V \cap Y}_{j}$. In both cases, it follows that there exists a homeomorphism $f: \overline{U \cap X}_i \rightarrow \overline{V \cap Y}_i$ such that $f[U \cap X_i] = V \cap Y_i$ (this is trivial if |A| = 1: then only case (1) can occur; if |A| > 1, it follows from the fact that $\overline{U \cap X}_i \approx C \approx \overline{V \cap Y}_i$ is homogeneous with respect to dense copies of A).

For each $s \in M$, we will now construct collections $U(s) = \{U(s^n): n < \omega\}$, $V(s) = \{V(s^n): n < \omega\}$ of clopen subsets of C, closed nowhere dense subsets D(s),E(s) of C, and for each $m \in N$ a homeomorphism $h_m: C \to C$ such that if n = |s|, then

- (1) $X_n \subset U_{|t| \le n} D(t)$, $Y_n \subset U_{|t| \le n} E(t)$; (2) $D(s) \subset U(s)$, $E(s) \subset V(s)$, $U(\emptyset) = V(\emptyset) = C$;
- (3) $h_{n+1}[U(s^i)] = V(s^i)$ for each $i < \omega$;
- (4) $h_{n+1}[D(s)] = E(s), h_{n+1}[D(s) \cap X] = E(s) \cap Y;$
- (5) if n > 0, then $h_{n+1} | U_{|t| < n} D(t) = h_n | U_{|t| < n} D(t)$;
 - (6) {(U(s^i),V(s^i)): $i < \omega$ } is a KR-covering for (U(s)\D(s), $V(s) \setminus E(s), h_{n+1}(D(s));$
 - (7) if $V \in V(s)$, then diam(V) < ε_{n+1} , where ε_{n+1} is as in 1emma 3.2.5.

Suppose that this has been done; we claim that $d(h_n, h_{n+1}) < \varepsilon_n$ and that $\lim_{n\to\infty}h_n = h: C \approx C \text{ has the property that } h[X] = Y.$ If n > 0, and $x \in U_{|t| < n} D(t)$, then by (5), $h_{n+1}(x) = h_n(x)$, so $d(h_n(x))$, $h_{n+1}(x) < \varepsilon_n$. If $x \notin U_{|t| \le n} D(t)$, then $x \notin D(\emptyset)$, hence $x \in U(m)$ for some m < ω , and proceeding inductively, using (6), x \in U(s^i) for some s ϵM with |s| = n-1, and some $i < \omega$, so by (3), $h_n(x) \epsilon V(s^i)$; if $x \epsilon$ D(s^i), then by (4), $h_{n+1}(x) \in E(s^i) \subset V(s^i)$ by (2), and if $x \notin D(s^i)$, then $x \in U(s^i_j)$ for some $j < \omega$, hence by (3), $h_{n+1}(x) \in V(s^i_j) \subset U(s^i_j)$ $V(s^i)$ by (6). Thus, $\{h_n(x), h_{n+1}(x)\} \subset V(s^i) \in V(s)$, |s| = n-1, hence by (7), $d(h_n(x), h_{n+1}(x)) < \varepsilon_n$. So, by lemma 3.2.5, $\lim_{n \to \infty} h_n$ is an autohomeomorphism of C. Suppose that $x \in X_n$; by (1), $x \in D(s)$ for some $s \in M$ with $|s| \le n$, hence by (4), $h_{|s|+1}(x) \in Y$; by (5), $h_m(x) = h_{|s|+1}(x)$ for each m > |s|+1, so $h(x) = h_{|s|+1}(x)$, i.e. $h(x) \in Y$. Finally, if $y \in Y_n$, then $y \in E(s)$ for some $s \in M$ with $|s| \le n$, hence $y = h_{|s|+1}(x)$ for some $x \in D(s) \cap X$; as above $h(x) = h_{|s|+1}(x)$, so $y \in h[X]$. We will now show how to carry out the inductive construction. First, put $U(\emptyset) = V(\emptyset) = C$, $D(\emptyset) = \overline{X}_0$, $E(\emptyset) = \overline{Y}_0$, and let $\widetilde{h}: D(\emptyset) \to E(\emptyset)$ be a homeomorphism such that $\widetilde{h}[X_0] = Y_0$ (see the remarks at the begin of this proof). Let $R = \{R_i: i < \omega\}, S = \{S_i: i < \omega\}$ be such that $\{(R_i, S_i): i < \omega\}$ is a

KR-covering for $(C \setminus D(\emptyset), C \setminus E(\emptyset), \widetilde{h})$; this is possible by lemma 3.2.2. Since $R_i \approx C \approx S_i$, there exists a homeomorphism $h^i: R_i \rightarrow S_i$. Put

$$h_1 = \tilde{h} \cup \bigcup_{i=0}^{\infty} h^i \colon C \approx C.$$

For each $i < \omega$, let $\{S(i,0),\ldots,S(i,n_i)\}$ be a disjoint clopen covering of S_i by sets of diameter less than ε_1 (as in (7)), and for $i < \omega, k \le n_i$, put $R(i,k) = h_1^{-1}[S(i,k)]$; then $U(\emptyset) = \{R(i,k): i < \omega, k \le n_i\}, V(\emptyset) = \{S(i,k): i < \omega, k \le n_i\}$ can be enumerated as $\{U(n): n < \omega\}, \{V(n): n < \omega\}$ such that $\{(U(n),V(n)): n < \omega\}$ is a KR-covering for $(C \setminus D(\emptyset), C \setminus E(\emptyset), h_1 \mid D(\emptyset))$ satisfying $h_1[U(n)] = V(n)$.

Now suppose that D(s), E(s), U(s), V(s), and h_m have been constructed for $|s| < n (\ge 1)$ and $m \le n$. Fix $s \in M$ with |s| = n-1. Let

$$k_i = \min\{j: U(s^i) \cap X_j \neq \emptyset\}, \ell_i = \min\{j: V(s^i) \cap Y_j \neq \emptyset\},$$

and put

$$D(s^i) = \overline{U(s^i) \cap X_{k_i}}, E(s^i) = \overline{V(s^i) \cap Y_{\ell_i}}$$

Let $\tilde{h}(s^{i}): D(s^{i}) \rightarrow E(s^{i})$ be a homeomorphism such that $\tilde{h}(s^{i})[D(s^{i}) \cap X] = E(s^{i}) \cap Y$. Choose $R(s^{i}) = \{R(s^{i},j): j < \omega\}$, $S(s^{i}) = \{S(s^{i},j): j < \omega\}$ such that $\{(R(s^{i},j),S(s^{i},j)): j < \omega\}$ is a KR-covering for $(U(s^{i})\setminus D(s^{i}),V(s^{i})\setminus E(s^{i}),\tilde{h}(s^{i}))$, and let $h^{j}(s^{i}): R(s^{i},j) \rightarrow S(s^{i},j)$ be a homeomorphism. Put

$$h(s^i) = \tilde{h}(s^i) \cup \bigcup_{j=0}^{\infty} h^j(s^i): U(s^i) \approx V(s^i),$$

and

$$h_{n+1} = U\{h(s^i): |s| = n-1, i < \omega\} \cup h_n |U_{|s| < n} D(s)$$

As above, we can refine each $S(s^i)$ by disjoint clopen sets of diameter less than ε_{n+1} , and obtain $U(s^i) = \{U(s^i^j): j < \omega\}$, $V(s^i) = \{V(s^i^j): j < \omega\}$ such that $\{(U(s^i^j), V(s^i^j)): j < \omega\}$ is a KR-covering for $(U(s^i) \setminus D(s^i), V(s^i) \setminus E(s^i), h_{n+1} \mid D(s^i))$ satisfying $h_{n+1}[U(s^i^j)] = V(s^i^j)$. As in the proof of theorem 3.2.4, it is easily seen that h_{n+1} is a homeomorphism, and that the inductive hypotheses are satisfied.

Taking |A| = 1 in the above theorem, we obtain the following corollary (see e.g. Bennett [3]).

3.2.7 COROLLARY: The Cantor set is homogeneous with respect to dense copies of Q (i.e. C is countable dense homogeneous).

Applying case (1) of theorem 3.2.6 for non-trivial A, we obtain:

3.2.8 THEOREM: Let A be a zero-dimensional, strongly homogeneous space, with |A| > 1. If C is homogeneous with respect to dense copies of A, then C is also homogeneous with respect to dense copies of $Q \times A$.

3.2.9 COROLLARY (van Mill [39]): The Cantor set is homogeneous with respect to dense copies of $Q \times C$.

3.3 Hurewicz-type theorems

In this chapter, we will often be concerned with the construction of closed copies of previously characterized spaces in the space which is yet to be characterized. The usefulness of this idea has already been demonstrated in the second remark following theorem 3.2.4, where the existence of Cantor sets in uncountable σ -compact spaces was used to obtain the characterization of $Q \times C$. In fact, Cantor sets exist in arbitrary uncountable analytic spaces; this is a theorem of Suslin, see Luzin [33]. Another theorem, which better exemplifies the type of result that we are looking for, was proved by Hurewicz [23]: he showed that, if a Borel subset A of a compact space X is not complete, then X contains a closed subset K such that K $\cap A \approx Q$, and K\A $\approx P$; clearly, the condition that A is not complete is also necessary, since Q is not complete.

In [49], Saint-Raymond proved a theorem of a similar nature: he showed that a Borel subset A of a compact space X is not the union of a σ -compact subspace and a complete subspace if and only if A contains a closed copy of $Q \times P$; this is also equivalent to X\A not being the intersection of a complete subspace and a σ -compact subspace, i.e. not being strongly σ -complete (see the proof of lemma 3.1.3).

In van Engelen and van Mill [14], the above results were improved, and some more Hurewicz-type theorems were established (here, S and T are the homogeneous complements of $Q \times P$ in C, see section 3.1 and the introduction to this monograph):

3.3.1 THEOREM (van Engelen and van Mill [14]): Let X be a compact space, and let A be a Borel subset of X. (a) A is not complete if and only if X contains a Cantor set K such that $K \cap A \approx Q$ and $K \setminus A \approx P$; (b) A is not complete \cup countable if and only if X contains a Cantor set K such that K \cap A $\approx Q \times C$ and K \setminus A $\approx P$;

(c) A is not strongly σ -complete if and only if X contains a Cantor set K such that K $\cap A \approx T$ and K $A \approx Q \times P$;

(d) A is not strongly σ -complete \cup countable if and only if X contains a Cantor set K such that $K \cap A \approx S$ and $K \setminus A \approx Q \times P$.

Parts (a) and (b) will be proved in this section, (c) and (d) in section 3.4. This theorem, and the other Hurewicz-type theorems that we will use in this chapter, will all be deduced from lemmas 4 and 5 of this section. First, however, we state the following definition and lemma:

3.3.2 DEFINITION: Let P be a topological property. (a) A space X is strongly σ -P if X = $\bigcup_{i=0}^{\infty} X_i$, where each X_i is a closed subspace of X which has property P. (b) P is strongly σ -additive if a space is P whenever it is strongly σ -P.

3.3.3 LEMMA: Let A be a non-empty, compact, nowhere dense subset of a space X, and let $(\varepsilon_n)_{n < \omega}$ be a given sequence of positive real numbers. Then there exists a countable discrete subset $D = \{d_n : n < \omega\}$ of X\A such that $\overline{D} = D \cup A$ and $d(d_n, A) < \varepsilon_n$ for each $n < \omega$.

Proof: For each $i < \omega$, let $\mathcal{D}_i = \{D(i,j): j < n_i\}$ be a collection of open subsets of X of diameter less than 1/(i+1) such that $A \subset U\mathcal{D}_i$, and each $A \cap D(i,j) \neq \emptyset$, say $p(i,j) \in A \cap D(i,j)$. For each $n < \omega$, there exists a unique $i < \omega$ such that $n = (\Sigma_{k < i} n_k) + j$ for some $j < n_i$, and we choose $d_n \in (B(p(i,j), \varepsilon_n) \cap D(i,j)) \setminus A$. Then $D = \{d_n: n < \omega\}$ is as required. \Box

We now come to our first Hurewicz-type result; the statement is rather awkward since we want to apply the lemma in a variety of situations. The technique of proof is inspired by that of Saint-Raymond [49]; see also van Douwen [9], and Tops e and Hoffman-Jørgensen [58], where similar techniques are used.

3.3.4 LEMMA: Let P be a topological property which is closed-hereditary and strongly σ -additive. Let Z be a non-empty strongly homogeneous space, and suppose that if B is a Borel subset of a compact space Y which is not P, then Y contains a compact zero-dimensional subset K such that

38

 $K = \overline{K \cap B}, K \cap B \approx Z$, and $K \setminus B$ is contained in a σ -compact subset G of Y with the property that $G \cap B$ is P. Then if A is a Borel subset of a compact space X which is not $P \cup$ complete, then X contains a Cantor set K such that $\overline{K \cap A} = K$ and $K \cap A \approx Q \times Z$.

Proof: Since A is a Borel subset of X there exists a continuous surjection $\phi: P \to X \setminus A$. Put

$$W = \{x \in P: \text{ there exist a neighborhood } V_x \text{ of } x \text{ in } P, \text{ and a} \\ \sigma \text{-compact subset } E_x \text{ of } X, \text{ such that } \phi[V_x] \subset E_x, \\ \text{ and } E_y \cap A \text{ is } P\}.$$

Then W is open in P, so there exist countably many open V_i in P, and σ -compact E_i in X such that $W = \bigcup_{i=0}^{\infty} V_i$, $\phi[V_i] \subset E_i$, and $E_i \cap A$ is P. Since E_i is σ -compact, we can write $E_i = \bigcup_{j=0}^{\infty} K_j^i$, where each K_j^i is compact; then $E = \bigcup_{i=0}^{\infty} \bigcup_{j=0}^{\infty} K_j^i$ is σ -compact, and since $K_j^i \cap A$ is P, being a closed subspace of $E_i \cap A$, $E \cap A = \bigcup_{i=0}^{\infty} \bigcup_{j=0}^{\infty} (K_j^i \cap A)$ is strongly σ -P, hence P. Now suppose that $X \setminus A \subset E$; then $A = (E \cap A) \cup (X \setminus E)$ is P \cup complete, a contradiction. Hence,

 $\mathbf{F} = \boldsymbol{P} \setminus \phi^{-1} [\mathbf{E}]$

is non-empty, and complete.

CLAIM 1: If U is non-empty and open in F, $G \subset X$ is σ -compact, and $\phi[U] \subset G$, then $G \cap A$ is not P.

Indeed, suppose $G = \bigcup_{i=0}^{\infty} G_i$, where G_i is compact, and $G \cap A$ is P; then also $G_i \cap A$ is P, and hence $(G \cup E) \cap A = \bigcup_{i=0}^{\infty} (G_i \cap A)$ $\cup \bigcup_{i=0}^{\infty} \bigcup_{j=0}^{\omega} (K_j^i \cap A)$ is strongly σ -P, hence P. Now let U' be open in P such that $U' \cap F = U$; then $\phi[U'] = \phi[U] \cup \phi[U' \setminus U] \subset G \cup E$, and since $U' \notin W$, $(G \cup E) \cap A$ is not P, a contradiction.

From this claim, it follows that if U is any non-empty open subset of F, then $\overline{\phi[U]} \cap A$ is not P, and hence $\overline{\phi[U]}$ contains a compact zero-dimensional subspace K such that $K = \overline{K \cap A}, K \cap A \approx Z$, and K\A is contained in a σ -compact subset G of $\overline{\phi[U]}$ (and hence of X) with the property that $G \cap A$ is P.

We will now construct compact zero-dimensional subsets K_s of X, open subsets U_s of X, open subsets W_s of F, and finite collections V_i of open subsets of X, for each $s \in M$ and each $i < \omega$, such that the following hold:

(1)
$$K_{s} = \overline{K_{s} \cap A} \subset \overline{\phi[W_{s}]} \subset U_{s};$$

(2) for each $n < \omega: \overline{U}_{s^{n}} \cap K_{s} = \emptyset;$

- (3) for each $n,m < \omega$: $\overline{U}_{s^n} \cap \overline{U}_{s^m} = \emptyset$ if $n \neq m$;
- (4) for each $n < \omega$: $\operatorname{Cl}_{F}(\overline{W}_{s^{n}}) \subset \overline{W}_{s};$
- (5) for each $n < \omega$: $\overline{U}_{s} \cdot n \in U_{s}$; (6) diam(W_{s}) $\leq 2^{-|s|}$ (here, the diameter is taken with respect to a complete metric for F);
- (7) diam(U_s) $\leq 2^{-\nu(s)}$;
- (8) for each $n < \omega$: $d(K_s, K_{s^n}) \le 2^{1-\nu(s,n)};$
- (9) $K_s \cap A \approx Z$, and $K_s \setminus A \subset G_s$ for some σ -compact subset G_s of X such that $G_{s} \cap A$ is P_{s}
- (10) K_s is nowhere dense in $K_s \cup \bigcup_{n=0}^{\infty} K_{s^n}$;
- (11) $B_i = U\{K_s: |s| \le i\}$ is compact zero-dimensional, and $B_i \subset UV_i$;
- (12) $UV_{i+1} \subset UV_i;$
- (13) for each $V \in V_i$: diam(V) < 1/(i+1);
- (14) $\{\overline{V}: V \in V_i\}$ is pairwise disjoint.

We proceed by induction on |s| and i. First, we put $W_{d} = F$, $U_{d} = X$. Then $\overline{\phi[W_{g}]}$ contains a compact zero-dimensional subspace $K_{g} = \overline{K_{g} \cap A}$ such that $K_{of} \cap A \approx Z$ and $K_{of} \setminus A \subset G_{of}$ for some σ -compact subset G_{of} of X such that $G_{0} \cap A$ is *P*. Put $V_{0} = \{X\}$. Then (1), (9), (11), and (14) are satisfied, and so are (6), (7), and (13), since all metrics are assumed to be bounded by 1 (see 1.5).

Next, suppose that K_s , U_s , W_s , and V_i have been constructed for $|s| \le k$ and $i \leq k$, in accordance with conditions (1) - (14). Fix $s \in M$ with |s|= k.

CLAIM 2: K is nowhere dense in $K_{s} \cup \phi[W_{s}]$.

We distinguish two cases.

<u>Case 1</u>: Let $y \in K_s \cap \phi[W_s]$, say $y = \phi(x)$, with $x \in W_s$, and let U be an open neighborhood of y in $K_{s} \cup \phi[W_{s}]$. By continuity of $\phi: W_s \rightarrow \phi[W_s]$, there exists an open neighborhood V of x in W_s such that $\phi[V] \subset U \cap \phi[W_s]$. Suppose that $U \cap \phi[W_s] \subset K_s$. Then $\phi[V] \subset K_s \land A \subset G_s$, and G_s is a σ -compact subset of X such that $G_s \cap A$ is P (apply (9)). Since V is a non-empty open subset of F, this contradicts claim 1, and hence $U \setminus K_s \neq \emptyset$.

<u>Case 2</u>: Let $y \in K_{s} \setminus \phi[W_{s}]$, and let U be an open neighborhood of y in $K_{s} \cup \phi[W_{s}]$. Since $K_{s} \subset \overline{\phi[W_{s}]}$ by (1), $U \cap \phi[W_{s}] \neq \emptyset$, say $z \in U \cap \phi[W_s]$. If $z \notin K_s$, then we are done, and if $z \in K_s$, then $z \in K_{c} \cap \phi[W_{c}]$, and case 1 applies. This proves the claim.

By this claim, we can apply lemma 3.3.3 to obtain a countable discrete sub-

set $D_s = \{y_{s\uparrow n}: n < \omega\}$ of $\phi[W_s]\setminus K_s$ such that $\overline{D}_s = D_s \cup K_s$, and $d(y_{s\uparrow n}, K_s) \le 2^{-\nu(s\uparrow n)}$ for each $n < \omega$. Of course, we may assume that $D_s \subset UV_k$. Let $U_{s\uparrow n}$ be a neighborhood of $y_{s\uparrow n}$ in X such that $\overline{U}_{s\uparrow n} \cap K_s = \emptyset$, $\overline{U}_{s\uparrow n} \cap \overline{U}_{s\uparrow m} = \emptyset$ if $n \neq m$, $diam(U_{s\uparrow n}) \le 2^{-\nu(s\uparrow n)}$, and $\overline{U}_{s\uparrow n} \subset U_s \cap UV_k$ for each $n, m < \omega$. Since $y_{s\uparrow n} \in \phi[W_s]$, $y_{s\uparrow n} = \phi(x_{s\uparrow n})$ for some $x_{s\uparrow n} \in W_s$; hence, there is an open neighborhood $W_{s\uparrow n}$ of $x_{s\uparrow n}$ in F such that $Cl_F(W_{s\uparrow n}) \subset W_s$, $diam(W_{s\uparrow n}) \le 2^{-|s|-1}$, and $\overline{\phi[W_{s\uparrow n}]} \subset U_{s\uparrow n}$. Then $\overline{\phi[W_{s\uparrow n}]}$ contains a compact zero-dimensional subset $K_{s\uparrow n}$ such that $K_{s\uparrow n} = \overline{K_{s\uparrow n}} \cap A$, $K_{s\uparrow n} \cap A \approx Z$, and $K_{s\uparrow n} \setminus A \subset G_{s\uparrow n}$ for some σ -compact subset $G_{s\uparrow n}$ of X such that $G_{s\uparrow n} \cap A$ is P. Then (1) - (7) and (9) are satisfied. For (8), note that

$$d(K_{s},K_{s^{n}}) \leq d(K_{s},U_{s^{n}}) + diam(U_{s^{n}}) \leq d(K_{s},y_{s^{n}}) + 2^{-\nu(s^{n})}$$
$$\leq 2^{1-\nu(s^{n})}.$$

To prove (10), take $x \in K_s$, and $\varepsilon > 0$. Choose $n < \omega$ so large that $2^{-\nu(s^n)} < \frac{1}{2}\varepsilon$. Since $\overline{D}_s = D_s \cup K_s$, $y_{s^m} \in B(x, \frac{1}{2}\varepsilon)$ for some m > n; and since $y_{s^m} \in U_{s^m}$ and diam $(U_{s^m}) \le 2^{-\nu(s^m)} \le 2^{-\nu(s^n)} < \frac{1}{2}\varepsilon$, we have $U_{s^m} \subset B(x,\varepsilon)$, so $K_{s^m} \subset B(x,\varepsilon)$. By (1) and (2), $K_{s^m} \cap K_s = \emptyset$, so $B(x,\varepsilon) \cap (K_s \cup \bigcup_{n=0}^{\infty} K_{s^n}) \setminus K_s \supset K_{s^m} \neq \emptyset$.

Since $B_{k+1} = U\{K_s: |s| \le k+1\}$ is a countable union of closed zero-dimensional subspaces, it is zero-dimensional (see e.g. Engelking [15] 7.2.1); so in order to establish (11), we must show that B_{k+1} is closed in X. For each $\varepsilon > 0$, put

$$B_k^{\varepsilon} = \{x \in X: d(x,B_k) \leq \varepsilon\},\$$

and

$$M_{\mathbf{k}}^{\varepsilon} = \{ \mathbf{s} \in M : |\mathbf{s}| = \mathbf{k}+1, \mathbf{K}_{\mathbf{s}} \notin \mathbf{B}_{\mathbf{k}}^{\varepsilon} \};$$

then each B_k^{ε} is compact, and M_k^{ε} is finite by (1), (7), and (8). Since B_k is compact by the inductive hypothesis, we have

$$B_{k} = \bigcap_{\varepsilon > 0} B_{k}^{\varepsilon},$$

and

$$\mathsf{B}_{k+1} = \mathsf{O}_{\varepsilon > 0}(\mathsf{B}_{k}^{\varepsilon} \cup \mathsf{U}\{\mathsf{K}_{s}: s \in \mathsf{M}_{k}^{\varepsilon}\})$$

is compact, being the intersection of compacta.

To complete the induction, let V be a covering of B_{k+1} by disjoint clopen subsets of B_{k+1} of diameter less than 1/(k+2). By normality, there exist open subsets V' of X (for $V \in V$), which can be taken to have diameter less than 1/(k+2), such that $V' \cap B_{k+1} = V$ and $V' = \{V': V \in V\}$

is pairwise disjoint; also, since $B_{k+1} \subset UV_k$, we can assume that $UV' \subset UV_k$. Again by normality, we can shrink V' to obtain a covering V_{k+1} of B_{k+1} by open subsets of X, satisfying (11) - (14). Now put

$$K = (U_{i=0}^{\infty} B_{i})^{-};$$

clearly, K is compact, and by (10), no point of B_i is isolated in B_{i+1} , so K contains no isolated points. Also, from (11) - (14) it follows that, for each $i < \omega$, $K \subset \bigcup\{\overline{V}: V \in V_i\}$, which is a pairwise disjoint finite closed covering of K by sets of diameter less than 1/(i+1); hence, $\{\overline{V} \cap K:$ $V \in V_i$, $i < \omega\}$ is a clopen basis for K, so K is zero-dimensional. Thus, $K \approx C$ by theorem 2.1.1.

We claim that $K \cap A = \bigcup_{i=0}^{\infty} (B_i \cap A)$. Suppose to the contrary that $x \in (K \cap A) \setminus (\bigcup_{i=0}^{\infty} B_i)$, and fix $i < \omega$. Since $x \notin B_i$, also $x \notin B_i^{\varepsilon}$ for some $\varepsilon > 0$. From (1) and (4) it follows that $\bigcup_{j=0}^{\infty} B_j \subset B_i \cup \bigcup_{|s|=i} \overline{\phi[W_s]}$, and from (1), (7), and (8) that $\overline{\phi[W_s]} \subset B_i^{\varepsilon}$ for all but finitely many $s \in M$ with |s| = i. Hence for some finite $M_0 \subset \{s \in M: |s| = i\}$, we have

$$K \subset B_i^{\varepsilon} \cup \bigcup_{s \in M_0} \overline{\phi[W_s]}.$$

Then $x \in \overline{\phi[W_s]}$ for some $s \in M_0$, and this s is unique with |s| = i by (1), (3), and (5) (or trivially, if i = 0). So by (4), there exists $\sigma \in \omega^{\omega}$ such that $x \in \bigcap_{s < \sigma} \overline{\phi[W_s]}$, which is a one-point set by (1) and (7). Also, $\bigcap_{s < \sigma} \overline{W}_s = \bigcap_{s < \sigma} W_s$ is a one-point set by (6), and by completeness of the metric on F. Hence, if $z \in \bigcap_{s < \sigma} W_s$, then $\phi(z) = x$, so $x \in \phi[P] = X \setminus A$, a contradiction since $x \in K \cap A$.

Thus, $K \cap A = \bigcup_{i=0}^{\infty} (B_i \cap A) = \bigcup_{s \in M} (K_s \cap A)$. Clearly, $K_s \cap A$ is closed in $K \cap A$, and by (1) and (10), $K_s \cap A$ is nowhere dense in $K \cap A$ for each s ϵ M. So by (9) and theorem 3.2.4, $K \cap A \approx \mathbf{Q} \times \mathbb{Z}$. Finally, using (1),

$$K \supset \overline{K \cap A} = (\bigcup_{s \in M} (\overline{K_s \cap A}))^{-} = (\bigcup_{s \in M} K_s)^{-} = K,$$
$$K = \overline{K \cap A}.$$

Before stating our second Hurewicz-type result, we show how to deduce theorem 3.3.1(a) and (b) from the above lemma.

Proof of theorem 3.3.1(a): Let a space have the property P if and only if it is empty, and let Z be a one-point space. Then clearly P is closedhereditary and strongly σ -additive, and if Y is compact and B is a non-

so

empty subset of Y, say $y \in B$, then $K = \{y\}$ and $G = \emptyset$ satisfy the requirements. So if A in X is not complete, then it is not $P \cup$ complete, so by the lemma, there is a closed copy Q of Q in A such that $K = \overline{Q} \approx C$; then $K \setminus A = K \setminus Q \approx P$, using theorem 2.3.1.

Proof of theorem 3.3.1(b): Let a space have the property P if and only if it is countable, and let Z = C. Again, P is as required, and if B is an uncountable Borel subset of the compact space Y, then B contains a Cantor set K by Suslin's theorem. So if A in X is not countable \cup complete, then it is not $P \cup$ complete, so there is a closed copy H of $Q \times C$ in A such that $K = \overline{H} \approx C$; again by theorem 2.3.1, $K \setminus A = K \setminus H \approx P$.

We only outline the proof of the following lemma, since it is entirely similar to that of lemma 3.3.4.

3.3.5 LEMMA: Let P be as in lemma 3.3.4, and suppose that for each $i < \omega$, Y_i and Z_i are non-compact strongly homogeneous spaces. Furthermore, assume that if B is a Borel subset of a compact space Y which is not P, then Y contains, for each $i < \omega$, a Cantor set K_i such that K_i = $\overline{K_i \setminus B}$, and

(I)
$$K_i \setminus B \approx Z_i;$$

(II) $K_i \cap B \approx Y_i is P.$

Then if A is a Borel subset of a compact space X which is not P, then X contains Cantor sets B_i , for $i < \omega$, such that $B_i = \overline{B_i \setminus A}$, $K = (\bigcup_{i=0}^{\infty} B_i)^{-1} \approx C$, and for each $i < \omega$,

> (i) B_i is nowhere dense in K; (ii) $K \setminus A = \bigcup_{i=0}^{\infty} (B_i \setminus A);$ (iii) $B_0 \setminus A \approx Z_0$, and $(B_{i+1} \setminus B_i) \setminus A \approx Z_{i+1};$ (iv) $B_0 \cap A \approx Y_0$, and $(B_{i+1} \setminus B_i) \cap A \approx Y_{i+1}.$

Proof: Since A is a Borel subset of X, there exists a continuous surjection $\phi: P \rightarrow A$. Put

$$\begin{split} \mathbb{W} &= \{\mathbf{x} \in P: \text{ there exist a neighborhood } \mathbb{V}_{\mathbf{x}} \text{ of } \mathbf{x} \text{ in } P, \text{ and a} \\ \sigma\text{-compact subset } \mathbb{E}_{\mathbf{x}} \text{ of } \mathbf{X}, \text{ such that } \phi[\mathbb{V}_{\mathbf{x}}] \subset \mathbb{E}_{\mathbf{x}}, \\ \text{ and } \mathbb{E}_{\mathbf{x}} \cap \mathbb{A} \text{ is } P \}. \end{split}$$

Then W is open in P, so there exist countably many open V_i in P, and σ -compact E_i in X, such that $W = \bigcup_{i=0}^{\infty} V_i$, $\phi[V_i] \subset E_i$, and $E_i \cap A$ is P. Put $E = \bigcup_{i=0}^{\infty} E_i$, then it follows as in the proof of lemma 3.3.4 that $E \cap A$ is P. Now suppose that $A \subset E$; then $A = E \cap A$ is P, a contradiction. Hence $F = P \setminus \phi^{-1}[E]$ is non-empty, and complete.

As in lemma 3.3.4, we can show that if U is a non-empty open subset of F, then $\overline{\phi[U]} \cap A$ is not P, and hence that $\overline{\phi[U]}$ contains Cantor sets K_i satisfying (I) and (II). Construct K_s , U_s , W_s , and V_i as above, satisfying (1) - (8), (10) - (14), and

(9')
$$K_{s} \setminus A \approx Z_{|s|}$$
, and $K_{s} \cap A \approx Y_{|s|}$ is *P*.

Since each K_s is a Cantor set, B_i is also a Cantor set, and from (10) it easily follows that B_i is nowhere dense in K. Since this time, ϕ is a mapping onto A, we find that $K \setminus (\bigcup_{i=0}^{\infty} B_i) \subset \phi[P] = A$, so that $K \setminus A = \bigcup_{i=0}^{\infty} (B_i \setminus A)$.

Since $B_{i+1} \setminus B_i = \bigcup\{K_s: |s| = i+1\}$, and since $K_s \subset \bigcup_s$ and $\bigcup_s \cap \bigcup_t = \emptyset$ if s,t $\in M$, |s| = |t|, $s \neq t$ (use (3) and (5)), we have $(B_{i+1} \setminus B_i) \setminus A = \bigcup\{K_s \setminus A: |s| = i+1\} \approx \omega \times Z_{i+1} \approx Z_{i+1}$ since Z_{i+1} is non-compact, strongly homogeneous and zero-dimensional (see 1.9.2). That $(B_{i+1} \setminus B_i) \cap A \approx Y_{i+1}$ follows in exactly the same way.

3.4 The small Borel classes $D_n(\Sigma_2^0)$ for $n \in N$

Recall that if $n < \omega$, then a space is S_n if it is the union of n strongly σ -complete subspaces. In this section, we describe and characterize certain homogeneous Borel sets in the Cantor set that are S_n for some $n < \omega$; in section 3.6 we will see that they are the only ones, apart from those of Chapter 2.

In section 1 of this chapter, we defined topological properties describing subclasses of Δ_3^0 :

X is P_{4k} if and only if X is $S_k \cup \text{complete}$; X is P_{4k+1} if and only if X is S_{k+1} ; X is P_{4k+2}^1 if and only if X is $S_k \cup \text{complete } \cup \text{ countable}$; X is P_{4k+3}^1 if and only if X is $S_{k+1} \cup \text{ countable}$; X is P_{4k+2}^2 if and only if X is $S_k \cup \text{ complete } \cup \sigma\text{-compact}$; X is P_{4k+3}^2 if and only if X is $S_{k+1} \cup \sigma\text{-compact}$.

Note the following equivalences:

44

$$P_{4k+2}^2 \equiv P_{4k} \cup \sigma \text{-compact} \equiv P_{4(k-1)+3}^2 \cup \text{complete};$$

$$P_{4k+3}^2 \equiv P_{4k+1} \cup \sigma \text{-compact}.$$

3.4.1 LEMMA:(a) For each $n < \omega$, and each i $\in \{1,2\}, \ P_n^{(i)}$ is closed-here-ditary.

(b) For each $n < \omega$, $P_n^{(1)}$ is hereditary with respect to G_{δ} -subsets.

Proof:(a) It suffices to show that a closed subset of a strongly σ -complete space is strongly σ -complete. But if $X = \bigcup_{i=0}^{\infty} X_i$, with X_i closed in X and complete, then for a closed $A \subset X$, we have $A = \bigcup_{i=0}^{\infty} (X_i \cap A)$, $X_i \cap A$ is closed in A, and also in X_i , so $X_i \cap A$ is complete. The proof of (b) is similar; P_{4k+2}^2 and P_{4k+3}^2 are not G_{δ} -hereditary since σ -compactness is not. Now

3.4.2 LEMMA: Let X be a topological space.
(a) For each
$$k < \omega$$
, X is P_{4k+1} if and only if X is strongly $\sigma - P_{4k}$.
(b) For each $k < \omega$, and each $i \in \{1,2\}$, X is P_{4k+3}^{i} if and only if X is strongly $\sigma - P_{4k+2}^{i}$.

Proof: The proof is similar to that of lemma 3.1.5; consider X as a subset of the compact space Y.

(a) If X is P_{4k+1} , then by lemma 3.1.4, there exist σ -compact subsets $\{A_m: m < 2(k+1)\}$ of Y such that $X = D_{2(k+1)}(\langle A_m: m < 2(k+1) \rangle)$. Let $A_{2k+1} = \bigcup_{i=0}^{\infty} K_i$, where each K_i is compact; then $K_i \cap X$ is closed in X, and $X = X \cap A_{2k+1} = \bigcup_{i=0}^{\infty} (K_i \cap X)$. Now

$$K_i \cap X = D_{2k}(\langle A_m \cap K_i : m < 2k \rangle) \cup K_i \setminus A_{2k};$$

since $D_{2k}((A_m \cap K_i: m < 2k))$ is S_k (lemma 3.1.4), and $K_i \setminus A_{2k}$ is complete, being a G_{δ} in K_i , we have that $K_i \cap X$ is P_{4k} , so X is strongly σP_{4k} . For the converse, assume that X is strongly σP_{4k} ; then X is certainly strongly σP_{4k+1} , so by lemma 3.1.4, we can write $X = \bigcup_{i=0}^{\infty} X_i$, where $X_i = D_{2(k+1)}((A_m^i: m < 2(k+1)))$, each A_m^i is σ -compact, and X_i is closed in X, say $X_i = K_i \cap X$, with K_i compact in Y. Put $L_i = K_i \setminus (\bigcup_{j < i} K_j)$, and for m < 2(k+1),

$$A_{m} = U_{i=0}^{\infty} (L_{i} \cap A_{m}^{i}).$$

As in the proof of theorem 3.1.5, we obtain that

$$\begin{split} \mathbf{X} &= \mathbf{U}_{i=0}^{\infty}(\mathbf{L}_{i} \cap \mathbf{X}_{i}) = \mathbf{D}_{2(k+1)}(\langle \mathbf{A}_{m}: m < 2(k+1) \rangle), \\ \text{so A is } S_{k+1}, \text{ i.e. A is } P_{4k+1}. \end{split}$$

(b) Let X be P_{4k+3}^{i} , say X = $D_{2(k+1)}((A_{m}: m < 2(k+1))) \cup B$, where B is countable or σ -compact, depending on whether i = 1 or i = 2. Let $A_{2k+1} \cup B = \bigcup_{j=0}^{\infty} K_{j}$, where K_{j} is compact. Then

$$K_{j} \cap X = D_{2k} (\langle A_{m} \cap K_{j} : m < 2k \rangle) \cup K_{j} \setminus A_{2k} \cup (K_{j} \cap B)$$

is $S_k \cup \text{complete } \cup \text{ countable (resp. } \sigma\text{-compact), i.e. } K_j \cap X \text{ is } P_{4k+2}^i; \text{ so } X = \bigcup_{j=0}^{\infty} (K_j \cap x) \text{ is strongly } \sigma\text{-}P_{4k+2}^i.$ Conversely, if X is strongly $\sigma\text{-}P_{4k+2}^i$, then X is strongly $\sigma\text{-}P_{4k+3}^i$, so we can write $X = \bigcup_{j=0}^{\infty} X_j$, where $X_j = D_{2(k+1)}((A_m^j: m < 2(k+1))) \cup B_j$ is closed in X, B_j is countable or $\sigma\text{-compact. Let } K_j$ be compact such that $K_j \cap X = X_j$, and put $L_j = K_j \setminus (\bigcup_{n < j} K_n)$. Then if $A_m = \bigcup_{j=0}^{\infty} (L_j \cap A_m^j)$

for m < 2(k+1), we have

$$X = D_{2(k+1)} (\langle A_m: m < 2(k+1) \rangle) \cup \bigcup_{j=0}^{\infty} B_j$$

is P_{4k+3}^i .

3.4.3 COROLLARY: For odd $n < \omega$, and each $i \in \{1,2\}$, $P_n^{(i)}$ is a strongly σ -additive property.

3.4.4 LEMMA: Let X be compact, and let A be a subset of X. Then for
each
$$k < \omega$$
,
(a) A is P_{4k} if and only if X\A is $P_{4(k-1)+3}^2$.
(b) A is P_{4k+1} if and only if X\A is P_{4k+2}^2 .
Proof:(a) This is trivial if $k = 0$, so suppose that $k > 0$. If A is P_{4k}
is $S_k \cup$ complete, then $A = D_{2k}(\langle A_i : i < 2k \rangle) \cup B$, where each A_i is σ -compact and B is complete, i.e. X\B is σ -compact. Hence

$$X \setminus A = (A_0 \cap X \setminus B) \cup D_{2(k-1)} ((A_{i+1} \cap X \setminus B: i < 2(k-1))) \cup (X \setminus B) \setminus A_{2k-1}$$

is σ -compact $\cup S_{k-1} \cup S_1$ is $P_4^2(k-1)+3$. Conversely, if X\A is $P_4^2(k-1)+3$ is $S_k \cup \sigma$ -compact, then X\A can be written as A above, but with B being σ -compact; hence

$$A = (A_0 \setminus B) \cup D_{2(k-1)}((A_{i+1} \cup B: i < 2(k-1))) \cup X \setminus (B \cup A_{2k-1})$$

is $S_1 \cup S_{k-1} \cup$ complete is P_{4k} .

(b) The case k = 0 follows from the observation that A is the difference of two σ -compacta if and only if it is the intersection of a σ -compact space and a complete space. So suppose that k > 0. If A is P_{4k+1} is S_{k+1} ,

then $A = D_{2(k+1)} ((A_i: i < 2(k+1)))$, so

$$X \setminus A = A_0 \cup D_{2k} (\langle A_{i+1} : i < 2k \rangle) \cup X \setminus A_{2k+1}$$

is σ -compact $\cup S_k \cup$ complete is P_{4k+2}^2 . Conversely, if X\A is P_{4k+2}^2 is $S_k \cup$ complete $\cup \sigma$ -compact, then X\A = $D_{2k}(\langle A_i: i < 2k \rangle) \cup G \cup B$, where G is complete and B is σ -compact; hence

$$A = (A_0 \cap X \setminus G) \setminus B \cup D_{2(k-1)} ((A_{i+1} \cup B) \cap X \setminus G: i < 2(k-1))) \cup (X \setminus G) \setminus (B \cup A_{2k-1})$$

is $S_1 \cup S_{k-1} \cup S_1$ is S_{k+1} is P_{4k+1} .

3.4.5 LEMMA: The following implications hold between the properties $P_n^{(i)}$:

Corresponding to each property $P_n^{(i)}$ we now define a class of spaces $X_n^{(i)}$, and show that, up to homeomorphism, each $X_n^{(i)}$ contains exactly one element.

3.4.6 DEFINITION: (a)
$$X_{-1}^1 = \{Q\}; X_{-1}^2 = \{Q \times C\}; X_0 = \{P\}.$$

(b) Let X be zero-dimensional; then for each $k < \omega$,
 $X \in X_{4(k+1)}$ if and only if X is $P_{4(k+1)}$, nowhere P_{4k+2}^2 , nowhere $P_{4k+1};$
 $X \in X_{4k+1}$ if and only if X is P_{4k+1} , nowhere P_{4k} , nowhere $P_{4(k-1)+3}^2;$
 $X \in X_{4k+2}^1$ if and only if X is P_{4k+2}^1 , nowhere P_{4k}^2 , nowhere $P_{4(k-1)+3}^2;$
 $X \in X_{4k+2}^2$ if and only if X is P_{4k+2}^1 , nowhere P_{4k+2}^1 , nowhere $P_{4(k-1)+3}^2;$
 $X \in X_{4k+2}^2$ if and only if X is P_{4k+2}^2 , nowhere P_{4k+2}^1 , nowhere $P_{4(k-1)+3}^2;$
 $X \in X_{4k+3}^1$ if and only if X is P_{4k+3}^1 , nowhere P_{4k+2}^1 , nowhere $P_{4k+1}^1;$
 $X \in X_{4k+3}^2$ if and only if X is P_{4k+3}^2 , nowhere P_{4k+2}^1 , nowhere $P_{4k+1}^1;$
 $X \in X_{4k+3}^2$ if and only if X is P_{4k+3}^2 , nowhere P_{4k+2}^1 , nowhere $P_{4k+1}^1;$

Observe that if $X \in X_n^{(i)}$, then X is nowhere compact, so X can be densely embedded in C, and any such embedding is also co-dense (corollary 2.1.4).

3.4.7 LEMMA: Let $n < \omega$, $i \in \{1,2\}$, and $X \in X_n^{(i)}$. (a) If n is odd, then X is first category. (b) If n is even, then X contains a dense complete subset, and hence X is Baire.

Proof: If n is odd, then X is strongly $\sigma - P_{n-1}^{(i)}$, and nowhere $P_{n-1}^{(i)}$, so X is first category. If n = 0, then X itself is complete. If n = 4k > 0,

then X is P_{4k} , hence $X = A \cup B$, where A is S_k and B is complete; since X is nowhere $P_{4(k-1)+1}$, X is nowhere S_k , hence B is dense in X. The other cases are proved similarly.

The following lemma gives alternative descriptions of the classes $X_n^{(i)}$ for even n, and of X_{4k+3}^1 ; in some cases, these are easier to use than those of definition 3.4.6.

3.4.8 LEMMA: Let X be zero-dimensional. Then for each $k < \omega$, (a) $X \in X_{4k}$ if and only if X is P_{4k} and nowhere $P_{4(k-1)+3}^2$. (b) $X \in X_{4k+2}^1$ if and only if X is P_{4k+2}^1 and nowhere P_{4k+1}^1 . (c) $X \in X_{4k+2}^2$ if and only if X is P_{4k+2}^2 and nowhere P_{4k+3}^1 . (d) $X \in X_{4k+3}^1$ if and only if X is P_{4k+3}^1 , nowhere P_{4k+2}^2 , nowhere P_{4k+1}^2 . Proof: We need only consider the "nowhere" properties. The "if" part follows immediately from lemma 3.4.5 ((a) from theorem 2.3.1 if k = 0). The "only if" of (a) is trivial if k = 0, so suppose that k > 0 and $X \in X_{4k}^2$. Let U be a non-empty open subset of X, and suppose that U is $P_{4(k-1)+3}^2$; then by lemma 3.4.2, U is strongly $\sigma -P_{4(k-1)+2}^2$. But X is nowhere $P_{4(k-1)+2}^2$, hence U is not Baire, contradicting lemma 3.4.7. For part (b), suppose that $X \in X_{4k+2}^1$, U is non-empty and open in X and P_{4k+1}^2 . Again, since X is nowhere P_{4k}^2 , U is not Baire, a contradiction. Part (c) is similar. For (d), note that if U is a non-empty open subset of X which is P_{4k+2}^2 , then $U \in X_{4k+2}^2$, so U is Baire, contradicting lemma 3.4.7.

We now prove that each of the classes defined above contains at least one element.

3.4.9 LEMMA: Let $n < \omega$ be even, $i \in \{1,2\}$, and $X \in X_n^{(i)}$. Then $Q \times X \in X_{n+1}^{(i)}$.

Proof: Since $Q \times X = \bigcup_{q \in Q} (\{q\} \times X)$, in all cases $Q \times X$ is strongly $\sigma - P_n^{(i)}$, hence $P_{n+1}^{(i)}$ by lemma 3.4.2. Let $U \times V$ be a non-empty basic clopen subset of $Q \times X$.

<u>Case 1</u>: n = 4k. Assume that $U \times V$ is $P_{4(k-1)+3}^2$, and non-empty. Then by lemma 3.4.1, V is $P_{4(k-1)+3}^2$. But $X \in X_{4k}$, and by lemma 3.4.8, X is nowhere $P_{4(k-1)+3}^2$, a contradiction. So $Q \times X$ is nowhere $P_{4(k-1)+3}^2$, hence so is $U \times V$. Now suppose that $U \times V$ is P_{4k}^2 ; then by lemma 3.4.8, $U \times V \in X_{4k}$,

so $U \times V$ is Baire by lemma 3.4.7. But $U \times V = \bigcup_{q \in U} (\{q\} \times V)$ is first category, contradiction. So $Q \times X$ is nowhere P_{4k} . <u>Case 2</u>: n = 4k+2, i=1. If $U \times V$ is P_{4k+1} , and non-empty, then so is V; but $V \in X_{4k+2}^1$, a contradiction. If $U \times V$ is P_{4k+2}^1 , then $U \times V$ is P_{4k+2}^1 and nowhere P_{4k+1} , so $U \times V \in X_{4k+2}^1$ by lemma 3.4.8, so $U \times V$ is Baire by lemma 3.4.7, and as in case 1, we have a contradiction. Case 3: n = 4k+2, i=2. As above.

As was noted in section 3, to obtain elements of $X_n^{(i)}$ for even n, we have to consider complements in the Cantor set of spaces belonging to $X_m^{(j)}$ for some odd m.

3.4.10 LEMMA: Let X be dense and co-dense in the Cantor set, and let $k < \omega.$ Then

(a) $X \in X_{4(k+1)}$ if and only if $C \setminus X$ is P_{4k+3}^2 , nowhere P_{4k+2}^2 , nowhere P_{4k+1} . (b) $X \in X_{4k+1}$ if and only if $C \setminus X$ is P_{4k+2}^2 , nowhere P_{4k}^2 , nowhere $P_{4(k-1)+3}^2$. Proof: (a) If $X \in X_{4(k+1)}$, then X is $P_{4(k+1)}$, so by lemma 3.4.4, $C \setminus X$ is P_{4k+3}^2 . If V is a non-empty clopen subset of C, and $V \setminus X$ is P_{4k+2}^2 , then $\emptyset \neq V \cap X$ is P_{4k+1} by lemma 3.4.4, contradicting the fact that X is nowhere P_{4k+1} . Similarly, $V \setminus X$ being P_{4k+1} would contradict X being nowhere P_{4k+2}^2 . The converse of (a), and (b), are proved in the same way. \Box

3.4.11 COROLLARY: Let X be a dense subset of the Cantor set. Then for each $k < \omega$, and $i \in \{1,2\}$, (a) if $X \in X_{4k+2}^{i}$ then $C \setminus X \in X_{4k+1}$; (b) if $X \in X_{4(k-1)+3}^{i}$ then $C \setminus X \in X_{4k}$.

Proof: As noted before, X is co-dense in C. If $X \in X_{4k+2}^{i}$, then X is P_{4k+2}^{2} , nowhere P_{4k}^{2} , and nowhere $P_{4(k-1)+3}^{2}$, using lemma 3.4.5. So by lemma 3.4.10, $C \setminus X \in X_{4k+1}^{2}$. Part (b) is proved similarly.

3.4.12 LEMMA: For each $n < \omega$ and $i \in \{1,2\}, X_n^{(i)} \neq \emptyset$. Proof: This is clear for n = 0, so assume that $X_m^{(j)} \neq \emptyset$ for $m < n (\geq 1)$, $j \in \{1,2\}$. If n is odd, then $X_n^{(i)} \neq \emptyset$ by lemma 3.4.9. So suppose that n is even.

<u>Case 1</u>: n = 4k. By the inductive hypothesis, $X_{4(k-1)+3}^1 \neq \emptyset$, say $X \in X_{4(k-1)+3}^1$. Embed X as a dense subset of C; then by corollary 3.4.11, $C \setminus X \in X_{4k}$. <u>Case 2</u>: n = 4k+2, $i \in \{1,2\}$. Let $X_i \in X_{4(k-1)+3}^i$, and embed X_i as a dense subset of the Cantor set; also embed Q densely in C, and let

$$Y_{i} = (C \times C) \setminus (Q \times (C \setminus X_{i})).$$

By corollary 3.4.11, $C \setminus X_i \in X_{4k}$; so by lemma 3.4.9, $Q \times (C \setminus X_i) \in X_{4k+1}$. Hence by lemma 3.4.10, Y_i is P_{4k+2}^2 , nowhere P_{4k} , and nowhere $P_{4(k-1)+3}^2$. Case 2a: i = 1. To show that $Y_1 \in X_{4k+2}^1$, it suffices to show that Y_1 is P_{4k+2}^1 . But

$$\mathbb{Y}_1 = (Q \times \mathbb{X}_1) \cup ((C \setminus Q) \times C);$$

and since X_1 is $P_4^1(k-1)+3$, so is $Q \times X_1$ by corollary 3.4.3. So Y_1 is $P_4^1(k-1)+3 \cup$ complete is P_{4k+2}^1 .

Case 2b: i = 2. To show that $Y_2 \in X_{4k+2}^2$, it suffices to show that Y_2 is nowhere P_{4k+2}^1 . So suppose that $U \times V$ is a non-empty basic clopen subset of $C \times C$, and suppose that $(U \times V) \cap Y_2$ is P_{4k+2}^1 . Since Q is dense in C, $U \cap Q \neq \emptyset$, say $x \in U \cap Q$; then $(U \times V) \cap Y_2 \cap (\{x\} \times C) = \{x\} \times (V \cap X_2)$, so $V \cap X_2$ is also P_{4k+2}^1 , i.e. $V \cap X_2 = A \cup B$, where A is $P_{4(k-1)+3}^1$, and Bis complete. Since X_2 is nowhere $P_{4(k-1)+3}^1$, B is dense in $V \cap X_2$, so $V \cap X_2$ is Baire. This contradicts lemma 3.4.7.

We now state and prove the main theorems of this section.

3.4.13 THEOREM: Up to homeomorphism, each $X_n^{(i)}$ contains exactly one element, which is strongly homogeneous, whence homogeneous.

3.4.14 THEOREM: If $X \in X_{4k+2}^{i} \cup X_{4k+3}^{i}$ for some $k < \omega$ and $i \in \{1,2\}$, then C is homogeneous with respect to dense copies of X.

3.4.15 THEOREM: Let A be a Borel subset of a compact space X. Then for each $k < \omega$, (a) A is not P_{4k} if and only if X contains a Cantor set K such that $K \cap A \in X_{4(k-1)+3}^1$ and $K \setminus A \in X_{4k}$. (b) A is not P_{4k+2}^1 if and only if X contains a Cantor set K such that $K \cap A \in X_{4(k-1)+3}^2$ and $K \setminus A \in X_{4k}$. (c) A is not P_{4k+1} if and only if X contains a Cantor set K such that $K \cap A \in X_{4(k-1)+3}^2$ and $K \setminus A \in X_{4k}$. (d) A is not P_{4k+3}^1 if and only if X contains a Cantor set K such that $K \cap A \in X_{4k+2}^2$ and $K \setminus A \in X_{4k+1}$. Remark: Theorem 3.4.13 for X_1 is the characterization of $Q \times P$ due to van Mill [38]. The unique elements of X_2^1 and X_2^2 are the sets T, resp. S, mentioned in the introduction to this monograph; for these classes, theorem 3.4.13 is due to van Douwen [7] (see also van Engelen and van Mill [14], which includes theorem 3.4.14 for k = 0, i = 1, resp. van Mill [39] (which includes theorem 3.4.14 for k = 0, i = 2). Theorem 3.4.13 for n = 3, i ϵ {1,2} is due to van Engelen and van Mill [14], and so is theorem 3.4.15 for k = 0 (see the introduction to section 3.3, and also theorem 3.3.1).

We will prove the above theorems by induction on n, using the following statements as inductive hypotheses:

- (1) Up to homeomorphism, $X_n^{(i)}$ contains at most one element. (2) If $n \in \{4k+2, 4k+3\}$ for some $k < \omega$, $i \in \{1, 2\}$, and $X \in \{i\}$ $X_n^{(i)}$, then C is homogeneous with respect to dense copies of X.
- (3) If n = 4k, then theorem 3.4.15(a) and (b) hold for this k; If n = 4k+2, then theorem 3.4.15(c) and (d) hold for this k.

From (1), it follows that if $X \in \chi^{(i)}$, and V is a non-empty clopen subset of X, then $V \approx X$ since $V \in \chi_n^{(i)}$; thus X is strongly homogeneous. So if (1), (2), and (3) have been proved for all $n < \omega$, then theorem 3.4.13 follows from (1) and lemma 3.4.12, and theorems 3.4.14 and 3.4.15 are clear from (2) and (3).

Since (1) is clear for n = 0, and since (3) has been proved in section 3.3 for n = 0, we may assume that (1), (2), and (3) have been proved for m < m < 1 $n (\geq 1)$. The proof of (1), (2), and (3) for m = n will consist of a series of lemmas.

3.4.16 LEMMA: If n is odd, then (1) holds.

Proof: Let $X \in X_{n}^{(i)}$; we will show that $X \approx Q \times Y$, where Y is the unique element of $\chi_{n-1}^{(i)}$. Since X is $P_{n}^{(i)}$, we can write $X = \bigcup_{j=0}^{\infty} X_{j}$, where X_{j} is closed in X and $P_{n-1}^{(i)}$ (lemma 3.4.2). Fix $j < \omega$, and let \mathcal{D} be a disjoint covering of Y). X\X; by clopen subsets of X such that for each $D \in \mathcal{D}$, diam(D) < d(D,X) (see the proof of lemma 3.2.2).

CLAIM: If U is open in X, and $U \cap X_j \neq \emptyset$, then $U \supset D$ for some $D \in \mathcal{D}$. Indeed, let $x \in U \cap X_j$, and $B(x,\varepsilon) \subset U$. Since X is nowhere $P_{n-1}^{(i)}$,

X_j is nowhere dense in X, so $B(x, \frac{1}{2}\epsilon) \cap D \neq \emptyset$ for some $D \in \mathcal{D}$. Then diam(D) < d(D,X_j) < $\frac{1}{2}\epsilon$ for this $D \in \mathcal{D}$, so $D \subset B(x,\epsilon) \subset U$. <u>Case 1</u>: n = 4k+1. Let $D \in \mathcal{D}$. Then D is not $P_{4(k-1)+3}^2$ since X is nowhere $P_{4(k-1)+3}^2$, so if we embed D in C, then $C \setminus D$ is not P_{4k} by lemma 3.4.4; since (3) holds for m = 4k, we can apply theorem 3.4.15(a) to obtain a closed subset E(D) of D such that $E(D) \approx Y \in X_{4k}$. Then

 $A_i = X_i \cup \bigcup_{D \in \mathcal{D}} E(D)$

is closed in X, and it is easily seen that A_j is P_{4k} , and hence nowhere dense in X. If U is a non-empty open subset of A_j , say $U = U' \cap A_j$, with U' open in X, then either $U' \cap X_j = \emptyset$, hence $U \cap E(D) \neq \emptyset$ for some $D \in \mathcal{D}$, or $U' \cap X_j \neq \emptyset$, hence $U \supset E(D)$ for some $D \in \mathcal{D}$, by the claim. In both cases it follows that U is not $P_{4(k-1)+3}^2$ since E(D) is nowhere $P_{4(k-1)+3}^2$, and hence $A_j \approx Y$ (lemma 3.4.8). Thus $X = \bigcup_{j=0}^{\infty} A_j$ satisfies the hypotheses of theorem 3.2.4, and hence $X \approx Q \times Y$. Case 2: n = 4k+3. This is similar to case 1: since (3) holds for m = 4k+2, we here apply theorem 3.4.15(c) (if i = 1) or (d) (if i = 2) to obtain $E(D) \in X_{4k+2}^i$. Then each $A_j \approx Y \in X_{4k+2}^i$, and again $X \approx Q \times Y$ by theorem 3.2.4.

3.4.17 LEMMA: If
$$n = 4k+3$$
 for some $k < \omega$, then (2) holds.
Proof: Let $X \in X_{4k+2}^{i}$ ($i \in \{1,2\}$). By lemmas 3.4.9 and 3.4.16, $Q \times X$ is the
unique element of X_{4k+3}^{i} . Since (2) holds for $m = 4k+2$, the Cantor set is
homogeneous with respect to dense copies of X. Now apply theorem 3.2.8. \Box

Lemmas 3.4.16 and 3.4.17 show that the inductive hypotheses are satisfied for m = n if n is odd. The remaining lemmas of this section deal with the case where n is even.

3.4.18 LEMMA: Let $k \in N$, and let $X \in X_{4k}$. Then X can be embedded in the Cantor set such that $C \setminus X \in X_{4(k-1)+3}^1$.

Proof: Embed X in C as a dense subset. By lemma 3.4.10, $C \setminus X$ is $P_{4(k-1)+3}^2$, nowhere $P_{4(k-1)+2}^2$, and nowhere $P_{4(k-1)+1}$. Thus we can write $C \setminus X = A \cup B$, where A is $P_{4(k-1)+1}$, and B is σ -compact; since A\B is a G_{δ} -subset of A, it is $P_{4(k-1)+1}$ by lemma 3.4.1, so we may assume that $A \cap B = \emptyset$. Note that $X \subset C \setminus B$, and that $C \setminus B$ is complete. Also, since $C \setminus X$ is nowhere $P_{4(k-1)+1}$, B is dense in $C \setminus X$, and hence in C, so $C \setminus B$ is nowhere compact.

52

Hence by theorem 2.3.1, $C \setminus B \approx P$. Put $Y = C \setminus B$, and embed Y in $K \approx C$ such that $K \setminus Y \approx Q$. We claim that $K \setminus X \in X_{4(k-1)+3}^1$. Indeed, by lemma 3.4.10, $K \setminus X$ is nowhere $P_{4(k-1)+2}^2$, and nowhere $P_{4(k-1)+1}$; and $K \setminus X = (Y \setminus X) \cup Q = A \cup Q$ is $P_{4(k-1)+1} \cup$ countable is $P_{4(k-1)+3}^1$.

3.4.19 LEMMA: If n = 4k for some $k \in N$, then (1) holds.

Proof: Let $X_0, X_1 \in X_{4k}$. By lemma 3.4.18, X_0 and X_1 can be embedded in C such that $C \setminus X_0, C \setminus X_1 \in X_{4(k-1)+3}^1$. Since (1) holds for $m = 4(k-1)+3, C \setminus X_0 \approx C \setminus X_1$, and hence by (2), there exists a homeomorphism h: $C \to C$ such that $h[C \setminus X_0] = C \setminus X_1$, and thus $h[X_0] = X_1$.

3.4.20 LEMMA: If n = 4k for some $k \in N$, then (3) holds.

Proof: We have to prove theorem 3.4.15(a) and (b) for this k. For the "if" part of (a), observe that if A is P_{4k} , then K\A is $P_{4(k-1)+3}^2$ by lemmas 3.4.4 and 3.4.1, contradicting lemma 3.4.8. For "if" of (b), note that if A is P_{4k+2}^1 , then K \cap A is P_{4k+2}^1 is $P_{4(k-1)+3}^1 \cup$ complete, and since K \cap A $\in X_{4(k-1)+3}^2$, it is nowhere $P_{4(k-1)+3}^1$, hence Baire, contradicting lemma 3.4.7. We will now prove the "only if " parts.

(a) Let $P = P_{4(k-1)+1}$, and let Z be the unique element of $X_{4(k-1)+2}^1$; we will show that the hypotheses of lemma 3.3.4 are satisfied. By lemma 3.4.1, P is closed-hereditary, and P is strongly σ -additive by corollary 3.4.3. Let B be a Borel subset of a compact space Y which is not P; since (3) holds for m = 4(k-1)+2, Y contains a Cantor set K such that $K \cap B \in X_{4(k-1)+2}^1$ and $K \setminus B \in X_{4(k-1)+1}$. Then $K \cap B \approx Z$, and by lemma 3.4.2 we can write $K \setminus B = \bigcup_{i=0}^{\infty} X_i$, where X_i is closed in $K \setminus B$ and $P_{4(k-1)}$. Put $G = \bigcup_{i=0}^{\infty} \overline{X}_i$; then G is σ -compact, and since $\overline{X}_i \cap B = \overline{X}_i \setminus X_i$ is $P_{4(k-2)+3}^2$ by lemma 3.4.4, $G \cap B$ is strongly $\sigma - P_{4(k-2)+3}^2$, hence $P_{4(k-2)+3}^2$ by corollary 3.4.3, and thus $G \cap B$ is P by lemma 3.4.5. So by lemma 3.3.4, if A is a Borel subset of a compact space X which is not P_{4k} , or equivalently, not $P \cup$ complete, then X contains a Cantor set K such that $K = \overline{K} \cap A$ and $K \cap A \approx Q \times Z$; then $K \cap A \in X_{4(k-1)+3}^1$ by lemma 3.4.9, and hence $K \setminus A \in X_{4k}$ by corollary 3.4.11.

(b) Since $P_{4k+2}^1 \equiv P_{4(k-1)+3}^1 \cup \text{complete}$, here we put $P = P_{4(k-1)+3}^1$; then again P is closed-hereditary and strongly σ -additive. Let Z be the unique element of $X_{4(k-1)+2}^2$, and suppose that B is a Borel subset of a compact space Y which is not P; since (3) holds for m = 4(k-1)+2, Y contains a Cantor set K such that $K \cap B \in X_{4(k-1)+2}^2$, and $K \setminus B \in X_{4(k-1)+1}$. As above, we put $K \setminus B = \bigcup_{i=0}^{\infty} X_i$, and $G = \bigcup_{i=0}^{\infty} \overline{X}_i$; then $G \cap B$ is $P_{4(k-2)+3}^2$, and hence $G \cap B$ is P by lemma 3.4.5. So by lemma 3.3.4, if A is a Borel subset of a compact space X which is not P_{4k+2}^1 , then X contains a Cantor set K such that $K = \overline{K \cap A}$ and $K \cap A \approx Q \times Z$; then $K \cap A \in X_{4(k-1)+3}^2$ by lemma 3.4.9, and hence $K \setminus A \in X_{4k}$ by corollary 3.4.11.

Lemmas 3.4.18, 3.4.19, and 3.4.20 show that the inductive hypotheses are satisfied if n = 4k for some $k \in N$.

3.4.21 LEMMA: Let $k < \omega$, $i \in \{1,2\}$, and let $X \in X_{4k+2}^i$ be a dense subset of the Cantor set. Then there exist closed nowhere dense subspaces A. of $C \setminus X$ such that $C \setminus X = \bigcup_{j=0}^{\infty} A_j$ and $\overline{A}_j \setminus A_j \in X_4^i(k-1)+3$ is dense in \overline{A}_j . Proof: By corollary 3.4.11, $C \setminus X \in X_{4k+1}$, so by lemma 3.4.2 we can write $C \setminus X = \bigcup_{\ell=0}^{\infty} X_{\ell}$, with X_{ℓ} closed in $C \setminus X$ and P_{4k} . Since $\overline{X}_{\ell} \setminus X_{\ell} = \overline{X}_{\ell} \cap X$ is closed in X, it is P_{4k+2}^{i} by lemma 3.4.1; hence $\overline{X}_{\ell} \setminus X_{\ell} = G_{\ell} \cup H_{\ell}$, where G_{ℓ} is complete, and H_{ℓ} is $P_{4(k-1)+3}^{i}$. Then $\overline{X}_{\ell} \setminus G_{\ell}$ is σ -compact, say $\overline{X}_{\ell} \setminus G_{\ell} = \bigcup_{m=0}^{\infty} K(\ell, m)$, with $K(\ell, m)$ compact; note that $X_{\ell} \subset \overline{X}_{\ell} \setminus G_{\ell}$. Since $\{K(\ell,m) \setminus X: \ell, m < \omega\}$ is countable, we can enumerate the non-empty elements of this set as $\{ \underbrace{\mathtt{M}}_{i} : j \in I \}$, where I is countable. Fix $j \in I$. Since M_j is a closed subset of some X_{ℓ} , M_j is P_{4k} ; and \overline{M}_j $\cap X = \overline{M}_{j} \cap H_{\ell}$ for this same ℓ , hence $\overline{M}_{j} \cap X$ is $\widetilde{P}_{4(k-1)+3}^{i}$. Observe that since $\overline{C} \setminus X$ is nowhere P_{4k} , each M_j is nowhere dense in $C \setminus X$, so we can take I = ω . Now let \mathcal{D} be a disjoint covering of $C \setminus \overline{M}_{j}$ by clopen subsets of C such that for each $D \in \mathcal{D}$, diam(D) < d(D, \overline{M}_j). Since X is nowhere P_{4k} (if i = 1), resp. nowhere P_{4k+2}^1 (if i = 2), and since (3) holds for m = 4k, each D ϵ D contains a Cantor set K(D) such that K(D) \cap X ϵ $X_{4(k-1)+3}^{i}$ and $K(D) \setminus X \in X_{4k}$. Put

 $K_{1} = \overline{M}_{1} \cup U_{D \in \mathcal{D}} K(D),$

and

$$A_i = K_i \setminus X_i$$

we claim that the sets A_i constructed in this way are as required. First note that $C \setminus X = \bigcup_{j=0}^{\infty} A_j$ since $K_j \setminus X \supset M_j$. Furthermore, since M_j is nowhere dense in $C \setminus X$, also M_j is nowhere dense in C; it then follows from the fact that diam(D) < $d(D, \overline{M}_j)$ for each $D \in \mathcal{D}$ that $(\bigcup_{D \in \mathcal{D}} K(D))^- = K_i$, and thus

$$K_{j} \supset \overline{A}_{j} \supset (\overline{A}_{j} \setminus A_{j})^{-} \supset (U_{D \in \mathcal{D}}(K(D) \cap X))^{-} = (U_{D \in \mathcal{D}}(K(D))^{-} = K_{j},$$

so $\overline{A}_{j} \setminus A_{j}$ is dense in \overline{A}_{j} . It remains to show that $\overline{A}_{j} \setminus A_{j} \in X_{4}^{i}(k-1)+3$. Clearly, $\overline{A}_{j} \setminus A_{j} = (\overline{M}_{j} \cap X) \cup \bigcup_{D \in \mathcal{D}} (K(D) \cap X)$ is strongly $\sigma - P_{4}^{i}(k-1)+3$ is $P_{4}^{i}(k-1)+3$ by corollary 3.4.3. And as in the proof of lemma 3.4.16, it follows that if U is a non-empty open subset of $\overline{A}_{j} \setminus A_{j}$, then U $\cap K(D) \cap X \neq \emptyset$ for some $D \in \mathcal{D}$; hence, since $K(D) \cap X \in X_{4}^{i}(k-1)+3$, U is not $P_{4}^{2}(k-1)+2$; and not $P_{4}(k-1)+1$ (if i = 1), resp. not $P_{4}^{1}(k-1)+3$ (if i = 2). Thus, $\overline{A}_{j} \setminus A_{j}$ $\in X_{4}^{i}(k-1)+3$; by corollary 3.4.11, $A_{j} \in X_{4k}$, and thus A_{j} is nowhere dense in $C \setminus X$ since $C \setminus X$ is nowhere P_{4k} .

3.4.22 LEMMA: If n = 4k+2 for some $k < \omega$, $i \in \{1,2\}$, and $X_0, X_1 \in X_n^1$ are dense subsets of C, then there exists a homeomorphism $h: C \rightarrow C$ such that $h[X_0] = X_1$. Thus, (1) and (2) are satisfied if n = 4k+2.

Proof: Let A be the unique element of $\chi_{4(k-1)+3}^{i}$. By lemma 3.4.21, we can write $C \setminus X_0 = \bigcup_{j=0}^{\infty} A_j$, $C \setminus X_1 = \bigcup_{j=0}^{\infty} B_j$, such that $\overline{A}_j \setminus A_j \approx \overline{B}_j \setminus B_j \approx A$, $\overline{A}_j \setminus A_j$ is dense in \overline{A}_j , $\overline{B}_j \setminus B_j$ is dense in \overline{B}_j , A_j is closed and nowhere dense in $C \setminus X_0$, and B_j is closed and nowhere dense in $C \setminus X_1$. Hence the required homeomorphism can be obtained from lemma 3.2.6.

3.4.23 LEMMA: If n = 4k+2 for some $k < \omega$, then (3) holds.

Proof: We have to prove theorem 3.4.15(c) and (d) for this k. The "if" part is easy: if A is P_{4k+1} (resp. P_{4k+3}^1), then so is $K \cap A$, so $K \cap A \notin X_{4k+2}^1$ (resp. X_{4k+2}^2) by lemma 3.4.8. For the "only if" parts, we will apply lemma 3.3.5.

(c) Let $P = P_{4k+1}$, let Z' be the unique element of X_{4k} , and let Y' be the unique element of $X_{4(k-1)+3}^1$. By lemma 3.4.1, P is closed-hereditary, and P is strongly σ -additive by corollary 3.4.3. Let B be a Borel subset of a compact space Y which is not P. Then B is not P_{4k} by lemma 3.4.5, hence since (3) holds for m = 4k, Y contains a Cantor set K such that $K \cap B \in X_{4(k-1)+3}^1$ and $K \setminus B \in X_{4k}$; thus, $K \cap B \approx Y'$, $K \setminus B \approx Z'$, and $K \cap B$ is P by lemma 3.4.5. Hence by lemma 3.3.5, if A is a Borel subset of a compact space X which is not P, then X contains Cantor sets B_i , for i < ω , such that $B_i = \overline{B_i \setminus A}$, $K = (\bigcup_{i=0}^{\infty} B_i)^{-i}$, and for each $i < \omega$:

- (i) B_{i} is nowhere dense in K;
- (ii) $K \setminus A = \bigcup_{i=0}^{\infty} (B_i \setminus A);$
- (iii) $B_0 \setminus A \approx Z' \approx (B_{i+1} \setminus B_i) \setminus A;$
- (iv) $B_0 \cap A \approx Y' \approx (B_{i+1} \setminus B_i) \cap A$.

Since $B_{i+1} \setminus B_i$ is open in B_{i+1} , we can write $B_{i+1} \setminus B_i = \bigcup_{j=0}^{\infty} K_j^i$, where K_j^i is clopen in B_{i+1} . Since both Y' and Z' are strongly homogeneous, either $K_j^i \cap A \approx Y'$ or $K_j^i \cap A = \emptyset$, and $K_j^i \setminus A \approx Z'$ or $K_j^i \setminus A = \emptyset$. Hence, $K \setminus A = (B_0 \setminus A) \cup \bigcup_{i=0}^{\infty} \bigcup_{j=0}^{\infty} (K_j^i \setminus A) \approx Q \times Z'$

by (i) and theorem 3.2.4, so $K \setminus A \in X_{4k+1}$ by lemma 3.4.9. Then $K \cap A$ is nowhere P_{4k}^{2} , and nowhere $P_{4(k-1)+3}^{2}$ by lemma 3.4.10. So we need only show that $K \cap A$ is P_{4k+2}^{1} . However, by (ii),

$$K \cap A = (B_0 \cap A) \cup \bigcup_{i=0}^{\infty} \bigcup_{j=0}^{\infty} (K_j^i \cap A) \cup K \setminus (\bigcup_{i=0}^{\infty} \overline{B}_i)$$

is strongly $\sigma - P_{4(k-1)+3}^{1} \cup \text{complete is } P_{4(k-1)+3}^{1} \cup \text{complete is } P_{4k+2}^{1}$. (d) Let $P = P_{4k+3}^{1}$, let Z' be the unique element of X_{4k} , and let Y' be the unique element of $X_{4(k-1)+3}^{2}$. Then P is closed-hereditary and strongly σ -additive. Let B be a Borel subset of a compact space Y which is not P. Then B is not P_{4k+2}^{1} , so since (3) holds for m = 4k, Y contains a Cantor set K such that $K \cap B \approx Y'$, $K \setminus B \approx Z'$, whence $K \cap B$ is P by lemma 3.4.5. So again by lemma 3.3.5, we find Cantor sets B₁ as under (a); then $K \setminus A \in X_{4k+1}$, hence by lemma 3.4.10, $K \cap A$ is P_{4k+2}^{2} , and nowhere $P_{4(k-1)+3}^{2}$, so it suffices to show that $K \cap A$ is nowhere P_{4k+2}^{1} . Put $B_{1+1} \setminus B_{1}$ $= \bigcup_{j=0}^{\infty} K_{j}^{1}$ as in (a), and note that if U is a non-empty open subset of K $\cap A$, then $U \cap K_{j}^{1} \neq \emptyset$ for some i, $j < \omega$; thus, U contains a closed copy of Y'. Now suppose that U is P_{4k+2}^{1} ; then so is Y' by the preceding remark, i.e. Y' = A \cup B where A is $P_{4(k-1)+3}^{1}$, and B is complete. Since Y' ϵ $X_{4(k-1)+3}^{2}$, Y' is nowhere $P_{4(k-1)+3}^{1}$, hence B is dense in Y', so Y' is Baire, contradicting lemma 3.4.7.

This completes the proof of theorems 3.4.13, 3.4.14, and 3.4.15. In section 6 of this chapter we will show that there are no other homogeneous non- σ -compact S_n -subsets of the Cantor set than those defined and characterized above.

We conclude this section with the following theorem, announced in section 3.1.

3.4.24 THEOREM: If we order the properties $P_{n}^{(i)}$ by $P_{-1}^{1} < P_{-1}^{2} < P_{0}$, and $P_{4k} < P_{4k+1} < P_{4k+2}^{1} < P_{4k+3}^{1} < P_{4k+2}^{2} < P_{4k+3}^{2} < P_{4(k+1)}$ if $k < \omega$, then for each $n < \omega$, and each $i \in \{1,2\}$, we have $X \in X_{n}^{(i)}$ if and only if X is $P_{n}^{(i)}$ and nowhere $P_{m}^{(j)}$ for all $P_{m}^{(j)} < P_{n}^{(i)}$, where m = -1 or $m < \omega$, and $j \in \{1,2\}$.

Proof: Apply lemmas 3.4.5 and 3.4.8.

3.4.25 COROLLARY: If $P_m^{(j)} < P_n^{(i)}$, then $X_m^{(j)} \cap X_n^{(i)} = \emptyset$; thus, all spaces characterized in this section are topologically distinct.

3.5 The small Borel classes $D_{\alpha}(\Sigma_2^0)$ for $\alpha \in [\omega, \omega_1)$

We recall the topological properties P_{β} determined by definition 3.1.8 and theorem 3.1.6; to avoid unnecessary complications, we will just write "Y is P_{γ} for some $\gamma < \omega$ " for "Y is $P_{m}^{(j)}$ for some $m < \omega$, $j \in \{1,2\}$ ". Let $\alpha < \omega_1$ be a limit ordinal, and let $n < \omega$. Then

X is
$$P_{\alpha+2n}$$
 if and only if $X \in D_{\alpha+2n}(\Sigma_2^0)$ if and only if $X = \bigcup_{i=0}^{\infty} A_i \cup B$, where B is S_n , A_i is closed in X, and A_i is P_{α_i} for some $\alpha_i < \alpha$;
X is $P_{\alpha+2n+1}$ if and only if X is $P_{\alpha+2n} \cup$ complete.

3.5.1 LEMMA: For each $\beta \in [\omega, \omega_1), \ P_\beta$ is hereditary with respect to $\ G_\delta^-$ subsets.

Proof: Embed X in a compact space Y. If β is even, then there are certain σ -compact subsets A_{ζ} of Y such that $X = D_{\beta}(\langle A_{\zeta}: \zeta < \beta \rangle)$; if G is a G_{δ} -subset of Y, then $X \cap G = D_{\beta}(\langle A_{\zeta} \cup Y \setminus G: \zeta < \beta \rangle) \in D_{\beta}(\Sigma_{2}^{0})$. The case β is odd follows trivially.

3.5.2 LEMMA: Let X be a topological space, let $\alpha < \omega_1$ be a limit ordinal, and let $n < \omega$. Then (a) X is $P_{\alpha+2n}$ if and only if X is $P_{\alpha} \cup S_n$; (b) X is $P_{\alpha+2n+1}$ if and only if $X = \bigcup_{i=0}^{\infty} A_i \cup B \cup G$, where A_i is P_{α_i} for some $\alpha_i < \alpha$, B is S_n , G is complete, and A_i is closed in X. Proof:(a) If X is $P_{\alpha+2n}$, then clearly X is $P_{\alpha} \cup S_n$. We will prove the converse by induction. Suppose that $X = \bigcup_{i=0}^{\infty} A_i \cup B$, where A_i is closed in $\bigcup_{i=0}^{\infty} A_i$, A_i is P_{α_i} for some $\alpha_i < \alpha$, and B is S_n , and first consider the case $\alpha = \omega$. Then $\overline{A}_i = A_i \cup (\overline{A}_i \cap B)$ is $P_{\alpha_i} \cup S_n$, hence \overline{A}_i is P_{β_i} for some $\beta_i < \omega$, so $X = \bigcup_{i=0}^{\infty} \overline{A}_i \cup B$ is $P_{\omega+2n}$. Now assume that we are done for limit ordinals $\delta < \alpha (> \omega)$; then we may assume that for each i, $\alpha_i \ge \omega$, and hence $\alpha_i = \gamma_i + n_i$ for some limit ordinal $\gamma_i < \alpha$, and some $n_i < \omega$. If n_i is even, then \overline{A}_i is $P_{\gamma_i + n_i} \cup S_n$ is $P_{\gamma_i} \cup S_{\frac{1}{2}n_i} \cup S_n$, hence $P_{\gamma_i + n_i-1+2n+2}$, again by the inductive hypothesis. So each \overline{A}_i is P_{β_i} is P_{β_i} is P_{β_i} is $P_{\gamma_i + n_i-1+2n+2}$, again by the inductive hypothesis. So each \overline{A}_i is P_{β_i} . for some $\beta_i < \alpha$, and thus $X = \bigcup_{i=0}^{\infty} \overline{A}_i \cup B$ is $P_{\alpha+2n}$. (b) The "if" part is clear, so suppose that $X = A \cup G$, where A is $P_{\alpha+2n}$, and G is complete. Put $A = \bigcup_{i=0}^{\infty} B_i \cup B$, where B_i is closed in A and P_{α_i} for some $\alpha_i < \alpha$, and B is S_n , and let $A_i = \overline{B}_i = B_i \cup (\overline{B}_i \cap G)$. Then A_i is $P_{\alpha_i} \cup$ complete, hence certainly A_i is $P_{\alpha_i} \cup S_1$, and thus, using (a), A_i is P_{β_i} for some $\beta_i < \alpha$. Since A_i is closed in X, $X = \bigcup_{i=0}^{\infty} A_i \cup B \cup G$ is as required.

3.5.3 COROLLARY: If
$$n < \omega$$
, $i \in \{1,2\}$, $\beta, \gamma \in [\omega, \omega_1)$, and $\beta \leq \gamma$, then $p_n^{(i)} \rightarrow P_\beta \rightarrow P_\gamma$.

3.5.4 LEMMA: Let X be a topological space, let $\alpha < \omega_1$ be a limit ordinal, and let $n \in N$. Then X is $P_{\alpha+2n}$ if and only if X is strongly σ - $P_{\alpha+2n-1}$.

Proof: The proof is exactly the same as that of lemma 3.4.2. We find that X is $P_{\alpha+2n}$ if and only if X is strongly $\sigma - (P_{\alpha+2(n-1)} \cup \text{complete})$, i.e. if and only if X is strongly $\sigma - P_{\alpha+2n-1}$.

3.5.5 COROLLARY: For even $\beta \in [\omega, \omega_1), \ P_\beta$ is a strongly $\sigma\text{-additive property.}$

3.5.6 LEMMA: Let X be compact, and let A be a subset of X. If A is P_{β} for some $\beta \in [\omega, \omega_1)$, then X\A is $P_{\beta+1}$.

Proof: If β is even, then there exist σ -compact subsets A_{ζ} of X such that $A = D_{\beta}(\langle A_{\zeta}: \zeta < \beta \rangle) = U\{A_{\zeta} \setminus (\bigcup_{\delta < \zeta} A_{\delta}): \zeta \text{ odd } < \beta\}$. Then

$$X \setminus A = U\{A_{\zeta} \setminus (U_{\delta < \zeta} A_{\delta}): \zeta \text{ even } < \beta\} \cup X \setminus (U_{\zeta < \beta} A_{\zeta}).$$

So if we put $B_{\zeta} = U_{\delta < \zeta} A_{\delta}$ for $\zeta < \beta$, then

$$X \setminus A = U\{B_{\zeta+1} \setminus B_{\zeta}: \zeta \text{ even } < \beta\} \cup X \setminus (U_{\zeta < \beta} A_{\zeta}) \\ = D_{\beta}(\langle B_{\zeta}: \zeta < \beta\rangle) \cup X \setminus (U_{\zeta < \beta} A_{\zeta})$$

is $P_{\beta} \cup \text{complete is } P_{\beta+1}$.

If β is odd, say $\beta = \alpha + 2n + 1$, with $\lim(\alpha)$, $n < \omega$, then there exist σ -compact A_{ζ} in X such that $A = D_{\alpha+2n}(\langle A_{\zeta}; \zeta < \alpha+2n \rangle) \cup G$, where G is complete. Defining B_{ζ} as above for $\zeta < \alpha+2n = \gamma$, we find that

58

is $P_{\alpha} \cup S_1$. By lemma 3.5.2, X\A is $P_{\alpha} \cup S_n \cup S_1$ is $P_{\alpha+2n+2}$ is $P_{\beta+1}$.

For each pair (β ,i), with $\beta \in [\omega, \omega_1)$, and i = 2 if β is a limit, resp. i ϵ {1,2} if β is a successor, we now inductively define a class of spaces X^{i}_{β} , and show that, up to homeomorphism, each X^{i}_{β} contains exactly one element.

3.5.7 DEFINITION: Let X be zero-dimensional, let $\alpha < \omega_1$ be a limit ordinal, and $n \in N$; then

 $\begin{array}{l} X \in X_{\alpha}^{2} \quad if \ and \ only \ if \ X \ is \ P_{\alpha}, \ nowhere \ P_{\beta} \quad for \ \beta < \alpha; \\ X \in X_{\alpha+n}^{1} \quad if \ and \ only \ if \ X \ is \ P_{\alpha+n}, \ nowhere \ P_{\alpha+n-1}, \ and \ X \ does \ not \ contain \ any \ closed \ subsets \ belonging \ to \ X_{\alpha+n-1}^{2}; \end{array}$

 $X \in X^2_{\alpha+n}$ if and only if X is $P_{\alpha+n}$, nowhere $P_{\alpha+n-1}$, and every non-empty clopen subset of X contains a closed subset belonging to $X^2_{\alpha+n-1}$.

3.5.8 LEMMA: Let $\beta \in [\omega, \omega_1)$, $i \in \{1, 2\}$, and $X \in X_{\beta}^1$. (a) If β is even, then X is first category. (b) If β is odd, then X contains a dense complete subset, hence X is Baire.

Proof:(a) By the definition of P_{β} (if β is a limit), resp. by lemma 3.5.4 (if β is a successor), we can write $X = \bigcup_{i=0}^{\infty} X_i$, where X_i is closed in X, and P_{β_i} for some $\beta_i < \beta$. Since X is nowhere P_{β_i} , each X_i is nowhere dense in X.

(b) Write $X = A \cup B$, where A is P_{B-1} , and B is complete; since X is nowhere $P_{\beta-1}$, B is dense in X. Ο

We now state and prove the main theorems of this section.

3.5.9 THEOREM: If $\beta \in [\omega, \omega_1)$, and i = 2 if β is a limit, $i \in \{1, 2\}$ if β is a successor, then up to homeomorphism, each χ^i_β contains exactly one element, which is strongly homogeneous, whence homogeneous.

3.5.10 THEOREM: If $X \in X^{i}_{\beta}$ for some $\beta \in [\omega, \omega_{1})$ and $i \in \{1, 2\}$, then C is homogeneous with respect to dense copies of X.

3.5.11 THEOREM: Let A be a Borel subset of a compact space X, and let $\beta \in [\omega, \omega_1).$ Then A is not $P_{\beta}^{}$ if and only if X contains a Cantor set K such that $K \cap A \in X^1_{B+1}$ and $K \setminus A \in X^2_B$.

We will prove the above theorems by induction on (ß,i), ordered lexicographically, using the following statements as inductive hypotheses (here, $\alpha < \omega_1$ is a limit ordinal, and $n < \omega$):

- If $\beta = \alpha + 2n$, i = 2, then:
 - (1) $|X_{\alpha+2n}^2| = 1$, and if $X \in X_{\alpha+2n}^2$, then *C* is homogeneous with respect to dense copies of X.
 - (2) If A is a Borel subset of a compact space X which is not $P_{\alpha+2n}$, then X contains a Cantor set K such that $K \setminus A \in X^2_{\alpha+2n}$.
- If $\beta = \alpha + 2n + 1$, i = 1, then:
 - (3) Let X be dense and co-dense in C; then $X \in X_{\alpha+2n+1}^{1}$ if and
 - only if $C \setminus X \in X^2_{\alpha+2n}$. (4) $|X^1_{\alpha+2n+1}| = 1$, and if $X \in X^1_{\alpha+2n+1}$, then C is homogeneous with respect to dense copies of X.
 - (5) If A is a Borel subset of a compact space X which is not $\mathcal{P}_{\alpha+2n}$, then X contains a Cantor set K such that K\A ϵ $X_{\alpha+2n}^2$ and $K \cap A \in X_{\alpha+2n+1}^1$.

If $\beta = \alpha + 2n + 1$, i = 2, then:

- (6) Let X be dense and co-dense in C; then $X \in X^2_{\alpha+2n+1}$ if and only if $C \setminus X \approx Q \times Y$ for some $Y \in X^1_{\alpha+2n+1}$.
- (7) $|X_{\alpha+2n+1}^2| = 1$, and if $X \in X_{\alpha+2n+1}^2$, then C is homogeneous with respect to dense copies of X.
- (8) If A is a Borel subset of a compact space X which is not $\mathcal{P}_{lpha+2\mathbf{n}+1}$, then X contains a Cantor set K such that K\A ϵ $X^2_{lpha+2n+1}$ and $K \cap A pprox Q imes Y$ for some $Y \in X^1_{lpha+2n+1}$.
- If $\beta = \alpha + 2n + 2$, i = 1, then:
 - (9) Let X be dense and co-dense in C; then $X \in X^{1}_{\alpha+2n+2}$ if
 - and only if $C \setminus X \in X^2_{\alpha+2n+1}$. (10) $|X^1_{\alpha+2n+2}| = 1$, and if $X \in X^1_{\alpha+2n+2}$, then C is homogeneous with respect to dense copies of X.
 - (11) If A is a Borel subset of a compact space X which is not $P_{\alpha+2n+1}$, then X contains a Cantor set K such that $K \setminus A \in X_{\alpha+2n+1}^2$ and $K \cap A \in X_{\alpha+2n+2}^1$.

So let $\alpha < \omega_1$ be a limit ordinal, let $m < \omega$ and i $\in \{1,2\}$, and suppose that the inductive hypotheses are satisfied for all $(\beta,j) < (\alpha+m,i);$ note that from the fact that $|X_{\beta}^{j}| = 1$ it follows as in section 3.4 that the

unique element of X_{β}^{j} is strongly homogeneous. We now show that the inductive hypotheses are satisfied for (α +m,i).

We start with the case m = 0 (hence i = 2); of course, we have to use the results of section 3.4 in case $\alpha = \omega$.

3.5.12 LEMMA: $X_{\alpha}^2 \neq \emptyset$.

Proof: Let $\langle \alpha_i : i < \omega \rangle$ be an increasing sequence of ordinals less than α , such that $\sup_{i < \omega} \alpha_i = \alpha$; for convenience of notation, let $\alpha_i \equiv 3 \pmod{4}$ if $\alpha_i < \omega$. Let $Y_i \in X^2_{\alpha_i}$ be densely embedded in *C*. Enumerate *Q* as $\{q_i : i < \omega\}$, and put

$$\mathbf{X} = \bigcup_{i=0}^{\infty} (\{\mathbf{q}_i\} \times \mathbf{Y}_i) \subset \mathbf{Q} \times \mathbf{C}.$$

Then clearly, X is P_{α} . Also, if $W = X \cap (U \times V)$ is a non-empty basic clopen subset of X, then $W = \bigcup_{j \in E} (\{q_j\} \times (V \cap Y_j))$ for some infinite E. Since Y_j is nowhere P_{γ} for each $\gamma < \alpha_j, \{q_j\} \times (V \cap Y_j)$ is not P_{γ} for each $\gamma < \alpha_j$, and hence by lemma 3.4.1 or 3.5.1, W is not P_{γ} for each $\gamma < \alpha_j$ since $\{q_j\} \times (V \cap Y_j)$ is closed in W. Hence X is nowhere P_{γ} for each $\gamma < \alpha$ since $\sup_{j \in E} \alpha_j = \alpha$. So $X \in X^2_{\alpha}$.

3.5.13 LEMMA: If n = 0, then (1) holds.

Proof: By lemma 3.5.12, $|X_{\alpha}^{2}| \ge 1$; so suppose that $X, Y \in X_{\alpha}^{2}$ are dense subsets of the Cantor set. In a way analogous to the proof of theorem 3.2.6, we construct a homeomorphism h: $C \rightarrow C$ such that h[X] = Y. We first consider the case where $\alpha > \omega$. Since X is P_{α} , we can write $X = \bigcup_{i=0}^{\infty} X_{i}$, where X_{i} is closed in X and $P_{\alpha_{i}}$ for some $\alpha_{i} < \alpha$; clearly, we may assume that $\alpha_{i} \ge \omega$. Note that, since X is nowhere $P_{\alpha_{i}}$ for each $i < \omega$, each X_{i} is nowhere dense in X. Similarly, $Y = \bigcup_{i=0}^{\infty} Y_{i}$, with Y_{i} closed and nowhere dense in Y, and Y_{i} is $P_{\beta_{i}}$ for some $\beta_{i} \in [\omega, \alpha)$. As in section 3.2, we construct, for each $s \in M$, collections $U(s) = \{U(s^{\circ}n): n < \omega\}, V(s) = \{V(s^{\circ}n): n < \omega\}$ of clopen subsets of C, closed nowhere dense subsets D(s), E(s) of C, and for each $n \in N$ a homeomorphism $h_{n}: C \rightarrow C$, satisfying the hypotheses of the proof of theorem 3.2.6; then again, $\lim_{n\to\infty} h_{n} = h: C \approx C$, and h[X] = Y.

First, let $\gamma_0 < \alpha$ be an even successor ordinal such that $\alpha_0, \beta_0 \leq \gamma_0$; then both X_0 and Y_0 are P_{γ_0} . Let \mathcal{D} be a disjoint covering of $C \setminus X_0$ by clopen subsets of C such that for each $D \in \mathcal{D}$, diam(D) < d(D, \overline{X}_0). Since X is a Borel subset of C which is nowhere $P_{\gamma_0+1}, C \setminus X$ is nowhere P_{γ_0} by lemma 3.5.6. Hence, applying hypothesis (5) above, each $D \in \mathcal{D}$ contains a Cantor set K(D) such that K(D) $\cap X = E(D) \in X^2_{YO}$. Then

$$A_0 = X_0 \cup U_{D \in \mathcal{D}} E(D)$$

is closed in X, and strongly $\sigma - P_{\gamma_0}$, hence P_{γ_0} by corollary 3.5.5. As in the proof of lemma 3.4.16, it can be shown that, for each non-empty open subset U of A_0 , $U \cap E(D) \neq \emptyset$ for some $D \in \mathcal{D}$, hence A_0 is nowhere P_{γ_0-1} since E(D) is nowhere P_{γ_0-1} . Also, $U \cap E(D)$ contains a subset F which is closed in E(D), hence in A_0 , such that $F \in X^2_{\gamma_0-1}$; thus $A_0 \in X^2_{\gamma_0}$. Similarly, we can find a closed subset B_0 of Y such that $Y_0 \subset B_0$, and $B_0 \in X^2_{\gamma_0}$; note that A_0 is nowhere dense in X, and B_0 is nowhere dense in Y. Now put $D(\emptyset) = \overline{A}_0$, $E(\emptyset) = \overline{B}_0$; by hypothesis (1), applied to γ_0 , there exists a homeomorphism \widetilde{h} : $D(\emptyset) \rightarrow E(\emptyset)$ such that $\widetilde{h}[A_0] = B_0$. We construct $U(\emptyset)$, $V(\emptyset)$, and h_1 as in the proof of theorem 3.2.6. If D(s), E(s), U(s), V(s), and h_m have been constructed for $|s| < n (\ge 1)$ and $m \le n$, then fix $s \in M$ with |s| = n-1, and $i < \omega$. Let

$$k_{i} = \min\{j: U(s^{i}) \cap X_{j} \neq \emptyset\}, \ \ell_{i} = \min\{j: V(s^{i}) \cap Y_{j} \neq \emptyset\},$$

and find an even successor $\gamma(s^i) < \alpha$ such that $\alpha_{k_i}, \beta_{\ell_i} \leq \gamma(s^i)$; then both $U(s^i) \cap X_{k_i}$ and $V(s^i) \cap Y_{\ell_i}$ are $P_{\gamma(s^i)}$. As above, we obtain closed subsets $A(s^i)$ of $U(s^i) \cap X$, $B(s^i)$ of $V(s^i) \cap Y$, such that $A(s^i) \supset U(s^i) \cap X_{k_i}, B(s^i) \supset V(s^i) \cap Y_{\ell_i}$, and both $A(s^i), B(s^i) \in X_{\gamma(s^i)}^2$. Put $D(s^i) = \overline{A(s^i)}, E(s^i) = \overline{B(s^i)}$, and let $\widetilde{h}(s^i): D(s^i) \rightarrow E(s^i)$ be a homeomorphism such that $\widetilde{h}(s^i)[A(s^i)] = B(s^i)$, applying hypothesis (1) to $\gamma(s^i)$; now finish the proof as in section 3.2. In case $\alpha = \omega$, the proof is similar; the sets A_0 , B_0 , $A(s^i)$, and $B(s^i)$ should be chosen in some X_{4k+2}^i or X_{4k+3}^i , because we want to apply theorem 3.4.14.

3.5.14 LEMMA: If n = 0, then (2) holds.

Proof: As in 3.5.13, for sake of simplicity, we will only consider the case $\alpha > \omega$, the case $\alpha = \omega$ being entirely analogous. Let A be a Borel subset of a compact space X which is not P_{α} , let $\langle \alpha_{1}: i < \omega \rangle$ be an increasing sequence of even successor ordinals in $[\omega, \alpha)$ such that $\sup_{i < \omega} \alpha_{i} = \alpha$, let Z_{i} be the unique element of $X_{\alpha_{1}-1}^{2}$, and let Y_{i} be the unique element of $X_{\alpha_{1}}^{1}$. Put $P = P_{\alpha}$; then P is closed-hereditary and strongly σ -additive. Let B be a Borel subset of a compact space Y which is not P. Then for each $i < \omega$, B is not $P_{\alpha_{1}-1}$, hence by hypothesis (11), Y contains a Cantor set

K_i such that K_i\B $\in X^2_{\alpha_i-1}$ and K_i \cap B $\in X^1_{\alpha_i}$. Thus K_i\B \approx Z_i, and K_i \cap B \approx Y_i is P_{α_i} , hence P. So by lemma 3.3.5, X contains Cantor sets B_i, for i < ω , such that B_i = $\overline{B_i \setminus A}$, and K = $(U_{i=0}^{\infty} B_i)^{-} \approx C$, and such that for each i < ω ,

(i)
$$B_i$$
 is nowhere dense in K;
(ii) $K \setminus A = \bigcup_{i=0}^{\infty} (B_i \setminus A);$
(iii) $B_0 \setminus A \approx Z_0, (B_{i+1} \setminus B_i) \setminus A \approx Z_{i+1}$

Since $(B_{i+1} \setminus B_i) \setminus A$ is open in $B_{i+1} \setminus A$, we can write $(B_{i+1} \setminus B_i) \setminus A = \bigcup_{j=0}^{\infty} K_j^i$, where K_j^i is clopen in $B_{i+1} \setminus A$; then $K_j^i = \emptyset$ or $K_j^i \approx Z_{i+1}^i$, so K_j^i is $P_{\alpha_{i+1}-1}$, and hence

$$K \setminus A = (B_0 \setminus A) \cup \bigcup_{i=0}^{\infty} \bigcup_{j=0}^{\infty} K_j^i$$

is P_{α} . By (i), each open subset of K\A intersects some K_{ji}^{1} for infinitely many i; hence since K_{ji}^{1} is nowhere P_{β} for each $\beta < \alpha_{i+1}^{-1}$, K\A is nowhere P_{β} for each $\beta < \alpha$. Thus, K\A $\in X_{\alpha}^{2}$.

Lemmas 3.5.13 and 3.5.14 show that the inductive hypotheses are satisfied for $(\alpha+m,i)$ if m = 0. We now turn to the case where m = 2n+1 for some $n < \omega$, and i = 1.

3.5.15 LEMMA: (3) holds.

Proof: First, let $X \in X_{\alpha+2n+1}^1$; then no closed subset of X belongs to $X_{\alpha+2n}^2$, and hence $C \setminus X$ is $P_{\alpha+2n}$ by hypothesis (2). Let U be a non-empty clopen subset of C.

<u>Case 1</u>: n = 0. Suppose that U\X is P_{β} for some $\beta < \alpha$. By lemmas 3.4.4 and 3.5.6, U \cap X is P_{γ} for some $\gamma < \alpha$, a contradiction, hence $C \setminus X \in X_{\alpha}^2$. <u>Case 2</u>: n > 0. Suppose that U\X is $P_{\alpha+2n-1}$. By lemma 3.5.6, U \cap X is $P_{\alpha+2n}$, a contradiction; hence $C \setminus X$ is nowhere $P_{\alpha+2n-1}$. Also, since U \cap X is not $P_{\alpha+2n-1}$, we obtain from hypothesis (8) a Cantor set K in U such that $K \setminus X \in X_{\alpha+2n-1}^2$; so $C \setminus X \in X_{\alpha+2n}^2$.

Conversely, suppose that $C \setminus X \in X^2_{\alpha+2n}$; then by lemma 3.5.6, X is $P_{\alpha+2n+1}$. Let U be a non-empty clopen subset of C.

<u>Case 1</u>: n = 0. As in case 1 above, X is nowhere P_{β} for each $\beta < \alpha$. Suppose that $U \cap X$ is P_{α} ; then $U \cap X \in X_{\alpha}$, so $U \cap X$ is first category by lemma 3.5.8. Hence $U \setminus X \in X_{\alpha}^2$ is Baire (see 1.12.3), contradicting lemma 3.5.8; so X is nowhere P_{α} . Finally, suppose that X contains a closed subset B belonging to X_{α}^2 . By the same argument as above, $\overline{B} \setminus B$ is nowhere P_{α} ; but $\overline{B} \setminus B$

is a closed subset of $C \setminus X$, so we have a contradiction with lemma 3.5.1. <u>Case 2</u>: n > 0. We must show that U \cap X is not $P_{\alpha+2n}$. Since U\X is nowhere $P_{\alpha+2n-1}$, $U \cap X$ is nowhere $P_{\alpha+2n-2}$ by lemma 3.5.6. Assume that $U \cap X$ is $P_{\alpha+2n-1}$; since U\X is nowhere $P_{\alpha+2n-2}$ (corollary 3.5.3), it follows from hypothesis (5) that every clopen subset of U \cap X contains a closed subset belonging to $X_{\alpha+2n-2}^2$, and hence that $U \cap X \in X_{\alpha+2n-1}^2$. But then by hypothesis (9), $U \setminus X \in X^1_{\alpha+2n}$, so $U \setminus X$ is a non-empty clopen subset of $C \setminus X$ which contains no closed subset belonging to $X^2_{\alpha+2n-1}$, contradicting $C \setminus X \in C$ $X^2_{\alpha+2n}$; so U \cap X is not $P_{\alpha+2n-1}$. But then U \cap X is not $P_{\alpha+2n}$ either, since by lemma 3.5.4, this would imply that $U \cap X$ is first category, and hence that $U \setminus X \in X^2_{\alpha+2n}$ is Baire, contradicting lemma 3.5.8. It remains to be shown that X contains no closed subsets belonging to $\chi^2_{\alpha+2n}$. So suppose to the contrary that X contains such a subset B. By the argument above, $\overline{B}\setminus B$ is nowhere $P_{\alpha+2n}$; but $\overline{B}\setminus B$ is a closed subset of $C\setminus X \in X^2_{\alpha+2n}$, a con-tradiction with lemma 3.5.1.

3.5.16 LEMMA: (4) and (5) hold.

Proof: By hypothesis (1), $X_{\alpha+2n}^2 \neq \emptyset$, say $X \in X_{\alpha+2n}^2$. Embed X densely in C, then by lemma 3.5.15, $C \setminus X \in X_{\alpha+2n+1}^1$, so $|X_{\alpha+2n+1}^1| \ge 1$. And if $X, Y \in X_{\alpha+2n+1}^1$ are dense subsets of C, then by lemma 3.5.15, $C \setminus X, C \setminus Y \in X_{\alpha+2n}^2$, and hence by hypothesis (1), there exists a homeomorphism h: $C \rightarrow C$ such that $h[C \setminus X]$ = $C \setminus Y$, and thus also h[X] = Y. This proves hypothesis (4). To prove (5), let K' be the Cantor set obtained from hypothesis (2), and put $K = \overline{K' \setminus A}$; then $K \cap A \in X_{\alpha+2n+1}^1$ by lemma 3.5.15.

The next case is m = 2n+1 for some $n < \omega$, and i = 2; we must show that hypotheses (6), (7), and (8) are valid.

3.5.17 LEMMA: (6) holds.

Proof: First, let $X \in X_{\alpha+2n+1}^2$. Then X is $P_{\alpha+2n+1}$, so we can write $X = A \cup B$, where A is $P_{\alpha+2n}$, and B is complete. Put $C \setminus B = \bigcup_{i=0}^{\infty} K_i$, where K_i is compact, and let $L_i = K_i \setminus X$; then L_i is closed in $C \setminus X$, and $C \setminus X = \bigcup_{i=0}^{\infty} L_i$. Since $K_i \cap X$ is a closed subset of A, $K_i \cap X$ is $P_{\alpha+2n}$, and hence L_i is $P_{\alpha+2n+1}$, by lemma 3.5.6. We will show that L_i is nowhere dense in $C \setminus X$, and that L_i does not contain any closed subsets belonging to $X_{\alpha+2n}^2$. Then we will construct closed nowhere dense subsets A_i of $C \setminus X$ such that $L_i \subset A_i \in X_{\alpha+2n+1}^1$; since $|X_{\alpha+2n+1}^1| = 1$ by hypothesis (4), it
then follows from theorem 3.2.4 that $C \setminus X \approx Q \times Y$ for $Y \in X_{\alpha+2n+1}^1$. Now let U be a non-empty clopen subset of C, and suppose that U\X is $P_{\alpha+2n}$.

<u>Case 1</u>: n = 0. Since $U \cap X$ is nowhere P_{α} , $U \setminus X$ is nowhere P_{β} for each $\beta < \alpha$, so $U \setminus X \in X^{2}_{\alpha}$.

<u>Case 2</u>: n > 0. Since U ∩ X is nowhere $P_{\alpha+2n}$, U\X is nowhere $P_{\alpha+2n-1}$ by lemma 3.5.6. Let V be a non-empty clopen subset of U. Since V ∩ X is not $P_{\alpha+2n-1}$, we can apply hypothesis (8) to obtain a closed subset of V\X which belongs to $X^2_{\alpha+2n-1}$; hence U\X $\in X^2_{\alpha+2n}$.

So in both cases, if U\X is $P_{\alpha+2n}$, then U\X $\in X^2_{\alpha+2n}$. But then by hypothesis (3), U \cap X $\in X^1_{\alpha+2n+1}$ is a non-empty clopen subset of X containing no closed subset belonging to $X^2_{\alpha+2n}$, contradicting X $\in X^2_{\alpha+2n+1}$; hence C \X is nowhere $P_{\alpha+2n}$. From this it follows that if U\X were $P_{\alpha+2n+1}$, say U\X = G \cup H, where G is $P_{\alpha+2n}$, and H is complete, then H would be a dense complete subset of U\X, so U \cap X would be first category, contradicting lemma 3.5.8 since U \cap X $\in X^2_{\alpha+2n+1}$. Thus, C \X is nowhere $P_{\alpha+2n+1}$, and, in particular, each L is nowhere dense in C \X.

If B is a closed subset of L_i belonging to $X_{\alpha+2n}^2$, then $\overline{B}\setminus B \in X_{\alpha+2n+1}^1$ by hypothesis (3), hence $\overline{B}\setminus B$ is not $P_{\alpha+2n}$; but on the other hand, $\overline{B}\setminus B = \overline{B} \cap X \subset K_i \cap X \subset A$ is a closed subset of A, and A is $P_{\alpha+2n}$, hence $\overline{B}\setminus B$ is $P_{\alpha+2n}$ by lemma 3.5.1. Thus, no closed subset of L_i belongs to $X_{\alpha+2n}^2$. Fix $i < \omega$, and let \mathcal{D} be a disjoint covering of $(C \setminus X) \setminus L_i$ by clopen subsets of $C \setminus X$ such that for each $D \in \mathcal{D}$, diam(D) < d(D,L_i). Since $C \setminus X$ is nowhere $P_{\alpha+2n}$, it follows from hypothesis (5) that each $D \in \mathcal{D}$ contains a closed subset $E(D) \in X_{\alpha+2n+1}^1$. Then

$$A_i = L_i \cup U_{D \in \mathcal{D}} E(D)$$

is closed in $C \setminus X$ and $P_{\alpha+2n+1}$, hence nowhere dense in $C \setminus X$. Clearly, A_i contains no closed subsets belonging to $X^2_{\alpha+2n}$, and since each open subset of A_i intersects some E(D), A_i is nowhere $P_{\alpha+2n}$; hence $A_i \in X^1_{\alpha+2n+1}$. This proves the first part of the lemma.

Conversely, assume that $C \setminus X \approx Q \times Y$ for $Y \in X_{\alpha+2n+1}^1$, say $C \setminus X = \bigcup_{i=0}^{\infty} A_i$, where each $A_i \in X_{\alpha+2n+1}^1$ is a closed nowhere dense subset of $C \setminus X$. Then

$$\mathbb{X} = \bigcup_{i=0}^{\infty} (\overline{\mathbb{A}}_{i} \setminus \mathbb{A}_{i}) \cup C \setminus (\bigcup_{i=0}^{\infty} \overline{\mathbb{A}}_{i})$$

By hypothesis (3), $\overline{A}_i \setminus A_i \in X^2_{\alpha+2n}$, hence X is strongly $\sigma - P_{\alpha+2n} \cup$ complete is $P_{\alpha+2n+1}$ by corollary 3.5.5; and since $\bigcup_{i=0}^{\infty} (\overline{A}_i \setminus A_i)$ is dense in X, every non-empty clopen subset of X contains a closed subset belonging to $X^2_{\alpha+2n}$. It remains to be shown that X is nowhere $P_{\alpha+2n}$; so let U be a non-empty clopen subset of C.

<u>Case 1</u>: n = 0. Suppose that U ∩ X is P_{β} for some $\beta < \alpha$; then U\X is P_{γ} for some $\gamma < \alpha$, hence for each i, $A_i \cap (U \setminus X)$ is P_{γ} . Since $A_i \cap (U \setminus X) \neq \emptyset$ for some i, this contradicts the fact that A_i is nowhere P_{γ} for each $\gamma < \alpha$; hence U ∩ X is nowhere P_{β} for each $\beta < \alpha$. So if U ∩ X were P_{α} , then it would be first category, hence U\X would be Baire, a contradiction. <u>Case 2</u>: n > 0. Suppose that U ∩ X is $P_{\alpha+2n-1}$; then U\X is $P_{\alpha+2n}$ by lemma 3.5.6, and hence $A_i \cap (U \setminus X)$ is $P_{\alpha+2n}$ for each i. Since some $A_i \cap (U \setminus X)$ is non-empty, this contradicts the fact that $A_i \in X_{\alpha+2n+1}^1$ is nowhere $P_{\alpha+2n}$; hence U ∩ X is nowhere $P_{\alpha+2n-1}$. Again, if U ∩ X were $P_{\alpha+2n}$, hence strongly σ - $P_{\alpha+2n-1}$ by lemma 3.5.4, then U ∩ X would be first category, and as in case 1, we obtain a contradiction.

3.5.18 LEMMA: (7) holds.

Proof: By hypothesis (4), $X_{\alpha+2n+1}^1 \neq \emptyset$, say $Y \in X_{\alpha+2n+1}^1$. Let X be a dense copy of $Q \times Y$ in the Cantor set; then by lemma 3.5.17, $C \setminus X \in X_{\alpha+2n+1}^2$, and hence $X_{\alpha+2n+1}^2 \neq \emptyset$. Now assume that $X_0, X_1 \in X_{\alpha+2n+1}^2$ are dense subsets of the Cantor set; by lemma 3.5.17, and since $|X_{\alpha+2n+1}^1| = 1$, we have $C \setminus X_0 \approx$ $C \setminus X_1 \approx Q \times Y$ for some $Y \in X_{\alpha+2n+1}^1$. By theorem 3.2.8, and hypothesis (4), there exists a homeomorphism h: $C \to C$ such that $h[C \setminus X_0] = C \setminus X_1$; hence $h[X_0] = X_1$.

3.5.19: (8) holds.

Proof: Let $P = P_{\alpha+2n}$, and let Z be the unique element of $X_{\alpha+2n+1}^1$. We will show that the hypotheses of lemma 3.3.4 are satisfied. By lemma 3.5.1, P is closed-hereditary, and P is strongly σ -additive by corollary 3.5.5. Let B be a Borel subset of a compact space Y which is not P; then by hypothesis (5), Y contains a Cantor set K such that $K \setminus B \in X_{\alpha+2n}^2$ and $K \cap B \in$ $X_{\alpha+2n+1}^1$. Then $K \cap B \approx Z$, and $K \setminus B = \bigcup_{i=0}^{\infty} X_i$, where X_i is closed in $K \setminus B$ and P_{β_i} for some $\beta_i < \alpha$ (if n = 0), resp. $P_{\alpha+2n-1}$ (if n > 0, applying lemma 3.5.4). Put $G = \bigcup_{i=0}^{\infty} \overline{X_i}$, then G is σ -compact, and $G \cap B = \bigcup_{i=0}^{\infty} (\overline{X_i} \setminus X_i)$ is P_{α} (if n = 0), resp. strongly $\sigma - P_{\alpha+2n}$ and hence $P_{\alpha+2n}$ (if n >0, applying lemma 3.5.6 and corollary 3.5.5). So by lemma 3.3.4, if A is a Borel subset of a compact space X which is not $P_{\alpha+2n} \cup$ complete, then X contains a Cantor set $K = \overline{K \cap A}$ such that $K \cap A \approx \mathbf{Q} \times Z$. Then $K \setminus A \in$ $X_{\alpha+2n+1}^2$ by lemma 3.5.17. We now consider the case m = 2n+2 for some $n < \omega$, and i = 1 (recall that if m = 0, then we only have the case $(\alpha+m,2)$; we must show that hypotheses (9), (10), and (11) are true for this n.

3.5.20 LEMMA: Let Y be the unique element of $X^1_{\alpha+2n+1}$; then $Q \times Y$ is the unique element of $X^1_{\alpha+2n+2}$, up to homeomorphism.

Proof: We first show that $Q \times Y \in X_{\alpha+2n+2}^1$. Clearly, $Q \times Y = \bigcup_{q \in Q} (\{q\} \times Y)$ is strongly $\sigma - P_{\alpha+2n+1}$ is $P_{\alpha+2n+2}$ by lemma 3.5.4. Note that, since every nonempty clopen subset of $Q \times Y$ contains a closed copy of Y, and since Y is not $P_{\alpha+2n}$, $Q \times Y$ is nowhere $P_{\alpha+2n}$ by lemma 3.5.1. Suppose that $U \times V$ is a non-empty basic clopen subset of $Q \times Y$ which is $P_{\alpha+2n+1}$; then we can write $U \times V = A \cup B$, where A is $P_{\alpha+2n}$, and B is complete. Since $U \times V$ is nowhere $P_{\alpha+2n}$ by the preceding remark, B is dense in $U \times V$, contradicting the fact that $U \times V$ is not Baire; so $Q \times Y$ is nowhere $P_{\alpha+2n+1}$. It remains to be shown that $Q \times Y$ contains no closed subset belonging to $X_{\alpha+2n+1}^2$; so embed $Q \times Y$ as a dense subset of the Cantor set, and suppose to the contrary that $B \in X_{\alpha+2n+1}^2$ is closed in $Q \times Y$. Then by hypothesis (6), $\overline{B} \setminus B \approx Q \times Y$, so $\overline{B} \setminus B$ is not $P_{\alpha+2n+1}$ by the above argument. On the other hand, $\overline{B} \setminus B$ is a closed subset of $C \setminus (Q \times Y)$, and $C \setminus (Q \times Y) \in X_{\alpha+2n+1}^2$ by another application hypothesis (6); so $\overline{B} \setminus B$ is $P_{\alpha+2n+1}$ by lemma 3.5.1, and we have the required contradiction.

To show that $Q \times Y$ is the only element of $\chi^1_{\alpha+2n+2}$, it suffices to show that if $X \in \chi^1_{\alpha+2n+2}$ is a dense subset of C, then $C \setminus X \in \chi^2_{\alpha+2n+1}$; hypothesis (6) then yields that $X \approx Q \times Y$. First note that, since X contains no closed subsets belonging to $\chi^2_{\alpha+2n+1}$, it follows from hypothesis (8) that $C \setminus X$ is $P_{\alpha+2n+1}$. Also, since X is nowhere $P_{\alpha+2n+1}$, $C \setminus X$ is nowhere $P_{\alpha+2n}$ by lemma 3.5.6. And if U is clopen in C, then since $U \cap X$ is not $P_{\alpha+2n}$, we can apply hypothesis (2) and obtain a closed subset of $U \setminus X$ belonging to $\chi^2_{\alpha+2n}$.

3.5.21 LEMMA: (9), (10), and (11) hold.

Proof: (9) follows from (6) and lemma 3.5.20. If $X_0, X_1 \in X_{\alpha+2n+2}^1$ are dense subsets of the Cantor set, then $C \setminus X_0, C \setminus X_1 \in X_{\alpha+2n+1}^2$, so by (7) there exists a homeomorphism h: $C \rightarrow C$ such that $h[C \setminus X_0] = C \setminus X_1$, and hence $h[X_0] = X_1$. (11) follows from (8) and lemma 3.5.20.

Finally, we must show that the hypotheses are satisfied if m = 2n+2 for

some $n < \omega$, and i = 2; i.e. we must show that (1) and (2) hold if n > 0.

3.5.22 LEMMA: Let $n \in N$, and let Y be the unique element of $X_{\alpha+2n-1}^2$; then $Q \times Y$ is the unique element of $X_{\alpha+2n}^2$, up to homeomorphism. Proof: As in the proof of lemma 3.5.20, it can be shown that $Q \times Y$ is $P_{\alpha+2n}$, and nowhere $P_{\alpha+2n-1}$; and since Y is strongly homogeneous, every non-empty clopen subset of $Q \times Y$ contains a closed copy of Y. Hence, $Q \times Y \in X_{\alpha+2n}^2$. Now suppose that $X \in X_{\alpha+2n}^2$. Then X is $P_{\alpha+2n}$, so by lemma 3.5.4, we can write $X = \bigcup_{i=0}^{\infty} X_i$, where each X_i is closed in X, and $P_{\alpha+2n-1}$. Since X is nowhere $P_{\alpha+2n-1}$, each X_i is nowhere dense in X. Fix $i < \omega$, and let D be a disjoint covering of $X \setminus X_i$ by clopen subsets of X such that diam(D) < d(D, X_i) for each $D \in D$. Then each $D \in D$ contains a closed copy E(D) of Y, and as before, it is easily shown that

 $A_i = X_i \cup U_{D \in D} E(D)$

is a closed nowhere dense subset of X such that $A_i \approx Y$. Hence, $X = \bigcup_{i=0}^{\infty} A_i \approx Q \times Y$ by theorem 3.2.4.

3.5.23 LEMMA: If
$$n > 0$$
, then (1) holds.

Proof: By lemma 3.5.22, $|X_{\alpha+2n}^2| = 1$. So suppose that $X_0, X_1 \in X_{\alpha+2n}^2$ are dense in *C*. If Y is the unique element of $X_{\alpha+2n-1}^2$ (hypothesis (7)), then $X_0 \approx X_1 \approx Q \times Y$ by lemma 3.5.22, so by hypothesis (7), we can apply theorem 3.2.8 to obtain a homeomorphism h: $C \rightarrow C$ such that $h[X_0] = X_1$.

3.5.24 LEMMA: If
$$n > 0$$
, then (2) holds.

Proof: Let $P = P_{\alpha+2n}$, let Z' be the unique element of $X^2_{\alpha+2n-1}$, and let Y' be the unique element of $X^1_{\alpha+2n}$. By lemma 3.5.1, P is closed-hereditary, and P is strongly σ -additive by corollary 3.5.5. Let B be a Borel subset of a compact space Y which is not P. Then by lemma 3.5.6, B is not $P_{\alpha+2n-1}$, and hence by hypothesis (11), Y contains a Cantor set K such that $K \setminus B \in X^2_{\alpha+2n-1}$ and $K \cap B \in X^1_{\alpha+2n}$; so $K \setminus B \approx Z'$, and $K \cap B \approx Y'$ is P. Thus, by lemma 3.3.5, if A is a Borel subset of a compact space X which is not $P_{\alpha+2n}$, then there exist Cantor sets B_i in X, for each $i < \omega$, ω , such that $K = (\bigcup_{i=0}^{\infty} B_i)^- \approx C$, and such that for each $i < \omega$,

(i) B_i is nowhere dense in K;

ii)
$$K \setminus A = \bigcup_{i=0}^{\infty} (B_i \setminus A)$$

(iii) $B_0 \setminus A \approx Z \approx (B_{i+1} \setminus B_i) \setminus A$.

As in the proof of lemma 3.4.23, we deduce from theorem 3.2.4 that $K \setminus A \approx Q \times Z$, and hence by lemma 3.5.22, $K \setminus A \in X^2_{\alpha+2n}$.

Since the "if" part of theorem 3.5.11 is trivial, we have completed the proof of theorems 3.5.9, 3.5.10, and 3.5.11.

We finish this section with the analogue to theorem 3.4.25, that all spaces defined above are topologically different:

3.5.25 THEOREM: Let $\beta, \gamma \in [\omega, \omega_1)$, and $i, j \in \{1, 2\}$ be such that i = 2if $\lim(\beta)$, and j = 2 if $\lim(\gamma)$. If $(\beta, i) \neq (\gamma, j)$, then $X_{\beta}^i \cap X_{\gamma}^j = \emptyset$. Furthermore, $X_{\beta}^i \cap \bigcup_{n=1}^{\infty} D_{2n}(\Sigma_2^0) = \emptyset$, so the spaces defined in this section are topologically distinct from those of section 3.4 and Chapter 2.

Proof: If $\beta \neq \gamma$, say $\beta < \gamma$, then $X \in X_{\gamma}^{j}$ is nowhere P_{β} by corollary 3.5.3, so $X \notin X_{\beta}^{i}$. If $\beta = \gamma$, then $i \neq j$, say i = 1, j = 2; hence $X \in X_{\gamma}^{j}$ contains a closed subset belonging to $X_{\gamma-1}^{2}$, whereas $X \in X_{\beta}^{i}$ does not. The last statement is clear since each of the spaces defined in this section is nowhere P_{γ} for $\gamma < \omega$ (corollary 3.5.3).

3.6 The main theorem, part 1

The principal purpose of this section is to show that the spaces, which were defined and characterized in sections 3.4 and 3.5, together with those of Chapter 2, are the only homogeneous zero-dimensional absolute Borel sets of ambiguous class 2.

3.6.1 LEMMA: Let X be homogeneous and zero-dimensional. (a) If $\gamma = -1$ or $\gamma < \omega_1$, $i \in \{1,2\}$, and X is not $P_{\gamma}^{(i)}$, then X is nowhere $P_{\gamma}^{(i)}$. (b) If Y is a strongly homogeneous closed subset of X, then every nonempty clopen subset of X contains a closed copy of Y. Proof: (a) Suppose that U is a non-empty clopen subset of X which is $P_{\gamma}^{(i)}$. Let $x \in U$, and for each $y \in X$, let $h_y: X \to X$ be a homeomorphism such that $h_y(x) = y$. If $\{U_j: j < \omega\}$ is a countable subcovering of $\{h_y[U]: y \in X\}$, then for each $n < \omega$, the clopen set $V_n = U_n \setminus (U_{j < n} U_j)$ is $P_{\gamma}^{(i)}$ by lemma 3.4.1 or 3.5.1. Hence $X = \bigoplus_{n=0}^{\infty} V_n$ is also $P_{\gamma}^{(i)}$. (b) Let U be a non-empty clopen subset of X, $x \in U$, and $y \in Y$. If h: $X \to X$ is a homeomorphism such that h(y) = x, then $h[Y] \cap U$ is a nonempty clopen subset of $h[Y] \approx Y$, so $h[Y] \cap U \approx Y$ since Y is strongly homogeneous. Since h[Y] is closed in X, $h[Y] \cap U$ is closed in X.

We are now ready to prove the main theorem of this chapter. For each $\alpha < \omega_1$, i $\in \{1,2\}$, let $X_{\alpha}^{(i)}$ denote the unique element of $X_{\alpha}^{(i)}$ (theorems 3.4.13 and 3.5.9).

3.6.2 THEOREM: The topological types of homogeneous zero-dimensional absolute Borel sets of ambiguous class 2 are precisely those of C, C \{p}, Q, Q × C, P, the discrete spaces, and the spaces $X_{\alpha}^{(i)}$, for $i \in \{1,2\}$ and $\alpha < \omega_1$.

Proof: From the results of Chapter 2, and sections 3.4 and 3.5, it follows that all these spaces are homogeneous. So let X be a homogeneous zero-dimensional absolute Borel set of ambiguous class 2. By theorem 3.1.2, X $\in D_{\gamma}(\Sigma_2^0)$ for some even $\gamma \in [1, \omega_1)$; let α be minimal in $[1, \omega_1)$ such that α is even and X $\in D_{\alpha}(\Sigma_2^0)$.

<u>Case 1</u>: $\alpha < \omega$. Let $\langle be$ the ordering on the properties $P_m^{(j)}$, defined in theorem 3.4.24, and let $k \in N$, $i \in \{1,2\}$ be such that $P_k^{(i)}$ is minimal with respect to \langle such that X is $P_k^{(i)}$. If X is σ -compact or complete, then either X is discrete, or $X \in \{C, C \setminus \{p\}, Q, Q \times C, P\}$, see the introduction of this monograph; otherwise, X is nowhere $P_{(j)}^{(j)}$ for each $P_m^{(j)} < P_k^{(i)}$ $(m \ge -1, j \in \{1,2\})$ by lemma 3.6.1, so $X \in X_k^{(i)}$ by theorem 3.4.24. <u>Case 2</u>: $\alpha \ge \omega$. Then X is P_{α} . If α is a limit, then X is P_{α} and nowhere P_{β} for each $\beta < \alpha$ by lemma 3.6.1, so $X \in X_{\alpha}^2$. If α is a successor, then $\alpha = \beta+2$ for some $\beta < \omega_1$, and X is $P_{\beta} \cup S_1$ by lemma 3.5.2. First suppose that X is not $P_{\beta} \cup$ complete; then by lemma 3.6.1, X is nowhere $(P_{\beta} \cup \text{complete})$. So if X contains no closed subsets belonging to $X_{\beta+1}^2$, then $X \in X_{\alpha}^1$; otherwise, applying lemma 3.6.1(b), we find that $X \in X_{\alpha}^2$. If on the other hand X is $P_{\beta} \cup \text{complete}$, then since X is nowhere P_{β} by lemma 3.6.1 and by minimality of α , we have in the same way as above that $X \in X_{B+1}^1$ or $X \in X_{B+1}^2$.

3.6.3 COROLLARY: There are exactly ω_1 homogeneous zero-dimensional absolute Borel sets of ambiguous class 2. $\hfill \Box$

Since there are spaces of arbitrarily high level in our subhierarchy of Δ_3^0 , it is impossible to find a single element of Δ_3^0 that contains all zero-dimensional absolute Borel sets of ambiguous class 2 as a closed subspace.

Also, "product theorems" similar to 2.1.3, 2.2.4, 2.3.3, 2.4.3, and 2.4.7 do not hold, i.e. if A is $P_{\gamma}^{(i)}$ in C, then it is not necessarily true that $A \times X_{\gamma}^{(i)} \approx X_{\gamma}^{(i)}$. For example, since S is neither σ -compact nor complete, by theorem 3.3.1 it contains a closed copy of P and a closed copy of Q; so $S \times S$ contains a closed copy of $Q \times P$, and thus $S \times S \not\approx S$ since $Q \times P$ is not σ -compact \cup complete.

However, we do have the following theorem.

(a) If $n \in N$, $i \in \{1,2\}$, and A is $P_n^{(i)}$, then A can be embedded as a closed subspace of $X_n^{(i)}$.

(b) If $\alpha \in [\omega, \omega_1)$, and A is P_{α} , then A can be embedded as a closed subspace of X_{α}^2 .

(c) If $\alpha \in [\omega, \omega_1)$ is a successor, A is P_{α} , and A does not contain any closed copies of $X_{\alpha-1}^2$, then A can be embedded as a closed subspace of X_{α}^1 . Proof:(a) For each $k \in N$, let Y_k be a dense copy of $X_{\alpha}^{(i)}$ in $\{1/k\} \times C$, and consider A as a subset of $\{0\} \times C$. We claim that, as a subset of $[0,1] \times C$,

$$\widetilde{A} = A \cup \bigcup_{k=1}^{\infty} Y_k \approx X_n^{(i)}$$
.

Indeed, since A is nowhere dense in \widetilde{A} , \widetilde{A} has the same "nowhere" properties as $X_n^{(i)}$, so it suffices to show that \widetilde{A} is $P_n^{(i)}$. Since \widetilde{A} is strongly σ - $P_n^{(i)}$, this is trivial if n is odd, by corollary 3.4.3. If n is even, then $P_n^{(i)} \equiv P_{n-3}^{(i)} \cup$ complete; so write $A = A_0 \cup A_1$, $Y_k = Y_k^0 \cup Y_k^1$, where A_0, Y_k^0 are $P_{n-3}^{(i)}$, and A_1, Y_k^1 are complete. Then $A_0 \cup \bigcup_{k=1}^{\infty} Y_k^0$ is strongly σ - $P_{n-3}^{(i)}$ is $P_{n-3}^{(i)}$ by corollary 3.4.3, and $A_1 \cup \bigcup_{k=1}^{\infty} Y_k^1 = A_1 \cup \bigoplus_{k=1}^{\infty} Y_k^1$ is complete \cup complete is complete; so \widetilde{A} is $P_{n-3}^{(i)} \cup$ complete is $P_n^{(i)}$. It is clear that $A = \widetilde{A} \cap (\{0\} \times C)$ is closed in \widetilde{A} . (b) Define \widetilde{A} as above, with Y_k a copy of X_α^2 . Again, \widetilde{A} is nowhere P_β for $\beta < \alpha$, and if α is a successor, then each non-empty clopen subset of \widetilde{A} contains a closed copy of $X_{\alpha-1}^2$, since it intersects some Y_k . If α is even, then \widetilde{A} is P_α by corollary 3.5.5, and if α is odd, then $P_\alpha \equiv P_{\alpha-1} \cup$ complete, so since $P_{\alpha-1}$ is strongly σ -additive, we can argue as under (a). Also as in (a), A is closed in \widetilde{A} .

(c) As in (b), we can show that \widetilde{A} is P_{α} and nowhere $P_{\alpha-1}$. If B is a closed copy of $X_{\alpha-1}^2$ in \widetilde{A} , then $B \notin A$, so $B \cap Y_k \neq \emptyset$ for some k. However, $B \cap Y_k = B \cap (\{1/k\} \times C)$ is clopen in B, so $B \cap Y_k$ is a closed copy of $X_{\alpha-1}^2$ in Y_k , a contradiction. Thus, $\widetilde{A} \approx X_{\alpha}^1$, and A is closed in \widetilde{A} .

The aim of van Douwen in constructing T was to give an example of a homogeneous absolute Borel set which cannot be given the structure of a topological group. We conclude this chapter with the theorem that neither S nor T admits a group structure; the proof is taken from van Mill [40]. We do not have an answer to the general problem of which of the spaces characterized in this chapter can be made into a topological group.

3.6.5 THEOREM: Suppose that X is the union of a σ -compact subset and a complete subset, and X is neither σ -compact nor complete. Then X does not admit the structure of a topological group.

Proof: Write $X = A \cup B$, where A is σ -compact and B is complete. Since X is not complete, by theorem 3.3.1, X contains a closed copy Q of Q. If there is a group structure on X, and we let \widetilde{A} be the subgroup generated by $A \cup Q$, then \widetilde{A} is σ -compact. Since X is not σ -compact, X\ \widetilde{A} is non-empty, say $x \in X \setminus \widetilde{A}$. Then the coset $xQ \approx Q$ is closed in X, and contained in $X \setminus \widetilde{A} \subset B$. But B is complete, a contradiction.

3.6.6 COROLLARY: S and T are homogeneous absolute Borel sets that do not admit the structure of a topological group. $\hfill \Box$

CHAPTER 4: BOREL SETS OF HIGHER CLASS

In Chapter 3, we used the hierarchy of small Borel classes in Δ_3^0 to characterize all homogeneous zero-dimensional absolute Borel sets of ambiguous class 2. However, when extended to classes Δ_{α}^{0} for $\alpha < \omega_{1}$, this hierarchy does not suffice to describe all homogeneous Borel sets in C in a way similar to the results of Chapter 3; in other words, the hierarchy of small Borel classes is too coarse to distinguish between all homogeneous zero-dimensional absolute Borel sets. Thus, we have to make further refinements in our hierarchy. And again, we are so fortunate that a description of such a hierarchy is already available: the so-called Wadge hierarchy of Borel sets, developed by Wadge in [60] (see Wadge [59] or Moschovakis [43]), suits our needs so well that it seems to have been defined just for the purpose of characterizing homogeneous zero-dimensional Borel sets. Since the reader is not assumed to be familiar with the theory of Wadge classes, we discuss the necessary background in section 4.1; among others, we present the interpretation of the Wadge hierarchy in terms of games, which enables one to prove very strong results with very little work (modulo a highly non-trivial, but well-known result, viz. determinacy of Borel games, Martin [35]). Also, we state a theorem of Steel [56], which will serve in many cases to prove that there exists at most one zero-dimensional space of a given type. In section 4.2, we discuss an inductive construction process of the Wadge classes, similar to (but much more complicated than) the usual definition of the Borel classes given in 1.10; this construction process is due to Louveau [32].

Then, in section 4.3, we give a definition of certain classes of topological spaces, based on Louveau's description of the Wadge hierarchy of Borel sets; we determine which of the classes are non-empty, and, using Steel's theorem, we show that the non-empty ones contain exactly one element, up to homeomor-

73

phism, and that this space is homogeneous. In section 4.4, we show that every homogeneous zero-dimensional absolute Borel set which is not in Δ_3^0 belongs to one of the classes of section 4.3 and hence is topologically characterized by the properties describing the class.

Thus, we have obtained topological characterizations of *all* homogeneous zerodimensional absolute Borel sets; it should be noted that the results of Chapter 3 provided us with *internal* topological characterizations, but that in general, this is *not* the case for the descriptions in terms of Wadge classes obtained in the present chapter.

Since the results in this chapter are rather abstract, we take a closer look at some special Wadge classes in section 4.5; in that section, we also prove some more Hurewicz-type theorems, and we partially answer a question of Sikorski from [54].

Finally, in section 4.6, we will exhibit some connections between the results of Chapters 2 and 3 and the Wadge hierarchy.

The main results of this chapter are taken from van Engelen [11].

4.1 The Wadge hierarchy of Borel sets

This section covers the basic notions from game theory; we list the main results about Wadge games and Wadge classes. A more detailed discussion can be found in Moschovakis [43] and Van Wesep [61], or in the unpublished dissertations of Wadge [60] (see also [59]), Steel [55] and Van Wesep [62]; a nice presentation of the subject is also given in Martin and Kechris [36].

Let X be a space, and let $A \subset X^{\omega}$. The game $G_{X}(A)$ is defined as follows: there are two players, I and II, who alternately choose elements of X, in this way:

 $\prod_{11}^{I} \gamma_{0} \gamma_{1} \gamma_{2} \gamma_{3} \gamma_{4} \gamma_{5} \cdots \gamma_{...}$

In the end, we find a point $\gamma = (\gamma_n)_{n < \omega} \in X^{\omega}$; I wins if $\gamma \in A$, II wins if $\gamma \notin A$. Since I and II are assumed to have knowledge of all previous moves, these games were called *infinite games with perfect information* by Gale and Stewart [18].

A strategy for player I is a function σ assigning an element of X to each finite sequence of elements of X with even length; player I follows the strategy σ if for each n > 0, his n^{th} move (i.e. γ_{2n}) is $\sigma(\gamma_0, ..$ \dots, γ_{2n-1}). Thus, a strategy prescribes the moves, depending only on previous moves. Similarly, a strategy for player II is a function τ assigning an element of X to each finite sequence of elements of X of odd length; player II follows τ if $\gamma_{2n+1} = \tau(\gamma_0, \dots, \gamma_{2n})$ for all $n < \omega$. If player I follows σ and player II follows τ in the above game, then we write $\gamma = \sigma \star \tau$. We say that σ is a winning strategy for player I if, for each strategy τ of II, $\sigma \star \tau \in A$; τ is a winning strategy for player II if, for each strategy σ of I, $\sigma \star \tau \notin A$. The game $G_{\chi}(A)$ is determined if either player I or player II has a winning strategy. We have the following famous theorem, due to Martin:

4.1.1 THEOREM (Martin [35]): If A is a Borel subset of X^{ω} , then $G_{\chi}(A)$ is determined.

Some special cases of this theorem were established earlier by Gale and Stewart [18], Wolfe [63], Davis [6], and Paris [45]. For a discussion of the *axiom of determinacy*, i.e. the statement (contradictory with the axiom of choice) that all games $G_{\omega}(A)$ are determined, see Moschovakis [43] and the references cited therein.

We will now describe a slightly different type of game, the *Lipschitz game* $G_{\ell}^{X}(A,B)$, where $A,B \subset X^{\omega}$. Again, we have two players, I and II, taking turns playing points of X, as follows:

I II

$$\alpha_0 \qquad \beta_0 \\ \alpha_1 \qquad \beta_1 \\ \alpha_2 \qquad \beta_2 \\ \vdots \qquad \vdots$$

In the end, I has played $\alpha = (\alpha_n)_{n < \omega}$, and II has played $\beta = (\beta_n)_{n < \omega}$; II wins $G_{\ell}^{X}(A,B)$ if and only if $(\alpha \in A \Leftrightarrow \beta \in B)$, otherwise I wins. Note that if h: $X^{\omega} \times X^{\omega} \approx X^{\omega}$ is defined by $h(x,y)_{2n} = x_n$, $h(x,y)_{2n+1} = y_n$ for each $n < \omega$, then II wins $G_{\ell}^{X}(A,B)$ if and only if $(\alpha,\beta) \in (A \times B) \cup (X^{\omega} \setminus A \times X^{\omega} \setminus B)$ if and only if $h(\alpha,\beta) \in E = h[(A \times B) \cup (X^{\omega} \setminus A \times X^{\omega} \setminus B)]$ if and only if II wins $G_{\chi}^{X}(X^{\omega} \setminus E)$; in particular, if A and B are Borel subsets of X^{ω} , then $G_{\ell}^{X}(A,B)$ is determined by theorem 4.1.1. If player I follows the strategy σ in $G_{\ell}^{X}(A,B)$, and ends up having played

a against II's β , then we write $\sigma * [\beta] = \alpha$; similarly, if II follows τ , then we write $[\alpha] * \tau = \beta$. Define $f_{\tau} : X^{\omega} \to X^{\omega}$ by $f_{\tau}(\alpha) = [\alpha] * \tau$, and g_{σ} : $X^{\omega} \to X^{\omega}$ by $g_{\sigma}(\beta) = \sigma \star [\beta]$. If X is discrete, then since an initial segment of $f_{\tau}(\alpha)$, resp. $g_{\sigma}(\beta)$, only depends upon an initial segment of α , resp. β , both f_{τ} and g_{σ} are continuous. Furthermore, we have

 τ is a winning strategy for player II if and only if $(\forall \alpha : \alpha \in A \Leftrightarrow [\alpha] * \tau \in B)$ if and only if $(\forall \alpha : \alpha \in A \Leftrightarrow f_{\tau}(\alpha) \in B)$ if and only if $A = f_{\tau}^{-1}[B]$; and

σ is a winning strategy for player I if and only if (∀β: σ*[β] ∉ A ↔ β ϵ B) if and only if (∀β: g_σ(β) ∉ A ↔ β ϵ B) if and only if B = g_σ⁻¹[X^ω\A]. Thus we have proved: if X is discrete, and A,B are Borel subsets of X^ω, then either there is a continuous f: X^ω → X^ω such that A = f⁻¹[B] or there is a continuous g: X^ω → X^ω such that B = g⁻¹[X^ω\A]. The case X = ω is the so-called "Wadge lemma".

Remark: One might wonder, whether the existence of a continuous f: $X^{\omega} \to X^{\omega}$ satisfying $A = f^{-1}[B]$, where X is discrete, yields a winning strategy τ for II in the game $G_{\ell}^{X}(A,B)$. This is not the case. However, the mapping f_{τ} defined above has the special property that an initial segment of length n of $f_{\tau}(\alpha)$ is determined by an initial segment of length n of α . In other words, if X^{ω} is given the Baire space metric $\rho(\alpha,\tilde{\alpha}) = 1/(\min\{k:$ $\alpha(k) \neq \tilde{\alpha}(k)\} + 1)$ if $\alpha \neq \tilde{\alpha}$, then f_{τ} satisfies the Lipschitz condition $\rho(f_{\tau}(\alpha), f_{\tau}(\tilde{\alpha})) \leq \rho(\alpha, \tilde{\alpha})$ (whence the name *Lipschitz game*). It is not hard to show that each f: $X^{\omega} \to X^{\omega}$ with $A = f^{-1}[B]$ and $\rho(f(\alpha), f(\tilde{\alpha})) \leq \rho(\alpha, \tilde{\alpha})$ for all $\alpha, \tilde{\alpha} \in X^{\omega}$ is some f_{τ} , where τ is a winning strategy for II in the game $G_{\ell}^{X}(A,B)$.

If f is just continuous, then in order to determine an initial segment of length n of $f(\alpha)$ we might need a much longer initial segment of α . In terms of games, player II might have to know more than n moves of player I before being able to play *his* nth move such that he wins. And indeed, it can be shown that the existence of a continuous f: $X^{\omega} \rightarrow X^{\omega}$ with $A = f^{-1}[B]$ is equivalent to II having a winning strategy in the *Wadge game* $G_{W}^{X}(A,B)$, which is played just as the Lipschitz game, granting however player II the possibility of passing.

If $X = 2 = \{0,1\}$, then we write $A \leq_{W} B$ (A is Wadge-reducible to B) for "A = $f^{-1}[B]$ for some continuous $f: X^{\omega} \to X^{\omega}$ ". So we have:

4.1.2 THEOREM (Wadge [59]): If A, B are Borel subsets of 2^{ω} , then either $A \leq_{w} B$ or $B \leq_{w} 2^{\omega} \setminus A$.

The reader should realize that here it is important to consider subsets of a particular copy of the Cantor set, and not the topological type; the relation $A \leq_{W} B$ is not the same as "for certain embeddings $\widetilde{A}, \widetilde{B}$ of A, B in certain Cantor sets C_0, C_1 , we have a continuous mapping g: $C_0 \rightarrow C_1$ such that $\widetilde{A} = g^{-1}[\widetilde{B}]$ ". For example, if $B = 2^{\omega}$, then 2^{ω} itself is the only copy A of the Cantor set for which $A \leq_{W} B$ (see also theorems 4.6.6 and 4.6.8). This is the reason that, to avoid inaccuracies, IN THIS CHAPTER WE USE 2^{ω} AS THE CANTOR SET.

We will now show how the relation \leq_{W} gives rise to a refinement of the Borel hierarchy in 2^{ω} . It will be convenient to have available the following notions:

4.1.3 DEFINITION: Let X be a space, and $\Gamma \subset P(X)$. (a) $\stackrel{V}{\Gamma} = \{X \setminus A: A \in \Gamma\}$ is the dual class of $\Gamma; \Delta(\Gamma) = \Gamma \cap \stackrel{V}{\Gamma}$ is the ambiguous class associated with Γ . (b) Γ is self-dual if $\Gamma = \stackrel{V}{\Gamma} (= \Delta(\Gamma))$. (c) Γ is continuously closed if for each $A \in \Gamma$, and each continuous f: X $\Rightarrow X$, also $f^{-1}[A] \in \Gamma$.

4.1.4 DEFINITION: (a) $A \equiv_{W} B$ if and only if both $A \leq_{W} B$ and $B \leq_{W} A$. (b) $A <_{W} B$ if and only if $A \leq_{W} B$ but $B \neq_{W} A$.

4.1.5 LEMMA: Let A and B be Borel subsets of 2^{ω} . (a) If $A \leq_{w} B$, then $A <_{w} 2^{\omega} \setminus B$. (b) Either $A <_{w} B$, or $B <_{w} A$, or $A \equiv_{w} B$, or $2^{\omega} \setminus A \equiv_{w} B$. Proof: (a) Since $B \neq_{w} A$, $A \leq_{w} 2^{\omega} \setminus B$ by theorem 4.1.2. If $2^{\omega} \setminus B \equiv_{w} A$, then $B \leq_{w} 2^{\omega} \setminus A \leq_{w} 2^{\omega} \setminus B \leq_{w} A$, a contradiction. (b) By theorem 4.1.2, either $A \leq_{w} B$ or $B \leq_{w} 2^{\omega} \setminus A$. Case 1: $A \leq_{w} B$. Then either $B \leq_{w} A$, whence $A \equiv_{w} B$; or $B \neq_{w} A$, whence $A \leq_{w} B$. Case 2: $B \leq_{w} 2^{\omega} \setminus A$. Then either $2^{\omega} \setminus A \leq_{w} B$, whence $2^{\omega} \setminus A \equiv_{w} B$; or $2^{\omega} \setminus A \neq_{w} B$, whence $B <_{w} 2^{\omega} \setminus A$, and hence $B <_{w} A$ by (a).

The = -equivalence classes are called *Wadge degrees*. Since we want to con-

struct a hierarchy, we prefer classes that are "closed downwards", i.e. continuously closed; these classes are called *Wadge classes*:

4.1.6 DEFINITION: (a) If A is a subset of 2^{ω} , then the Wadge class of A is $[A] = \{B \in 2^{\omega}: B \leq A\}.$ (b) If A is a Borel subset of 2^{ω} , then [A] is called a Borel Wadge class.

Note that a Borel Wadge class consists exclusively of Borel sets. By the discussion following theorem 4.1.2, it is not always true that if B ϵ [A] and B \approx \tilde{B} , then \tilde{B} ϵ [A].

4.1.7 DEFINITION: The Wadge ordering < on dual pairs $\{\Gamma, \overset{Y}{\Gamma}\}$ of Wadge classes is defined by $\{\Gamma_0, \overset{Y}{\Gamma}_0\} < \{\Gamma_1, \overset{Y}{\Gamma}_1\}$ if and only if $\Gamma_0 \subsetneqq \Gamma_1$.

Since $[A] \not\subseteq [B]$ if and only if $A <_{w} B$, it follows from lemma 4.1.5(a) that for Borel Wadge classes, if $\{\Gamma_0, \Gamma_0\} < \{\Gamma_1, \Gamma_1\}$, then $\Gamma_0 \not\subseteq \Gamma_1$ and $\Gamma_0 \not\subseteq \Gamma_1$, hence $\Gamma_0 \not\subseteq \Gamma_1$ and $\Gamma_0 \not\subseteq \Gamma_1$. We will often abuse notation and write $\Gamma_0 < \Gamma_1$ for $\Gamma_0 \not\subseteq \Gamma_1$, etc..

A translation of lemma 4.1.5(b) into Wadge classes yields:

4.1.8 LEMMA: < is a linear ordering on the pairs of Borel Wadge classes.

In fact, we have the following theorem, whose proof is beyond the scope of this section.

4.1.9 THEOREM (Wadge [59]): < well-orders the pairs of Borel Wadge classes.

The analogues of the above results for ω^{ω} instead of 2^{ω} were all established by Wadge in his thesis [60]; theorem 4.1.9, for ω^{ω} , was announced by Wadge [59] in the 1972 Notices of the AMS, under the assumption of Borel determinacy (which was only proved in 1975). Martin has shown that the axiom of determinacy implies that < well-orders all Wadge classes (for a proof, see Martin and Kechris [36], Moschovakis [43], or Van Wesep [61]).

Remark: The reason that we have presented the above discussion for 2^{ω} rather than ω^{ω} is, that for our characterization of homogeneous Borel sets, we heavily rely on the theorem of Steel, which will be stated at the end of this section; this theorem deals with classes of subsets of 2^{ω} , not of ω^{ω} .

For ω^{ω} , Van Wesep [61] has shown that the pattern of the Borel Wadge classes is as follows: the first element is $\{\{\emptyset\}, \{\omega^{\omega}\}\}$; a successor is self-dual if and only if its predecessor is not; at limit stages of cofinality ω stands a self-dual class, and at limit stages of cofinality ω_1 a non-self-dual pair. Furthermore, the predecessor of the non-self-dual pair $\{\Gamma, \Gamma\}$ is $\Delta(\Gamma)$. Thus we have:



As we shall see in section 4.2, the only difference between the situation for ω^{ω} and that for 2^{ω} is that in the Borel Wadge ordering on 2^{ω} , the limit stages of cofinality ω are occupied by non-self-dual pairs.

Remark: Wadge [59] has proved that the order-type of the Wadge ordering on pairs of Borel Wadge classes in ω^{ω} is $\epsilon_1^{\omega_1}$, where for each pair of ordinals γ,μ we define ϵ^{μ}_{γ} as follows:

- (1) ε⁰_γ is the γth ε-number (δ is an ε-number if ω^δ = δ, see Rubin [48] or Sierpiński [53])
 (2) for μ > 0, ε^μ_γ is the γth ordinal number δ such that ε^ν_δ = δ for all ν < μ.

The cardinality of $\varepsilon_1^{\omega_1}$ is ω_1 . In section 4.2, we will see that the ordertype of the Borel Wadge ordering on 2^{ω} is $\varepsilon_1^{\omega_1}$ as well.

Remark: The fact that the Wadge hierarchy really refines the Borel hierarchy can be easily deduced from the observation that if Γ is a continuously closed class of Borel sets, and $A \in \Gamma \setminus \Gamma$, then $\Gamma = [A]$. Indeed, clearly $[A] \subset \Gamma$; and if $B \in \Gamma$, then by theorem 4.1.2 we have either $B \leq_{W} A$ or $A \leq_{W} 2^{\omega} \setminus B$. However, since $2^{\omega} \setminus B \in \Gamma$, and since Γ is continuously closed if Γ is, $A \leq_{W} 2^{\omega} \setminus B$ would imply that $A \in \Gamma$, a contradiction; so $B \in [A]$. Since Borel sets of exact class α exist for each $\alpha < \omega_1$ (see also section 4.5), this implies that each Σ_{α}^{0} and each Π_{α}^{0} is a Wadge class. The same method can be used to prove that in fact the Wadge hierarchy of Borel sets refines the hierarchy of small Borel classes $D_{\alpha}(\Sigma_{F}^{0})$.

The last part of this section is devoted to a theorem of Steel [56]. Let $Q_i = \{x \in 2^{\omega}: \exists m \forall n \ge m: x_n = i\}$, for $i \in \{0,1\}$; then $Q_0 \approx Q \approx Q_1$. If $x \notin Q_0 \cup Q_1$, then x consists of blocks of zeros separated by blocks of ones; define $\phi: 2^{\omega} \setminus (Q_0 \cup Q_1) \rightarrow 2^{\omega}$ by $\phi(x)_n = 0$ if the n^{th} block of zeros in x has even length, $\phi(x)_n = 1$ otherwise (we start counting with the 0^{th} block). Note that ϕ is continuous.

4.1.10 DEFINITION: (a) A class $\Gamma \subset P(2^{\omega})$ is reasonably closed if Γ is continuously closed, and $\phi^{-1}[A] \cup Q_0 \in \Gamma$ for each $A \in \Gamma$. (b) A subset A of 2^{ω} is everywhere properly Γ if for each non-empty open subset U of 2^{ω} , we have $U \cap A \in \Gamma \setminus \Gamma$.

Note that in (b), if $\emptyset \in \overset{V}{\Gamma}$, then A is dense in 2^{ω}. We are now ready to state Steel's theorem (in fact, a special case of it).

4.1.11 THEOREM (Steel [56]): If Γ is a reasonably closed class of Borel subsets of 2^{ω} , and $A, B \subset 2^{\omega}$ are everywhere properly Γ and either both first category or both Baire, then h[A] = B for some autohomeomorphism h of 2^{ω} .

In sections 4.4, 4.5, and 4.6, it will be made clear to what kind of classes Steel's theorem can be applied.

4.2 An inductive description of the Borel Wadge classes

This section is devoted to a discussion of part of Louveau's paper [32]; in that part of his paper, Louveau provides an inductive definition of the Borel Wadge classes in ω^{ω} . Since we want to work inside the Cantor set 2^{ω} , we briefly indicate what changes are to be made in order to obtain analogous results for 2^{ω} . Finally, using this description of the Borel Wadge classes, we prove some closure properties of these classes, which will be used in sections 4.3 and 4.4.

Unfortunately, the inductive defintion is rather complicated, and we need many new notions; all are taken from Louveau [32], except that we define them for 2^{ω} instead of ω^{ω} .

One of the operations we use is the difference operation, which was defined in definition 3.1.1. However, since we will need results about the Wadge hierarchy, and since Wadge classes are not defined for arbitrary spaces, it is not convenient to consider arbitrary classes $D_{\eta}^{X}(\Sigma_{\xi}^{0})$, or the absolute class $D_{\eta}(\Sigma_{\xi}^{0})$: we have to work with $D_{\eta}^{2\omega}(\Sigma_{\xi}^{0})$. To simplify notation, therefore, IN THIS CHAPTER, $D_{\eta}(\Sigma_{\xi}^{0})$ DENOTES $D_{\eta}^{2\omega}(\Sigma_{\xi}^{0})$

Similarly, all the other operations will only be explicitly defined for 2^{ω} .

4.2.1 DEFINITION: Let
$$\Gamma, \Gamma' \subset P(2^{\omega})$$
, and let $A \subset 2^{\omega}$.
(a) $A \in \text{Sep}(D_n(\Sigma_F^0), \Gamma)$ if and only if

$$A = (A_0 \cap C) \cup (A_1 \setminus C)$$

for some $C \in D_{\eta}(\Sigma_{\xi}^{0})$, $A_{0} \in \Upsilon$, and $A_{1} \in \Gamma$. (b) $A \in Bisep(D_{\eta}(\Sigma_{\xi}^{0}), \Gamma, \Gamma')$ if and only if

$$A = (A_0 \cap C_0) \cup (A_1 \cap C_1) \cup B \setminus (C_0 \cup C_1)$$

for some disjoint $C_0, C_1 \in D_{\eta}(\Sigma_{\xi}^0)$, and some $A_0 \in \Gamma$, $A_1 \in \Gamma$, and $B \in \Gamma'$. (c) $A \in SU(\Sigma_{\xi}^0, \Gamma)$ if and only if

$$A = \bigcup_{n=0}^{\infty} (A_n \cap C_n)$$

for some collection $\{C_n: n < \omega\}$ of pairwise disjoint Σ^0_{ξ} -sets in 2^{ω} , and some family $\{A_n: n < \omega\}$ of elements of Γ .

The set $\bigcup_{n=0}^{\infty} C_n$ is called the envelop of A. If the envelop of A is all of 2^{ω} (i.e. $\{C_n: n < \omega\}$ partitions 2^{ω}), then we write $A \in PU(\Sigma_{\xi}^0, \Gamma)$. (d) $A \in SD_n(\langle \Sigma_{\xi}^0, SU(\Sigma_{\xi}^0, \Gamma) \rangle, \Gamma')$ if and only if

$$A = U_{\zeta < \eta} (A_{\zeta} \setminus (U_{\beta < \zeta} C_{\beta})) \cup B \setminus (U_{\zeta < \eta} C_{\zeta})$$

for some increasing sequences $\langle A_{\zeta}: \zeta < n \rangle$ of elements of $SU(\Sigma_{\xi}^{0},\Gamma)$, and $\langle C_{\zeta}: \zeta < n \rangle$ of Σ_{ξ}^{0} -sets in 2^{ω} such that $A_{\zeta} \subset C_{\zeta} \subset A_{\zeta+1}$ and C_{ζ} is the envelop of A_{γ} , and some $B \in \Gamma'$.

In (b) and (d), we always omit Γ' if $\Gamma' = \{\emptyset\}$.

To simplify the exposition, if we write e.g. "A ϵ Sep $(D_{\eta}(\Sigma_{\xi}^{0}),\Gamma)$, say A = $(A_{0} \cap C) \cup (A_{1} \setminus C)$ ", then it is tacitly understood that the sets A_{0} , A_{1} , and C are chosen as required by the above definition.

Louveau now selects a certain subset D of $(\omega_1)^{\omega}$, its elements being called *descriptions*, and for each $u \in D$, he defines a *non-self-dual Borel Wadge class* Γ_u , according to the following inductive definition (where sometimes $v \in (\omega_1)^{\omega}$ is considered as a pair $\langle v_0, v_1 \rangle$ or a sequence $\langle v_n : n < \omega \rangle$ of elements of $(\omega_1)^{\omega}$; $\underline{0} \in (\omega_1)^{\omega}$ has all coordinates 0):

4.2.2 DEFINITION: (a) $\underline{0} \in D$, $\Gamma_{\underline{0}} = \{\emptyset\}$. (b) If $u = \xi^{1} \cap \underline{0}$, where $\xi \ge 1$, $n \ge 1$, then $u \in D$, and $\Gamma_{u} = D_{\eta}(\Sigma_{\xi}^{0})$. (c) If $u = \xi^{2} \cap u^{*}$, where $\xi \ge 1$, $n \ge 1$, $u^{*} \in D$, and $u^{*}(0) > \xi$, then $u \in D$, and $\Gamma_{u} = \operatorname{Sep}(D_{\eta}(\Sigma_{\xi}^{0}), \Gamma_{u^{*}})$. (d) If $u = \xi^{3} \cap (u_{0}, u_{1})$, where $\xi \ge 1$, $n \ge 1$, $u_{0}, u_{1} \in D$, $u_{0}(0) > \xi$, $u_{1}(0) \ge \xi$ or $u_{1} = \underline{0}$, and $\Gamma_{u_{1}} < \Gamma_{u_{0}}$, then $u \in D$, and $\Gamma_{u} = \operatorname{Bisep}(D_{\eta}(\Sigma_{\xi}^{0}), \Gamma_{u_{0}}, \Gamma_{u_{1}})$. (e) If $u = \xi^{4}(u_{n}: n < \omega)$, where $\xi \ge 1$, each $u_{n} \in D$, $\Gamma_{u_{n}} < \Gamma_{u_{n+1}}$ for each n, $\langle u_{n}(0): n < \omega \rangle$ is non-decreasing, and $\sup_{n < \omega} u_{n}(0) > \xi$, then $u \in D$, and $\Gamma_{u} = \operatorname{SU}(\Sigma_{\xi}^{0}, \bigcup_{n=0}^{\infty} \Gamma_{u_{n}})$. (f) If $u = \xi^{5} \cap \langle u_{0}, u_{1} \rangle$, where $\xi \ge 1$, $n \ge 2$, $u_{0}, u_{1} \in D$, $u_{0}(0) = \xi$, $u_{0}(1) = 4$, $u_{1}(0) \ge \xi$ or $u_{1} = \underline{0}$, and $\Gamma_{u_{1}} < \Gamma_{u_{0}}$, then $u \in D$, and $\Gamma_{u} = \operatorname{SD}_{\eta}(\langle \Sigma_{\xi}^{0}, \Gamma_{u_{0}}, \Gamma_{u_{1}})$.

Note that in part (f), since $u_0(0) = \xi$ and $u_0(1) = 4$, we have that $\Gamma_{u_0} = SU(\Sigma_{\epsilon}^0, \Gamma)$ for some Γ , so that Γ_{u_1} is well-defined (definition 4.2.1(d)).

Remark: The restrictions that we put on $\langle u_n: n < \omega \rangle$ in case $u = \xi^4 \langle u_n: n < \omega \rangle \epsilon D$ are slightly different from those in Louveau [32], definition 1.2(e); it can be seen that the definition in Louveau does not yield the desired results. For example, if $\Gamma_{u_0} = \operatorname{Sep}(\Sigma_1^0, \Sigma_{\omega}^0)$, $\Gamma_{u_n} = \Sigma_{n+1}^0$ for $n \epsilon N$, and $u = 1^4 \langle u_n: n < \omega \rangle$, then u would be a description for Louveau; however, $\Gamma_u = \operatorname{SU}(\Sigma_1^0, \bigcup_{n=0}^{\infty} \Gamma_{u_n}) = \operatorname{SU}(\Sigma_1^0, \Gamma_{u_0}) = \Gamma_{u_0}$ by Louveau [32], lemma 1.4 (see lemma 11(a) of this section). This contradicts Louveau's own lemma 1.23.

Remark: If we put $D_0 = \{\underline{0}\} \cup \{\xi^{1}, n \in [1, \omega_1)\} \subset (\omega_1)^{\omega}, D_{\alpha+1} = D_{\alpha} \cup \{u \in (\omega_1)^{\omega}: u \text{ is defined as in 4.2.2(c)-(f), with } u^*, u_0, u_1, \text{ and } \langle u_n: n < \omega \rangle$ elements of D_{α} , and $D_{\beta} = \bigcup_{\alpha < \beta} D_{\alpha}$ if $\lim(\beta)$, then $D = D_{\omega_1}$. Furthermore, since we have increasing sequences $\langle \Gamma_{u_n}: n < \omega \rangle$ in 4.2.2(e), each set $\{\Gamma_u: u \in D_{\alpha}\}$ has cardinality ω_1 , so $|\{\Gamma_u: u \in D\}| = \omega_1$.

For each description u, the *type* $t(u) \in \{0,1,2,3\}$ of u is defined (inductively) as follows:

4.2.3 DEFINITION: Let u be a description.
(a) If u = 0, then t(u) = 0.
(b) If u = ξ¹(η), then t(u) = 1 if n is a successor, and t(u) = 2 if n is a limit.
(c) If u(1) = 2, then t(u) = 3.
(d) If u = ξ³(η)(u₀, u₁), then t(u) = 1 if u₁ = 0 and n is a successor, t(u) = 2 if u₁ = 0 and n is a limit, t(u) = 2 if u₁ = 0 and n is a limit, t(u) = t(u₁) if u₁(0) = ξ, and t(u) = 3 if u₁(0) > ξ.
(e) If u(1) = 4, then t(u) = 2.
(f) If u = ξ⁵(η)(u₀, u₁), then t(u) = 2 if u₁ = 0, t(u) = t(u₁) if u₁(0) = ξ, and t(u) = ξ if u₁(0) > ξ.

For the meaning of the type of a description u in relation with the ordinal of $\{\Gamma_{u}, \overset{\vee}{\Gamma}_{u}\}$ in the Wadge well-ordering, see Louveau [32].

4.2.4 DEFINITION: If u is a description of type 1, then we associate a description ū with u, as follows:
(a) If u = ξ¹1ⁿ0, say n = n₀+1, then ū = 0 if n₀ = 0, and ū = ξ¹1ⁿ0⁰ otherwise.
(b) If u = ξ³n^(u0,0), say n = n₀+1, then ū = u₀ if n₀ = 0, and ū = ξ²n₀^{u0} otherwise.
(c) If u = ξ³n^(u0,u1), and t(u₁) = 1, then ū = ξ³n^(u0,u1).
(d) If u = ξ⁵n^(u0,u1), then ū = ξ⁵n^(u0,u1).

Parts (c) and (d) are in fact induction steps. Note that in all cases, either $\overline{u} = \underline{0}$ or $\overline{u}(0) \ge \xi$, and $\Gamma_{\overline{u}} \subset \Gamma_{u}$, so that in (c) and (d), we have $\Gamma_{\overline{u}_{1}} < \Gamma_{u_{0}}$, and $\overline{u}_{1} = \underline{0}$ or $\overline{u}_{1}(0) \ge \xi$; hence, \overline{u} is well-defined in (c) and (d).

The main result in the first part of Louveau's paper is, that the Borel Wadge classes in ω^{ω} are exactly

 $\{\Gamma_{\mathbf{u}}^{\dagger}: \mathbf{u} \in D\} \cup \{\Gamma_{\mathbf{u}}^{\dagger}: \mathbf{u} \in D\} \cup \{\Delta(\Gamma_{\mathbf{u}}^{\dagger}): \mathbf{u} \in D, \mathbf{u}(0) = 1, \text{ and } \mathbf{t}(\mathbf{u}) \in \{1, 2\}\},\$

where we let Γ_{u}^{\dagger} denote the classes Γ_{u} defined in ω^{ω} instead of 2^{ω} . Using the fact that the Wadge-ordering well-orders the Wadge classes, it is not hard to deduce the above from the following statements:

(A) If u(0) = 1 and t(u) = 1, then $\Delta(\Gamma_u^{\dagger})$ is the unique Borel Wadge class Γ in ω^{ω} such that $\Gamma_{\overline{u}}^{\dagger} < \Gamma < \Gamma_{u}^{\dagger}$.

(B) If u(0) = 1 and t(u) = 2, then there exists a strictly increasing sequence $\langle \Gamma_{u_n}^{\dagger}: n < \omega \rangle$ of described classes such that $\Delta(\Gamma_{u}^{\dagger})$ is the unique Borel Wadge class Γ in ω^{ω} such that $\Gamma_{u_n}^{\dagger} < \Gamma < \Gamma_{u}^{\dagger}$ for all $n < \omega$. (C) If u(0) > 1 or t(u) = 3, then there exists a strictly increasing sequence $\langle \Gamma_{u_{\alpha}}^{\dagger}: \alpha < \omega_1 \rangle$ of described classes such that $\Delta(\Gamma_{u}^{\dagger}) = U\{\Gamma_{u_{\alpha}}^{\dagger}: \alpha < \omega_1\}$.

The statement (C) is proved in Louveau's lemmas 1.14, 1.19, 1.24, 1.25, and 1.28; analyzing the proofs of these lemmas, it can be seen that they go through for the Cantor set, and thus we find that statement (C) holds for the Cantor set as well.

Statements (A) and (B) are deduced by Louveau from his lemmas 1.11(b) and 1.23(b), respectively (note: the last one, as well as 1.23(a), is stated incorrectly: for $U{\Gamma_{u'}: u' \in Qu}$, read $U{\Gamma_{s_u(n)}: n < \omega}$; he appeals to "the game theoretical characterization of the Wadge ordering", but does not give any details. Since it is for (B) that the situation in the Cantor set is going to be different, we will give the correct statement of the lemma 1.23(b) of Louveau, show how to prove (B) from it, and then indicate what has to be changed in order to describe the situation for the Cantor set; this will also give the reader an impression of how to work with the original definition of the Wadge classes. The deduction of (A) from Louveau's lemma 1.11(b) is similar, and both this lemma and this deduction go through for the Cantor set.

4.2.5 LEMMA ([32], lemma 1.23(b)): Let u be a description with u(0) = 1and t(u) = 2. Then $\Delta(\Gamma_{u}^{\dagger}) = PU(\Sigma_{1}^{0}, U_{n=0}^{\infty} \Gamma_{u_{n}}^{\dagger})$ for some strictly increasing sequence $\langle \Gamma_{u_{n}}^{\dagger}: n < \omega \rangle$ of described classes. Remark: This lemma also holds for the Cantor set (so without the [†]), with the same proof.

Proof of (B): Let $\langle \Gamma_{u_n}^{\dagger} : n < \omega \rangle$ be as in the lemma. Since this sequence is strictly increasing, we may assume that $\{\{\emptyset\}, \{\omega^{\omega}\}\} < \Gamma_{u_0}^{\dagger}$, and hence if $\Gamma_{u_n}^{\dagger}$ = $[E_n]$, then $\emptyset \neq E_n \neq \omega^{\omega}$, for each $n < \omega$. Fix $n < \omega$, and put $B_n = \{n^{x}: x \in E_n\}$. Define $\psi_0: \omega^{\omega} \neq \omega^{\omega}$ by $\psi_0(x) = n^{x}$; then $E_n = \psi_0^{-1}[B_n]$, so E_n $\leq B_n$. Also, define $\psi_1: \omega^{\omega} \rightarrow \omega^{\omega}$ by $\psi_1(n^*x) = x, \psi_1(m^*x) = p \notin E_n$ if m CLAIM: $[B] = PU(\Sigma_{1}^{0}, \Gamma')$.

Indeed, if $f: \omega^{\omega} \to \omega^{\omega}$ is continuous, then $f^{-1}[B_n] \in \Gamma_{u_n}^{\dagger}$ for each $n < \omega$, so if we put $U_n = \{n^* x: x \in \omega^{\omega}\}$, then $f^{-1}[B] = \bigcup_{n=0}^{\infty} (f^{-1}[U_n])$ $\cap f^{-1}[B_n]) \in PU(\Sigma_1^0, \Gamma'), \text{ i.e. } [B] \subset PU(\Sigma_1^0, \Gamma').$

Conversely, let $A \in PU(\Sigma_1^0, \Gamma')$, say $A = \bigcup_{n=0}^{\infty} (A_n \cap C_n)$, where $\{C_n:$ $n < \omega$ partitions ω^{ω} . For each $n < \omega$, let $k(n) < \omega$ be such that $A_n \in \Gamma_{u_k(n)}^{\dagger}$, and let $f_n: \omega^{\omega} \rightarrow \omega^{\omega}$ be continuous such that $f_n^{-1}[B_k(n)]$ = A_n ; since $B_{k(n)} \neq U_n$, we may assume that $f_n(x)_0 = k(n)$ for each $x \in \omega^{\omega}$. Put $f = \bigcup_{n=0}^{\infty} (f_n | C_n)$; then $A = f^{-1}[B]$. For if $x \in A$, say $x \in A_n \cap C_n$, then $f(x) = f_n(x) \in B_k(n) \subset B$; and if $f(x) \in B$, say $f(x) \in B_{k}, \text{ then } x \in C \text{ for some } n < \omega \text{ with } k(n) = k, \text{ and thus } x \in f^{-1}[B_{k}(n)] \cap C_{n} = f_{n}^{-1}[B_{k}(n)] \cap C_{n} = A_{n} \cap C_{n} \subset A.$ Thus, we have shown that $\Delta(\Gamma_{u}^{\dagger})$ is a Wadge class, and clearly $\Gamma_{un}^{\dagger} < \Delta(\Gamma_{u}^{\dagger})$

 $< \Gamma_{u}^{\dagger}$, since Γ_{u}^{\dagger} is non-self-dual.

To complete the proof of (B), note that if Γ is a Borel Wadge class such that $\Gamma_{u_n}^+ < \Gamma < \Gamma_u^+$ for all n, then $\Gamma \subset \Delta(\Gamma_u^+)$, and also $\Delta(\Gamma_u^+) = PU(\Sigma_1^0, \Gamma)$ $c \operatorname{PU}(\Sigma_{1}^{0}, \Gamma). \text{ Now if } \Gamma = [E], \text{ and } A = \bigcup_{n=0}^{\infty} (A_{n} \cap C_{n}) \in \operatorname{PU}(\Sigma_{1}^{0}, \Gamma), \text{ where } \{C_{n}: n < \omega\} \text{ partitions } \omega^{\omega}, \text{ and each } A_{n} \in \Gamma, \text{ then taking } g_{n}: \omega^{\omega} \to \omega^{\omega} \text{ such that } A_{n} = g_{n}^{-1}[E], \text{ and defining } g: \omega^{\omega} \to \omega^{\omega} \text{ by } g = \bigcup_{n=0}^{\infty} (g_{n}|C_{n}), \text{ we see that } A = g^{-1}[E] \in \Gamma, \text{ so } \Delta(\Gamma_{u}^{\dagger}) \subset \Gamma; \text{ Thus, } \Delta(\Gamma_{u}^{\dagger}) = \Gamma.$

It is immediately clear that the above proof cannot be copied for the Cantor set; indeed, being compact, 2 $^\omega$ cannot be partitioned into infinitely many non-empty open sets.

So let $\langle \Gamma_{u_n}: n < \omega \rangle$ be the strictly increasing sequence of described classes obtained from the analogue to lemma 4.2.5 for 2^{ω} . Then each $A \in \Delta(\Gamma_{u})$ = $PU(\Sigma_1^0, U_{n=0}^{\infty} \Gamma_{u_n})$ can be written as $U_{n < k}(A_n \cap C_n)$ for some $k < \omega$, and thus $A \in PU(\Sigma_1^0, \Gamma_{u_m})$ for some $m < \omega$. So we find that

$$\Delta(\Gamma_{\mathbf{u}}) = \operatorname{PU}(\Sigma_{1}^{0}, \bigcup_{n=0}^{\infty} \Gamma_{\mathbf{u}_{n}}) = \bigcup_{n=0}^{\infty} \operatorname{PU}(\Sigma_{1}^{0}, \Gamma_{\mathbf{u}_{n}}) = \bigcup_{n=0}^{\infty} \Gamma_{\mathbf{u}_{n}},$$

the last equality by an argument as in the proof of (B). Clearly, $\bigcup_{n=0}^{\infty} \Gamma_{u_n}$ is not a Wadge class, for if $[A] = \bigcup_{n=0}^{\infty} \Gamma_{u_n}$, then $A \in \Gamma_{u_n}$ for some n, and hence $[A] \subset \Gamma_{u_n} \subsetneqq \Gamma_{u_{n+1}} \subset [A]$.

So for the Cantor set, we obtain the following theorem as the equivalent of statements (A), (B), and (C).

4.2.6 THEOREM: (a) If u(0) = 1 and t(u) = 1, then $\Delta(\Gamma_u)$ is the unique Borel Wadge class Γ such that $\Gamma_{\overline{u}} < \Gamma < \Gamma_u$. (b) If u(0) = 1 and t(u) = 2, then there exists a strictly increasing sequence $\langle \Gamma_{u_n}: n < \omega \rangle$ of described classes such that $\Delta(\Gamma_u) = \bigcup_{n=0}^{\infty} \Gamma_{u_n}$. (c) If u(0) > 1 or t(u) = 3, then there exists a strictly increasing sequence $\langle \Gamma_{u_{\alpha}}: \alpha < \omega_1 \rangle$ of described classes such that $\Delta(\Gamma_u) = U_{n=0}^{\infty} \Gamma_{u_{\alpha}}$.

From this, we easily obtain:

4.2.7 THEOREM: The collection of Borel Wadge classes in 2^{ω} is $\{\Gamma_{\mathbf{u}}: \mathbf{u} \in D\} \cup \{\Upsilon_{\mathbf{u}}: \mathbf{u} \in D\} \cup \{\Delta(\Gamma_{\mathbf{u}}): \mathbf{u} \in D, \mathbf{u}(0) = 1, and \mathbf{t}(\mathbf{u}) = 1\}.$

Thus, if we compare the pattern of the Borel Wadge classes in 2^{ω} with that of ω^{ω} as described in the first remark following theorem 4.1.9, then we see that the only difference is that in the Borel Wadge ordering on 2^{ω} , the limit stages of cofinality ω are occupied by non-self-dual pairs. Indeed, statement (A) and theorem 4.2.6(a) are similar, and so are statement (C) and theorem 4.2.6(c); since they cover the successor case and the cofinality ω_1 case, respectively no difference occurs at those levels. Theorem 4.2.6(b) says that the would-be self-dual class $\Delta(\Gamma_u)$ at a limit stage of cofinality ω is not a Wadge class, and hence its place is occupied by the non-selfdual pair $\{\Gamma_u, \check{\Gamma}_u\}$.

Remark: Since removing the ordinals of cofinality ω from a limit ordinal does not change the order-type, the order-type of the Borel Wadge oredering on 2^{ω} is still $\varepsilon_1^{\omega_1}$ (see the second remark following theorem 4.1.9).

We now derive some properties of the classes defined in definition 4.2.2. We start with a theorem concerning the classes $D_{\alpha}(\Sigma_{\xi}^{0})$. It is used extensively in Louveau's paper, but no proof seems to be available in the literature. Since we need this result for lemma 4.2.15, we include a proof here; it was suggested to us by A. Louveau.

86

Recall the definition of the reduction property from 1.11.

4.2.8 THEOREM: For each $\alpha, \xi \in [1, \omega_1)$, the reduction property holds for $D_{\alpha}(\Sigma_{F}^{0})$.

Before proving the theorem, we state two lemmas:

4.2.9 LEMMA: If $A_0 \subset A_1$, $B_0 \subset B_1$ are Σ_{ξ}^0 -sets in 2^{ω} , then there exist Σ_{ξ}^0 -sets A'_1, B'_1 such that $A_0 \subset A'_1 \subset A_1$, $B_0 \subset B'_1 \subset B_1$, and $A'_1 \setminus A_0, B'_1 \setminus B_0$ reduce $A_1 \setminus A_0, B_1 \setminus B_0$. Proof: Let $\widetilde{A}_1, \widetilde{B}_1$ reduce A_1, B_1 (see 1.11), and put $A'_1 = A_0 \cup (A_1 \cap B_0) \cup \widetilde{A}_1$, and $B'_1 = B_0 \cup (B_1 \cap A_0) \cup \widetilde{B}_1$.

4.2.10 LEMMA: Let $A_0 \subset A_1$ be Σ_{ξ}^0 -sets in 2^{ω} , and let $B \in D_n(\Sigma_{\xi}^0)$ for some $n \in N$. Then $(A_1 \setminus A_0) \setminus B = D_m(\langle C_k : k < m \rangle)$ for some even $m \in N$, and some increasing sequence $\langle C_k : k < m \rangle$ of Σ_{ξ}^0 -sets in 2^{ω} such that $C_0 = A_0$ and $C_{k-1} = A_1$.

Proof: Clearly, it suffices to prove the lemma for n = 2. So let $B_0 \subset B_1$ be Σ_{ξ}^0 -sets in 2^{ω} ; then $(A_1 \setminus A_0) \setminus (B_1 \setminus B_0) = (A_1 \cap B_0) \setminus A_0 \cup A_1 \setminus (A_0 \cup B_1) = (A_0 \cup (A_1 \cap B_0)) \setminus A_0 \cup A_1 \setminus (A_1 \cap (A_0 \cup B_1))$. So take $C_0 = A_0$, $C_1 = A_0 \cup (A_1 \cap B_0)$, $C_2 = A_1 \cap (A_0 \cup B_1)$, and $C_3 = A_1$; then $(A_1 \setminus A_0) \setminus (B_1 \setminus B_0) = D_4 ((C_k: k < 4))$.

Proof of theorem 4.2.8: Let $A, B \in D_{\alpha}(\Sigma_{\xi}^{0})$, say $A = D_{\alpha}(\langle A_{\zeta}: \zeta < \alpha \rangle)$, $B = D_{\alpha}(\langle B_{\zeta}: \zeta < \alpha \rangle)$ for certain increasing sequences $\langle A_{\zeta}: \zeta < \alpha \rangle$, $\langle B_{\zeta}: \zeta < \alpha \rangle$ of Σ_{ξ}^{0} -sets in 2^{ω} .

<u>Case 1</u>: $\alpha = n \in N$ is even. This is easily proved by successive applications of lemma 4.2.9.

 $\begin{array}{l} \underline{\text{Case 2}}: \ \alpha = n \in N \quad \text{is odd. By case 1, we can reduce } \mathbb{D}_{n-1}(\langle \mathbb{A}_{k+1}: k < n-1 \rangle), \\ \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle) \quad \text{by } \widetilde{\mathbb{A}} = \mathbb{D}_{n-1}(\langle \widetilde{\mathbb{A}}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \widetilde{\mathbb{B}}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \widetilde{\mathbb{B}}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \widetilde{\mathbb{B}}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \widetilde{\mathbb{B}}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D}_{n-1}(\langle \mathbb{B}_{k+1}: k < n-1 \rangle), \\ \widetilde{\mathbb{B}} = \mathbb{D$

$$A' = D_n(\langle A'_m : m < n \rangle) = A'_0 \cup A \setminus (A'_0 \cup B'_0),$$

$$B' = D_n(\langle B'_m : m < n \rangle) = B'_0 \cup B \setminus (A'_0 \cup B'_0)$$

reduce A,B.

<u>Case 3</u>: $\alpha = \beta + n$, where $\beta < \omega_1$ is a limit, and $n < \omega$. We will prove the case where α is odd, the other case is similar.

For even $\zeta < \beta$, put $C_{\zeta} = A_{\zeta} \setminus (U_{\gamma < \zeta} A_{\gamma})$, $D_{\zeta} = B_{\zeta} \setminus (U_{\gamma < \zeta} B_{\gamma})$, and let $\psi: \beta \rightarrow \omega$ be a bijection. Also, put

$$\begin{split} \widetilde{A} &= \cup \{ A_{\beta+m} \setminus (\cup_{\gamma < \beta+m} A_{\gamma}) : m \text{ even } < n \} = A \setminus (\cup_{\gamma < \beta} A_{\gamma}), \\ \widetilde{B} &\cup \{ B_{\beta+m} \setminus (\cup_{\gamma < \beta+m} B_{\gamma}) : m \text{ even } < n \} = B \setminus \bigcup_{\gamma < \beta} B_{\gamma} \}. \end{split}$$

For each even $\zeta < \beta$, define

$$C'_{\zeta} = C_{\zeta} \setminus (\widetilde{B} \cup U\{D_{\gamma}: \gamma \text{ even } < \beta, \psi(\gamma) < \psi(\zeta)\}),$$

$$D'_{\zeta} = D_{\zeta} \setminus (\widetilde{A} \cup U\{C_{\gamma}: \gamma \text{ even } < \beta, \psi(\gamma) \le \psi(\zeta)\}).$$

By lemma 4.2.10, there exist even $k(\zeta), \ell(\zeta) \in N$, and increasing sequences $\langle E_m^{\zeta}: m < k(\zeta) \rangle, \langle F_m^{\zeta}: m < \ell(\zeta) \rangle$ of Σ_{ξ}^0 -sets in 2^{ω} , such that

where furthermore $E_0^{\zeta} = U_{\gamma < \zeta} A_{\gamma}$, $F_0^{\zeta} = U_{\gamma < \zeta} B_{\gamma}$, and $E_k^{\zeta}(\zeta) - 1 = A_{\zeta}$, $F_{\ell}^{\zeta}(\zeta) - 1 = B_{\gamma}$.

Fix $\gamma < \beta$ satisfying $[\gamma = 0 \text{ or } \lim(\gamma)]$, and $t < \omega$. Then there are unique $n < \omega$ and $i < k(\gamma+2n)$ such that

$$t = (\Sigma_{m < n} k(\gamma + 2m)) + i - 1$$

(note that $i \ge 1$ if n = 0!). Now put

$$A_{\gamma+t}' = E_{1}^{\gamma} \cdot 2R,$$
i.e.
$$A_{\gamma}' \quad A_{\gamma+1}' \cdots \quad A_{\gamma+k(\gamma)-2}' \quad A_{\gamma+k(\gamma)-1}' \cdots \quad A_{\gamma+k(\gamma)+k(\gamma+2)-2}' \cdots$$

$$\equiv \quad E_{1}^{\gamma} \quad E_{2}^{\gamma} \quad \cdots \quad E_{k(\gamma)-1}^{\gamma} \quad E_{0}^{\gamma+2} \quad \cdots \quad E_{k(\gamma+2)-1}^{\gamma+2} \quad \cdots$$

$$\equiv \quad E_{1}^{\gamma} \quad E_{2}^{\gamma} \quad \cdots \quad A_{\gamma} \quad A_{\gamma+1} \quad \cdots \quad A_{\gamma+2} \quad \cdots$$

Then $\langle A_\zeta^{\,\prime} \colon \zeta < \beta \rangle$ is an increasing sequence of $\Sigma^0_\xi\text{-sets},$ and

$$U\{C_{\zeta}^{\prime}: \zeta \text{ even } < \beta\} = U\{A_{\zeta}^{\prime} \setminus (U_{\gamma < \zeta} A_{\gamma}^{\prime}): \zeta \text{ even } < \beta\}.$$

Similarly, define an increasing sequence $\langle B_{\zeta}^{\prime}: \zeta < \beta \rangle$ such that

$$B' = U\{D'_{\zeta}: \zeta \text{ even } < \beta\} = U\{B'_{\zeta} \setminus (U_{\gamma < \zeta} B'_{\gamma}): \zeta \text{ even } < \beta\}.$$

Note that $\bigcup_{\zeta < \beta} A_{\zeta} = \bigcup_{\zeta < \beta} A_{\zeta}$, and $\bigcup_{\zeta < \beta} B_{\zeta} = \bigcup_{\zeta < \beta} B_{\zeta}$. Clearly, $A' \cap (B' \cup \widetilde{B}) = \emptyset = B' \cap (A' \cup \widetilde{A})$, and $A' \cup \widetilde{A} \cup B' \cup \widetilde{B} = A \cup B$. By lemma 4.2.9, we can reduce $\widetilde{A}, \widetilde{B}$ by $D_{n+1}(\langle E_k: k < n+1 \rangle), D_{n+1}(\langle F_k: k < n+1 \rangle)$ such that $E_0 = \bigcup_{\zeta < \beta} A_{\zeta}$, $F_0 = \bigcup_{\zeta < \beta} B_{\zeta}$. Hence, if we put

$$A'_{\beta+m} = E_{m+1}, B'_{\beta+m} = F_{m+1}$$

for all
$$m < n$$
, then $D_{\alpha}(\langle A_{\gamma}^{\dagger}: \zeta < \alpha \rangle)$, $D_{\alpha}(\langle B_{\gamma}^{\dagger}: \zeta < \alpha \rangle)$ reduce A,B.

89

The other lemmas that we derive in this section are concerned with closure properties of the classes Γ_u . We first list some results from Louveau [32] that also hold for 2^{ω} , with the same proof.

4.2.11 LEMMA: Let $u \in D$, with $u(0) = \xi$. (a) ([32], lemma 1.4 and its proof) $SU(\Sigma_{\xi}^{0},\Gamma_{u}) = \Gamma_{u}$, and if $\eta < \xi$, then $SU(\Sigma_n^0, \Gamma_u) = \Gamma_u$. (b) ([32], 1emma 1.4) Γ_{u} and Γ_{u} are closed under union and intersection with a $\Delta^0_{\mathcal{E}}$ -set. (c) ([32], proof of lemma 1.4) If u(1) = 4, then Γ_u is closed under union with a Σ^0_{ξ} -set. (d) ([32], lemma 1.26, and proof of lemma 1.4) If t(u) = 3, then Γ_{u} and set. (e) ([32], corollary 1.6) If $u = \xi^3 \eta^{u_0, u_1}$, and $A \in \Gamma_u$, then there exist $C \in \Sigma^0_{\mathcal{E}}$ and $B \in \Gamma_{u_1}$ such that $A = (A \cap C) \cup (B \setminus C)$, and both $A \cap C$ and $C\setminus A$ are in $Bisep(D_n(\Sigma^0_{\xi}),\Gamma_{u_0})$. (f) ([32], lemma 1.11) If t(u) = 1, then $\Gamma_{u} = \text{Bisep}(\Sigma_{\xi}^{0}, \Gamma_{\overline{u}})$, with \overline{u} as in definition 4.2.4. (g) ([32], lemma 1.23) If t(u) = 2, then $\Gamma_u = SU(\Sigma_{\xi}^0, \bigcup_{n=0}^{\infty} \Gamma_{u_n})$ for some strictly increasing sequence $\langle \Gamma_{u_n}: n < \omega \rangle$ of described classes with $u_n(0)$ $\geq \xi$ for all $n < \omega$.

The following lemmas are of a similar nature. We note here that, since the homogeneous zero-dimensional absolute Borel sets of ambiguous class 2 were characterized in the preceding chapters, it would suffice now to consider Wadge classes Γ_u with $\Delta_3^0 \subset \Gamma_u$ in order to obtain the complete picture. However, for the discussion in section 4.6, it will be convenient to have the results of this section for classes Γ_u containing $\Delta(D_{\omega}(\Sigma_2^0))$.

4.2.12 LEMMA: If $\widetilde{\Delta} = \Delta(D_{\omega}(\Sigma_2^0)) \subset \Gamma_u$, and $u(0) \ge 2$, then Γ_u is closed under intersection with a Π_2^0 -set and under union with a Σ_2^0 -set; hence so is Υ_u .

Proof: First note that, by lemma 3.5.1, $D_{\omega}(\Sigma_2^0)$ is closed under intersection with a Π_2^0 -set; and if $A = D_{\omega}(\langle A_{\zeta}: \zeta < \omega \rangle) \in D_{\omega}(\Sigma_2^0)$, and F is σ -compact,

then $A \cup F = D_{\omega}(\langle A_{\zeta}^{\dagger} : \zeta < \omega \rangle)$, where $A_{0}^{\dagger} = A_{0}$, and $A_{\zeta}^{\dagger} = A_{\zeta} \cup F$ if $0 < \zeta$ < ω . So the lemma holds if Γ_u or Γ_u is $D_{\omega}(\Sigma_2^0)$. If the lemma is false, then there is a minimal class Γ_{μ} for which it fails. By lemma 4.2.11(d), the lemma holds if t(u) = 3, and by lemma 4.2.11(b) also if $u(0) \ge 3$; so we have u(0) = 2 and $t(u) \in \{1, 2\}$. <u>Case 1</u>: t(u) = 1. By lemma 4.2.11(f), $\Gamma_{u} = \text{Bisep}(\Sigma_{2}^{0}, \Gamma_{u})$. Suppose that $\widetilde{\Delta} \notin$ $\Gamma_{\overline{u}}; \text{ then } \Gamma_{\overline{u}} \subset \widetilde{\Delta}, \text{ so } \widetilde{\Delta} \subset \Gamma_{u} = \text{Bisep}(\Sigma_{2}^{0}, \Gamma_{\overline{u}}) \subset \text{Bisep}(\Sigma_{2}^{0}, \widetilde{\Delta}) \subset \text{SU}(\Sigma_{2}^{0}, \mathbb{D}_{u}(\Sigma_{2}^{0})) =$ $D_{\omega}(\Sigma_2^0)$ by lemma 4.2.11(a), and hence, since Γ_u is non-self-dual, $D_{\omega}(\Sigma_2^0) \in$ $\{\overline{\Gamma}_{u}, \overline{\Gamma}_{u}\}$, contradicting the above remark. Thus, $\Gamma_{\overline{u}} \subset \widetilde{\Delta}$, and from definition 4.2.4 it is easily seen that $\overline{u}(0) \ge 2$, since $\overline{u} = 0$ would imply $\Gamma_{u} =$ $D_1(\Sigma_2^0) \neq \widetilde{\Delta}$. Since $\Gamma_u = \text{Bisep}(\Sigma_2^0, \Gamma_{\overline{u}}) \supset \Gamma_{\overline{u}} \cup \Upsilon_{\overline{u}}$, we have $\Gamma_{\overline{u}} < \Gamma_u$, so by minimality of $\Gamma_{\rm u}$, $\Gamma_{\rm u}$ and $\Gamma_{\rm u}$ have the desribed closure properties. Now let A $\epsilon \Gamma_{u}$, say $A = (A_0 \cap C_0) \cup (A_1 \cap C_1)$, where C_0, C_1 are disjoint Σ_2^0 -sets, $A_0 \in \check{\Gamma}_{\overline{u}}$, and $A_1 \in \Gamma_{\overline{u}}$. If $F \in \Pi_2^0$, then $A_0 \cap F \in \check{\Gamma}_{\overline{u}}$, $A_1 \cap F \in \Gamma_{\overline{u}}$, so $A \cap F$ = $(A_0 \cap F \cap C_0) \cup (A_1 \cap F \cap C_1) \in \Gamma_u$. If $G \in \Sigma_2^0$, let C_0^*, C_1^* reduce $C_0 \cup G$, $C_1 \cup G$. Then $A \cup G = ((A_0 \cup G) \cap C_0^*) \cup ((A_1 \cup G) \cap C_1^*) \in \Gamma_1$ since $A_0 \cup G$ $\epsilon \stackrel{\bullet}{\Gamma}_{\overline{u}}, A_1 \cup G \epsilon \Gamma_{\overline{u}}.$ <u>Case 2</u>: t(u) = 2. By lemma 4.2.11(g), $\Gamma_u = SU(\Sigma_2^0, \bigcup_{n=0}^{\infty} \Gamma_{u_n})$ for some strictly increasing sequence $\langle \Gamma_{u_n}: n < \omega \rangle$ of described classes with $u_n(0) \ge 2$

If increasing sequence $\langle \Gamma_{u_n}: n < \omega \rangle$ of described classes with $u_n(0) \ge 2$ for all $n < \omega$. If each $\Gamma_{u_n} \subset \widetilde{\Delta}$, then as in case 1 we obtain a contradiction; thus we conclude that $\widetilde{\Delta} \subset \Gamma_{u_k}$ for some k, and hence $\widetilde{\Delta} \subset \Gamma_{u_m}$ for all $m \ge k$. Since $u_m(0) \ge 2$ and $\Gamma_{u_m} < \Gamma_u$, each Γ_{u_m} with $m \ge k$ has the described closure properties. Now let $A \in \Gamma_u$, say $A = \bigcup_{n=0}^{\infty} (A_n \cap C_n)$, where the C_n are pairwise disjoint Σ_2^0 -sets, and $A_n \in \Gamma_{u_m(n)}$ for some $m(n) < \omega$; of course, we may assume that $m(n) \ge k$. As in case 1, if $F \in \Pi_2^0$, then $A_n \cap F \in \Gamma_{u_m(n)}$, so $A \cap F = \bigcup_{n=0}^{\infty} (A_n \cap F \cap C_n) \in \Gamma_u$; and if $G \in \Sigma_2^0$, and $\langle C_n^*: n < \omega \rangle$ reduces $\langle C_n \cup G: n < \omega \rangle$, then $A \cup G = \bigcup_{n=0}^{\infty} ((A_n \cup G) \cap C_n^*) \in \Gamma_u$.

4.2.13 LEMMA: If $u(0) \ge 3$, or u(0) = 2 and t(u) = 3, then Γ_u is closed under union with a Π_2^0 -set.

Proof: If the lemma fails, then there is a minimal Γ_u for which it does. Since the lemma is true if $u(0) \ge 3$ by lemma 4.2.11(b), we have u(0) = 2and t(u) = 3. Let $F \in \Pi_2^0$.

90

$$\begin{split} &\Gamma_{u_1} < \Gamma_{u_0}, \ u_0(0) > 2, \ \text{and} \ u_1(0) > 2 \quad \text{or} \quad (u_1(0) = 2 \quad \text{and} \quad t(u_1) = 3). \text{ If } A \\ & \in \ \Gamma_{u}, \ \text{say} \quad A = (A_0 \cap C_0) \cup (A_1 \cap C_1) \cup B \setminus (C_0 \cup C_1), \ \text{then} \quad A_0 \cup F \in \ T_{u_0}, \ A_1 \cup F \in \ \Gamma_{u_0} \quad \text{by lemma 4.2.11(b) ; also, } B \cup F \in \ \Gamma_{u_1} \quad \text{by lemma 4.2.11(b) if } u_1(0) \\ > 2, \ \text{and by minimality of} \quad \Gamma_u \quad \text{if } u_1(0) = 2 \ (\text{note that} \quad \Gamma_{u_1} < \ \Gamma_{u_0} \subset \ \Gamma_u). \\ \text{So } \quad A \cup F = ((A_0 \cup F) \cap C_0) \cup ((A_1 \cup F) \cap C_1) \cup (B \cup F) \setminus (C_0 \cup C_1) \in \ \Gamma_u. \\ \hline \underline{Case \ 3}: \ u(1) = 5, \ \text{so} \quad \Gamma_u = \ \text{SD}_\eta (\langle \Sigma_2^0, \Gamma_{u_0} \rangle, \Gamma_{u_1}) \quad \text{for some} \quad u_0, u_1 \in D \quad \text{with} \quad \Gamma_{u_1} \\ < \ \Gamma_{u_0}, \ u_0(0) = 2, \ u_0(1) = 4, \ \text{and} \quad u_1(0) > 2 \quad \text{or} \quad (u_1(0) = 2 \quad \text{and} \quad t(u_1) = 3). \\ \text{Let} \quad A \in \ \Gamma_u, \ \text{say} \quad A = \ U_{\zeta < \eta} (A_{\zeta} \setminus (U_{\beta < \zeta} \ C_{\beta})) \cup B \setminus (U_{\zeta < \eta} \ C_{\zeta}). \ \text{Since} \quad 2^\omega \setminus F \in \ \Sigma_2^0, \ \text{and} \\ u_0(0) = 2, \ \text{by lemma 4.2.11(a) we have} \quad A_{\zeta} \cap (2^\omega \setminus F) \in \ \Gamma_{u_0}, \ \text{and it is easily} \\ \text{verified that we can choose} \quad C_{\zeta} \cap (2^\omega \setminus F) \quad \text{as envelop of} \quad A_{\zeta} \cap (2^\omega \setminus F); \ \text{also} \\ B \cup F \in \ \Gamma_{u_1} \quad \text{as in case 2. So} \end{split}$$

$$A \cup F = U_{\zeta < \eta}((A_{\zeta} \cap (2^{\omega} \setminus F)) \setminus U_{\beta < \zeta}(C_{\beta} \cap (2^{\omega} \setminus F))) \cup (B \cup F) \setminus U_{\zeta < \eta}(C_{\zeta} \cap (2^{\omega} \setminus F))$$

$$\in \Gamma_{u}.$$

4.2.14 COROLLARY: If $u(0) \ge 3$, or u(0) = 2 and t(u) = 3, then $SU(\Sigma_2^0, \Gamma_u) = \Gamma_u$.

Proof: If $A \in SU(\Sigma_2^0, \Upsilon_u)$, say $A = \bigcup_{n=0}^{\infty} (A_n \cap C_n)$, then $2^{\omega} \setminus A = \bigcup_{n=0}^{\infty} (C_n \cap (2^{\omega} \setminus A_n)) \cup 2^{\omega} \setminus (\bigcup_{n=0}^{\infty} C_n)$. Now $2^{\omega} \setminus A_n \in \Gamma_u$, so $\bigcup_{n=0}^{\infty} (C_n \cap (2^{\omega} \setminus A_n)) \in SU(\Sigma_2^0, \Gamma_u)$ = Γ_u by lemma 4.2.11(a), and $2^{\omega} \setminus (\bigcup_{n=0}^{\infty} C_n) \in \Pi_2^0$; thus, $2^{\omega} \setminus A \in \Gamma_u$ by lemma 4.2.13, whence $A \in \Upsilon_u$.

For the final lemmas of this section, it will be convenient to introduce the following notation: if $X \, \subset \, 2^{\omega}$, then $\Gamma_{u}(X)$ is the class of all subsets of X that are formed by applying the corresponding operations to the corresponding subsets of X. For example, $\Sigma_{\xi}^{0}(X)$ consists of all Σ_{ξ}^{0} -sets in X, $D_{\eta}(\Sigma_{\xi}^{0})(X)$ is the class $D_{\eta}^{X}(\Sigma_{\xi}^{0})$ of definition 3.1.1(b), $\operatorname{Sep}(D_{\eta}(\Sigma_{\xi}^{0}),\Gamma_{u*})(X) = \operatorname{Sep}(D_{\eta}(\Sigma_{\xi}^{0})(X),\Gamma_{u*}(X))$, etc..

4.2.15 LEMMA: If $X \in 2^{\omega}$, and $A \in X$, then $A \in \Gamma_{u}(X)$ if and only if $A = \widetilde{A} \cap X$ for some $\widetilde{A} \in \Gamma_{u}$; similarly for Υ_{u} .

Proof: The "if" part is clear (use induction), and it is also elementary to show that if the lemma holds for Γ_u , then it also holds for $\stackrel{Y}{\Gamma}_u$. The "only if" is trivial for $\Gamma_u = D_1(\Sigma_{\xi}^0) = \Sigma_{\xi}^0(2^{\omega})$; and if $A = D_{\eta}(\langle A_{\zeta}: \zeta < \eta \rangle)$ for some increasing sequence $\langle A_{\zeta}: \zeta < \eta \rangle$ of Σ_{ξ}^0 -sets in X, say $A_{\zeta} = B_{\zeta} \cap X$, where B_{ζ} is Σ_{ξ}^0 in 2^{ω} , then $A = D_{\eta}(\langle B_{\zeta}: \zeta < \eta \rangle) \cap X$, where $B_{\zeta}' = U_{\gamma \leq \zeta} B_{\gamma}$. If the lemma holds for Γ_{u*} , and $A \in \Gamma_u(X)$, where $\Gamma_u = \operatorname{Sep}(D_{\eta}(\Sigma_{\xi}^0), \Gamma_{u*})$,

say $A = (A_0 \cap C) \cup (A_1 \setminus C)$, then there exist $\widetilde{C} \in D_{\eta}(\Sigma_{\xi}^0)$, $\widetilde{A}_0 \in \Gamma_{u*}$, and \widetilde{A}_1 $\epsilon \Gamma_{u*}$ such that $\widetilde{C} \cap X = C$, $\widetilde{A}_0 \cap X = A_0$, and $\widetilde{A}_1 \cap X = A_1$; hence, $\widetilde{A} = (\widetilde{A}_0 \cap X)$ \widetilde{C}) \cup $(\widetilde{A}_1 \setminus \widetilde{C}) \in \Gamma_u$, and $\widetilde{A} \cap X = A$. If the lemma holds for Γ_{u_0} and Γ_{u_1} , and $A \in \Gamma_u(X)$, where $\Gamma_u = \text{Bisep}(D_n(\Sigma_{\xi}^0))$, $\Gamma_{u_0}, \Gamma_{u_1}$, say $A = (A_0 \cap C_0) \cup (A_1 \cap C_1) \cup B \setminus (C_0 \cup C_1)$, then there exist \widetilde{C}_0 , $\widetilde{C}_1 \in \widetilde{D}_{\eta}(\Sigma_{\xi}^0), \widetilde{A}_0 \in \widetilde{T}_{u_0}, \widetilde{A}_1 \in \Gamma_{u_0}, \text{ and } \widetilde{\widetilde{B}} \in \Gamma_{u_1} \text{ such that } \widetilde{C}_0 \cap X = C_0, \widetilde{C}_1 \cap \widetilde{C}_1 \cap \widetilde{C}_1$ $X = C_1$, $\widetilde{A}_0 \cap X = A_0$, $\widetilde{A}_1 \cap X = A_1$, and $\widetilde{B} \cap X = B$. By theorem 4.2.8, there are $C'_{0}, C'_{1} \in D_{\eta}(\Sigma^{0}_{2,\xi})$ reducing $\widetilde{C}_{0}, \widetilde{C}_{1}$. Then $\widetilde{A} = (\widetilde{A}_{0} \cap C'_{0}) \cup (\widetilde{A}_{1} \cap C'_{1}) \cup \widetilde{B} \setminus (C'_{0} \cup C'_{1})$ $\epsilon \Gamma_{u}$, and $\widetilde{A} \cap X = A$. The case $\Gamma_{u} = SU(\Sigma_{\xi}^{0}, \bigcup_{n=0}^{\infty} \Gamma_{u_{n}})$ is proved similarly, using the reduction property for infinite sequences of Σ_{ξ}^{0} -sets (see 1.11). Finally, let $A \in \Gamma_{u}(X)$, where $\Gamma_{u} = SD_{\eta}((\Sigma_{\xi}^{0}, \Gamma_{u_{0}}), \Gamma_{u_{1}})$, say $A = \bigcup_{\zeta < \eta}(A_{\zeta} \setminus (\bigcup_{\beta < \zeta} C_{\beta})) \cup B \setminus (\bigcup_{\zeta < \eta} C_{\zeta})$. Let $\widetilde{A}_{\zeta} \in \Gamma_{u_{0}}$, $\widetilde{B} \in \Gamma_{u_{1}}$ be such that $\widetilde{A}_{\zeta} \cap X = A_{\zeta}$, $\widetilde{B} \cap X = B$, and let \widetilde{C}_{ζ} be the envelop of \widetilde{A}_{ζ} . Put $A_{\zeta}' = \widetilde{A}_{\zeta} \cup \bigcup_{\beta < \zeta} \widetilde{C}_{\beta}$, and $C_{\zeta} = \bigcup_{\beta \leq \zeta} \widetilde{C}_{\beta}$. By lemma 4.2.11(c), we have $A_{\zeta} \in \Gamma_{u_0}$, and in fact, arguing as in the proof of lemma 4.2.12, case 2, it can be shown that we can choose C as its envelop. Thus, $\widetilde{A} = \bigcup_{\zeta < \eta} (A_{\zeta}^{\prime} \setminus (\bigcup_{\beta < \zeta} C_{\beta}^{\prime})) \cup \widetilde{B} \setminus (\bigcup_{\zeta < \eta} C_{\zeta}^{\prime}) \in \Gamma_{u}$, and $\widetilde{A} \cap X^{\prime}$ = A.

4.2.16 LEMMA: If $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 2$, and if $B \subset 2^{\omega}$, $A \in \Gamma_{u}$, $B \approx A$, then $B \in \Gamma_{u}$; similarly for Γ_{u} , and hence for $\Delta(\Gamma_{u})$.

Proof: Let f: A \approx B. By Lavrentieff's theorem, there exist Π_2^0 -sets G,H in 2^{ω} with A \subset G, B \subset H, and a homeomorphism $\tilde{f}: G \rightarrow H$ extending f. Since A $\epsilon \Gamma_u$, also A $\epsilon \Gamma_u(G)$ by lemma 4.2.15, and hence B $\epsilon \Gamma_u(H)$. Again by lemma 4.2.15, there exists $\tilde{B} \epsilon \Gamma_u$ with $\tilde{B} \cap H = B$. Since $H \epsilon \Pi_2^0$, by lemma 4.2.12 we have $\tilde{B} \cap H \epsilon \Gamma_u$, i.e. B $\epsilon \Gamma_u$. The proof for Γ_u is analogous.

4.2.17 LEMMA: If $\Delta(D_{\omega}(\Sigma_{1}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 2$, then Γ_{u} is reasonably closed; similarly for Γ_{u} .

Proof: Being a Wadge class, Γ_u is continuously closed. Let $\phi: 2^{\omega} \setminus (Q_0 \cup Q_1) \rightarrow 2^{\omega}$ be as in definition 4.1.10, and put $P = 2^{\omega} \setminus (Q_0 \cup Q_1)$. If $A \in \Gamma_u$, then clearly $\phi^{-1}[A] \in \Gamma_u(P)$. Hence for some $\widetilde{A} \in \Gamma_u$, we have $\phi^{-1}[A] = \widetilde{A} \cap P$. Since $P \in \Pi_2^0$, $\phi^{-1}[A] \in \Gamma_u$ by lemma 4.2.12, and hence $\phi^{-1}[A] \cup Q_0 \in \Gamma_u$ by lemma 4.2.12 since $Q_0 \in \Sigma_2^0$. The proof for Γ_u is the same.

4.3 Existence of homogeneous Borel sets

Since we want to describe topological types rather than subsets of 2^{ω} , we use the following topological properties instead of membership of Γ_{u} or Υ_{u} :

4.3.1 DEFINITION: Let $u \in D$, and let X be a zero-dimensional space. (a) X has property P_u if and only if each copy of X in 2^{ω} is in Γ_u . (b) X has property P_u if and only if each copy of X in 2^{ω} is in Υ_u .

Remark: By lemma 4.2.16, if $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 2$, then X is P_{u} (resp. P_{u}) if and only if *some* copy of X in 2^{ω} is in Γ_{u} (resp. Y_{u}).

4.3.2 DEFINITION: Let $u \in D$, and let X be a zero-dimensional space. (a) $X \in Y_{u}^{0}$ if and only if X is P_{u} , nowhere \not{P}_{u} , and first category. (b) $X \in Y_{u}^{1}$ if and only if X is P_{u} , nowhere \not{P}_{u} , and Baire. (c) $X \in Z_{u}^{0}$ if and only if X is \not{P}_{u} , nowhere P_{u} , and first category. (d) $X \in Z_{u}^{1}$ if and only if X is \not{P}_{u} , nowhere P_{u} , and Baire.

4.3.3 LEMMA: Let $u \in D$, $\Delta(D_{\omega}(\Sigma_2^0)) \subset \Gamma_u$, $u(0) \ge 2$, and let X be a zero-dimensional space.

(a) X is P_u , nowhere $\overset{\downarrow}{P}_u$ if and only if some copy of X in 2^{ω} is everywhere properly Γ_u .

(b) X is $\overset{\circ}{P}_{u}$, nowhere P_{u} if and only if some copy of X in 2^{ω} is everywhere properly $\overset{\circ}{T}_{u}$.

Proof: We prove (a); (b) is similar. Let X be P_u and nowhere \not{P}_u , and let Y be a dense copy of X in 2^{ω} (such a copy exists, since any isolated point of X would be in f_u when "embedded" in 2^{ω}). Then Y ϵ Γ_u by definition, and hence $U \cap Y \epsilon$ Γ_u for each open U in 2^{ω} , by lemma 4.2.11(b). Now let U be any non-empty open subset of 2^{ω} , and suppose that $U \cap Y \epsilon$ f_u ; then by the remark following definition 4.3.1, X contains a non-empty open subset which is \not{P}_u , a contradiction.

Conversely, suppose that Y is a copy of X in 2^{ω} which is everywhere properly Γ_u . Then some copy of X in 2^{ω} is Γ_u , so X is P_u ; and if U is a non-empty open subset of X which is \check{P}_u , then U corresponds to V \cap Y for some non-empty open subset V of 2^{ω} , whence $V \cap Y \in \check{T}_u$, a contradiction. We will now show exactly which of the classes $y_u^0, y_u^1, Z_u^0, Z_u^1$, with $u(0) \ge 2$ and $\Delta(D_{\omega}(\Sigma_2^0)) \subset \Gamma_u$, are non-empty, and we prove that the non-empty ones contain exactly one topological type, which is homogeneous.

4.3.4 LEMMA: Let $u \in D$, $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 2$. Then $y_{u}^{0} \neq \emptyset$ if and only if $Z_{u}^{1} \neq \emptyset$, and $y_{u}^{1} \neq \emptyset$ if and only if $Z_{u}^{0} \neq \emptyset$.

Proof: Since $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, both Γ_{u} and $\stackrel{Y}{\Gamma}_{u}$ contain all compact subsets of 2^{ω} ; hence if a subset X of 2^{ω} is everywhere properly Γ_{u} or everywhere properly $\stackrel{Y}{\Gamma}_{u}$, then X is dense and co-dense in 2^{ω} . Thus, by 1.12.3, X is first category if and only if $2^{\omega}\setminus X$ is Baire. So by lemma 4.3.3, it suffices to show that $X \subset 2^{\omega}$ is everywhere properly Γ_{u} if and only if $2^{\omega}\setminus X$ is everywhere properly $\stackrel{Y}{\Gamma}_{u}$; for then, if $Y \in \mathcal{Y}_{u}^{0}$, and $\stackrel{Y}{Y}$ is a copy of Y in 2^{ω} which is everywhere properly Γ_{u} . Since $X \in \Gamma_{u}$, $2^{\omega}\setminus X \in \stackrel{Y}{\Gamma}_{u}$, and hence for each open subset U of 2^{ω} , we have $U \cap (2^{\omega}\setminus X) \in \stackrel{Y}{\Gamma}_{u}$ by lemma 4.2.11(b), since $u(0) \ge 2$. Now let U be any non-empty open subset of 2^{ω} , and suppose that $V = U \cap (2^{\omega}\setminus X) \in \Gamma_{u}$. Then $2^{\omega}\setminus V \in \stackrel{Y}{\Gamma}_{u}$, and again by lemma 4.2.11(b), also $U \cap (2^{\omega}\setminus V) = U \cap X \in \stackrel{Y}{\Gamma}_{u}$, a contradiction. The converse is established in a similar way.

4.3.5 LEMMA: If $\Delta(D_{\omega}(\Sigma_2^0)) \subset \Gamma_{u}$, and $u(0) \ge 2$, then y_u^0 and Z_u^1 are non-empty.

Proof: By lemma 4.3.4, it suffices to show that y_u^0 is non-empty. Since Γ_u is non-self-dual, there is a subset Z of 2^{ω} such that $Z \in \Upsilon_u \setminus \Gamma_u$. Let

 $0 = U\{U: U \text{ is open in } Z, \text{ and } U \in \Gamma_{U}\}.$

Then for some $\bigcup_{n} \in \Gamma_{u}$ with \bigcup_{n} open in Z, we have $0 = \bigcup_{n=0}^{\infty} \bigcup_{n}$. Let \bigcup_{n}^{∞} be open in 2^{ω} with $\bigcup_{n} \cap Z = \bigcup_{n}$, and let $\langle V_{n} : n < \omega \rangle$ be open subsets of 2^{ω} reducing $\langle \widetilde{U}_{n} : n < \omega \rangle$. Then $V_{n} \cap Z = V_{n} \cap \widetilde{U}_{n} \cap Z = V_{n} \cap \bigcup_{n}$, and since $\bigcup_{n} \in \Gamma_{u}$, $u(0) \geq 2$, and $\bigvee_{n} \in \Sigma_{1}^{0}$, we have $\bigvee_{n} \cap Z \in \Gamma_{u}$ by lemma 4.2.11(b). So $0 = \bigcup_{n=0}^{\infty} (\bigvee_{n} \cap Z) \in SU(\Sigma_{1}^{0}, \Gamma_{u}) = \Gamma_{u}$ by lemma 4.2.11(a). Put $\widetilde{Z} = Z \setminus 0$; then \widetilde{Z} is non-empty since $Z \notin \Gamma_{u}$. Since $\widetilde{Z} = Z \setminus (\bigcup_{n=0}^{\infty} \bigvee_{n})$, and $2^{\omega} \setminus (\bigcup_{n=0}^{\infty} \bigvee_{n}) \in \Pi_{1}^{0} \subset \Delta_{2}^{0}$, we have $\widetilde{Z} \notin \Gamma_{u}$ by lemma 4.2.11(b). We claim that no non-empty open subset U of \widetilde{Z} is in Γ_{u} . Indeed, if $U \in \Gamma_{u}$, then choose \widetilde{U} open in 2^{ω} with $\widetilde{U} \cap \widetilde{Z} = U$; then

 $\widetilde{U} \cap Z = ((2^{\omega} \setminus (\bigcup_{n=0}^{\infty} V_n)) \cap (\widetilde{U} \cap \widetilde{Z})) \cup \bigcup_{n=0}^{\infty} ((\widetilde{U} \cap V_n) \cap (V_n \cap Z))$

 \in SU(Σ_2^0, Γ_u) = Γ_u by lemma 4.2.11(a), so $\widetilde{U} \cap Z \subset 0$, contradicting $\emptyset \neq U \subset$

(Ũ ∩ Z)\O.

Now let Z' be a densely embedded copy of \widetilde{Z} in 2^{ω} (which exists since \widetilde{Z} contains no isolated points), and put $Y = 2^{\omega} \setminus Z'$. Also, let Q be a countable dense subset of 2^{ω} , let h: $2^{\omega} \times 2^{\omega} \rightarrow 2^{\omega}$ be a homeomorphism, and put

 $Y_{u}^{0} = h[Q \times Y].$

We claim that $Y_{u}^{0} \in Y_{u}^{0}$. First note that $Z' \in F_{u}$ by lemma 4.2.16, whence $Y \in \Gamma_{u}$. Hence $Q \times Y = \bigcup_{q \in Q} (\{q\} \times Y \cap \{q\} \times 2^{\omega}) \in SU(\Sigma_{2}^{0}, \Gamma_{u}) = \Gamma_{u}$, and thus $\bigcup \cap Y_{u}^{0} \in \Gamma_{u}$ for each open subset \bigcup of 2^{ω} , by lemmas 4.2.11(a) and 4.2.16. Now let \bigcup be any non-empty open subset of 2^{ω} , and suppose that $\bigcup \cap Y_{u}^{0} \in F_{u}$. If \bigcup_{0}, \bigcup_{1} are non-empty open subsets of 2^{ω} such that $h[\bigcup_{0} \times \bigcup_{1}] \subset \bigcup$, then also $h[\bigcup_{0} \times \bigcup_{1}] \cap Y_{u}^{0} \in F_{u}$ by lemma 4.2.11(b), and hence $(\bigcup_{0} \times \bigcup_{1}) \cap (Q \times Y) = V \in F_{u}$ by lemma 4.2.16. Let $q \in \bigcup_{0} \cap Q$, then $(\{q\} \times 2^{\omega}) \cap V = \{q\} \times (Y \cap \bigcup_{1}) \in F_{u}$ by lemma 4.2.11(b), and hence $Y \cap \bigcup_{1} \in F_{u}$ by lemma 4.2.16. But now $\bigcup_{1} \cap 2^{\omega} \setminus (Y \cap \bigcup_{1}) = \bigcup_{1} \cap Z'$ is a non-empty open subset of Z' which is in Γ_{u} . Hence \widetilde{Z} also contains such a subspace, and we have a contradiction. So Y_{u}^{0} is everywhere properly Γ_{u} , and first category, so $Y_{u}^{0} \in Y_{u}^{0}$ using lemma 4.3.3.

If we try to prove that Z_u^0 and Y_u^1 are non-empty by replacing Γ_u in the above argument by Υ_u , then we see that we need that $SU(\Sigma_2^0, \Upsilon_u) = \Upsilon_u$ holds; as we shall see in lemma 4.3.7, this is not always the case. However, from corollary 4.2.14, we see that the following holds:

4.3.6 LEMMA: If $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 3$ or (u(0) = 2 and t(u) = 3), then Z_{u}^{0} and Y_{u}^{1} are non-empty.

In fact, if $Z_{u}^{1} \in Z_{u}^{1}$, then $Q \times Z_{u}^{1} \in Z_{u}^{0}$.

4.3.7 LEMMA: If $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, u(0) = 2, and $t(u) \in \{1,2\}$, then $Y_{u}^{1} = \emptyset$, and $Z_{u}^{0} = \emptyset$.

Proof: By lemma 4.3.4, it suffices to show that $\mathcal{Y}_{u}^{1} = \emptyset$. We will show that if $X \in 2^{\omega}$ is everywhere properly Γ_{u} , then X is first category. First take t(u) = 2. By lemma 4.2.11(g), $\Gamma_{u} = SU(\Sigma_{2}^{0}, \bigcup_{n=0}^{\infty} \Gamma_{u_{n}})$ for some increasing sequence $\langle \Gamma_{u_{n}}: n < \omega \rangle$ of described classes with $u_{n}(0) \ge 2$ for all $n < \omega$. Write $X = \bigcup_{n=0}^{\infty} (A_{n} \cap C_{n})$, and let $C_{n} = \bigcup_{m=0}^{\infty} C_{m}^{n}$, with $C_{m}^{n} \in \Pi_{1}^{0}$; then if $A_{n} \in$ $\Gamma_{u_{k}}$, also $C_{m}^{n} \cap X = C_{m}^{n} \cap A_{n} \in \Gamma_{u_{k}}$ since $u_{k}(0) \ge 2$ and $C_{m}^{n} \in \Delta_{2}^{0}$, using lemma 4.2.11(b). If U is non-empty and open in X, say $U = \widetilde{U} \cap X$, with \widetilde{U} open in 2^{ω} , and if $U \subset C_m^n \cap A_n$, then $U = \widetilde{U} \cap C_m^n \cap A_n \in \Gamma_{u_k}$; thus $U \in \overset{Y}{\Gamma}_u$ since $\Gamma_{u_k} < \Gamma_u$, a contradiction. So C_m^n is closed and nowhere dense in X, whence $X = \bigcup_{n=0}^{\infty} \bigcup_{m=0}^{\infty} (C_m^n \cap X)$ is first category. If t(u) = 1, then note that since $\Delta(D_{\omega}(\Sigma_2^0)) \subset \Gamma_u$, we have $\overline{u}(0) \ge 2$. Since $\Gamma_u = \text{Bisep}(\Sigma_2^0, \Gamma_u)$ by lemma 4.2.11(f), we can argue as above.

4.3.8 THEOREM: Let $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 2$. Then up to homeomorphism, each of $y_{u}^{0}, y_{u}^{1}, Z_{u}^{0}, Z_{u}^{1}$ contains at most one space, and if it exists, then this space is strongly homogeneous, whence homogeneous.

Proof: If X, Y $\in Y_u^0$, then there are copies \tilde{X}, \tilde{Y} of X, Y, respectively, in 2^{ω} such that \tilde{X}, \tilde{Y} are everywhere properly Γ_u . By lemma 4.2.17, Γ_u is reasonably closed, whence by theorem 4.1.11, there is a homeomorphism h: $2^{\omega} \rightarrow 2^{\omega}$ such that $h[\tilde{X}] = \tilde{Y}$, i.e. $\tilde{X} \approx \tilde{Y}$. For the other classes, the proof is similar. For the second part of the lemma, it now suffices to show that if $X \in Y_u^0$ (resp. Y_u^1, Z_u^0, Z_u^1), and U is a non-empty clopen subset of X, then $U \in Y_u^0$ (resp. Y_u^1, Z_u^0, Z_u^1). Again, we only give the argument for Y_u^0 . So let X be a subset of 2^{ω} which is everywhere properly Γ_u , and first category, and let U be non-empty and open in X. Embed U as a dense subset A of 2^{ω} ; this is possible since U contains no isolated points. Let V be a non-empty open subset of 2^{ω} ; then $V \cap A \approx W$ for some open subset W of U. Since U is open in X, so is W, and hence $W = \widetilde{W} \cap X$ for some non-empty open subset \widetilde{W} of 2^{ω} . Since X is everywhere properly Γ_u , we have $W \in \Gamma_u \setminus \tilde{Y}_u^0$, and thus $V \cap A \in \Gamma_u \setminus \tilde{Y}_u^0$ by lemma 4.2.16. So A is everywhere properly Γ_u , hence by theorem 4.3.3, $U \in Y_u^0$ since U is clearly first category.

Remark: From the proof of lemma 4.3.7 it follows that if $\Delta(D_{\omega}(\Sigma_2^0)) \subset \Gamma_{u}$, u(0) = 2, and t(u) ϵ {1,2}, then the unique element of y_{u}^{0} (resp. Z_{u}^{1}) is in fact characterized by being zero-dimensional, P_{u} , and nowhere P_{u} (resp. P_{u} , and nowhere P_{u}).

Finally, from the proof of theorem 4.3.8, we see that

4.3.9 THEOREM: If $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, $u(0) \geq 2$, and $X \in Y_{u}^{0} \cup Y_{u}^{1} \cup Z_{u}^{0} \cup Z_{u}^{1}$, then the Cantor set is homogeneous with respect to dense copies of X. \Box

4.4 The main theorem, part 2

This section will provide the justification for the use of the hypothesis " $u(0) \ge 2$ " in many of the lemmas of the preceding sections: we will show that, if X is a homogeneous Borel set in 2^{ω} with X $\notin \Delta(D_{\omega}(\Sigma_2^0))$, then [X] $\in \{\Gamma_u, \Upsilon_u\}$ for some $u \in D$ with $u(0) \ge 2$. Together with the results of section 4.3, this will easily yield the main theorem of this chapter, completing the classification of all homogeneous zero-dimensional absolute Borel sets according to topological type.

4.4.1 LEMMA: Let $u \in D$, $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and u(0) = 1. If $X \in \Gamma_{u} \cup \overset{\vee}{\Gamma}_{u}$ is non-empty, then X contains a non-empty clopen subset U such that $U \in \Gamma_{v} \cup \overset{\vee}{\Gamma}_{v}$ for some described class Γ_{v} with $\Gamma_{v} < \Gamma_{u}$.

Proof: <u>Case 1</u>: t(u) = 1. By lemma 4.2.11(f), $\Gamma_{u} = \text{Bisep}(\Sigma_{1}^{0},\Gamma_{u}^{-})$, and since $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, we have $\overline{u}(0) \geq 1$. If $X \in \Gamma_{u}$, say $X = (A_{0} \cap C_{0}) \cup (A_{1} \cap C_{1})$, then there are clopen subsets C_{m}^{n} of 2^{ω} , for each $n \in \{0,1\}$ and $m < \omega$, such that $C_{n} = \bigcup_{m=0}^{\infty} C_{m}^{n}$. Since $X = \bigcup_{m=0}^{\infty} (A_{0} \cap C_{m}^{0}) \cup \bigcup_{m=0}^{\infty} (A_{1} \cap C_{m}^{1}) \neq \emptyset$, also $A_{n} \cap C_{m}^{n} \neq \emptyset$ for some $n \in \{0,1\}$, $m < \omega$. But $A_{n} \cap C_{m}^{n} = X \cap C_{m}^{n}$ since $C_{0} \cap C_{1}^{0} \neq \emptyset$, so $A_{n} \cap C_{m}^{n}$ is clopen in X; and by lemma 4.2.11(b), $A_{n} \cap C_{m}^{n} \in \Gamma_{\overline{u}}$. If $X \in \Upsilon_{u}$, say $2^{\omega} \setminus X = (A_{0} \cap C_{0}) \cup (A_{1} \cap C_{1})$, then $X = (C_{0} \cap 2^{\omega} \setminus A_{0}) \cup (C_{1} \cap 2^{\omega} \setminus A_{1}) \cup 2^{\omega} \setminus (C_{0} \cup C_{1})$. If some $C_{n} \cap 2^{\omega} \setminus A_{n} \neq \emptyset$, then proceed as above; otherwise, $X = 2^{\omega} \setminus (C_{0} \cup C_{1}) \in \Pi_{1}^{0} = \Upsilon_{1}^{-1} \cap (A_{1} \cap C_{n})$ for some strict-ly increasing sequence $(\Gamma_{u_{1}}: n < \omega)$ of described classes with $u_{n}(0) \ge 1$ for each $n < \omega$. If $X \in \Gamma_{u}$, say $X = \bigcup_{n=0}^{\infty} (A_{n} \cap C_{n})$, then we can write $C_{n} = \bigcup_{m=0}^{\infty} C_{m}^{n}$ for certain clopen subsets C_{m}^{n} of 2^{ω} , and proceed as in case 1. Similarly if $X \in \Upsilon_{u}$.

(i) If $C_{\zeta} \cap X = \emptyset$ for all $\zeta < \eta$, then $X = A_1 \setminus (\bigcup_{\zeta < \eta} C_{\zeta})$. Since $A_1 \in \Gamma_{u*}$, and $2^{\omega} \setminus (\bigcup_{\zeta < \eta} C_{\zeta}) \in \Pi_1^0 \subset \Delta_2^0$, we have $X \in \Gamma_{u*}$ by lemma 4.2.11(b), and $\Gamma_{u*} < \Gamma_u$ since $\Gamma_{u*} \cup \Upsilon_{u*} \subset \Gamma_u$.

(ii) Let $\alpha < \eta$ be minimal with $C_{\alpha} \cap X \neq \emptyset$. If α and η are both even or both odd, then $C_{\alpha} \setminus (U_{\beta < \alpha} C_{\beta}) \subset 2^{\omega} \setminus C$, so $C_{\alpha} \cap X = C_{\alpha} \cap A_{1} \setminus C$. Since $C_{\alpha} \cap 2^{\omega} \setminus C \in \Delta_{2}^{0}$, we have $C_{\alpha} \cap X \in \Gamma_{u*}$ as above, and $C_{\alpha} \cap X$ is a non-empty open subset of X; if U is clopen in 2^{ω} with $\emptyset \neq U \cap X \subset C_{\alpha} \cap X$, then $U \cap X$ $\epsilon \Gamma_{u^*}$ by lemma 4.2.11(b), so $U \cap X$ is the required clopen subset of X. If α is even and η is odd or conversely, then $C_{\alpha} \setminus (U_{\beta < \alpha} C_{\beta}) \subset C$, so $C_{\alpha} \cap X = C_{\alpha} \cap C \cap A_0 \in \Gamma_{u^*}$; we find the required clopen subset of X as above.

If $X \in \overset{Y}{\Gamma}_{u}$, then $2^{\omega} \setminus X = (A_{0} \cap C) \cup (A_{1} \setminus C)$, so $X = ((2^{\omega} \setminus A_{0}) \cap C) \cup (2^{\omega} \setminus A_{1}) \setminus C$. Put $\widetilde{A}_{0} = 2^{\omega} \setminus A_{0} \in \Gamma_{u*}$, $\widetilde{A}_{1} = 2^{\omega} \setminus A_{1} \in \overset{Y}{\Gamma}_{u*}$, and argue as above. <u>Case 4</u>: $u = 1^{3} \cap \langle u_{0}, u_{1} \rangle$, where $\Gamma_{u_{1}} < \Gamma_{u_{0}}$, $u_{0}(0) > 1$, and $u_{1}(0) \ge 1$; then $\Gamma_{u} = \text{Bisep}(D_{\eta}(\Sigma_{1}^{0}), \Gamma_{u_{0}}, \Gamma_{u_{1}})$. If $X \in \Gamma_{u}$, then by lemma 4.2.11(e), we can write $X = (X \cap C) \cup B \setminus C$ for some $C \in \Sigma_{1}^{0}$, $B \in \Gamma_{u_{1}}$ such that $X \cap C \in \text{Bisep}(D_{\eta}(\Sigma_{1}^{0}), \Gamma_{u_{0}}) = \Gamma_{v}$, where $v = 1^{3} \cap \langle u_{0}, \underline{0} \rangle$. Clearly, $\Gamma_{v} \subset \Gamma_{u}$, and if $Y \in \overset{Y}{\Gamma}_{v}$, say $2^{\omega} \setminus Y = (A_{0} \cap C_{0}) \cup (A_{1} \cap C_{1})$, then

$$\mathbb{Y} = ((2^{\omega} \setminus \mathbb{A}_{1}) \cap \mathbb{C}_{1}) \cup ((2^{\omega} \setminus \mathbb{A}_{0}) \cap \mathbb{C}_{0}) \cup 2^{\omega} \setminus (\mathbb{C}_{0} \cup \mathbb{C}_{1})$$

 $\epsilon \ \Gamma_{u}$, because $2^{\omega} \epsilon \ \Gamma_{u_{1}}$ since $u_{1}(0) \ge 1$; so $\Gamma_{v} \subset \Gamma_{u}$, whence $\Gamma_{v} < \Gamma_{u}$. Now if $X \cap C \ne \emptyset$, and U is a clopen subset of 2^{ω} such that $\emptyset \ne U \cap X \subset C \cap X$, then $U \cap X \epsilon \Gamma_{v}$ and we are done. Otherwise, $X = B \setminus C$; but $B \epsilon \Gamma_{u_{0}}$, $2^{\omega} \setminus C \epsilon \ \Delta_{2}^{0}$, and $u_{0}(0) \ge 2$, so $B \setminus C \epsilon \ \Gamma_{u_{0}}$ by lemma 4.2.11(b). Hence $X \epsilon \ \Gamma_{u_{0}} < \Gamma_{u}$.

If $X \in \mathring{\Gamma}_{u}$, then again by lemma 4.2.11(e), we have $2^{\omega} \setminus X = ((2^{\omega} \setminus X) \cap C) \cup B \setminus C$ for some $C \in \Sigma_{1}^{0}$, $B \in \Gamma_{u_{1}}$ with $C \setminus (2^{\omega} \setminus X) = C \cap X \in Bisep(D_{\eta}(\Sigma_{1}^{0}), \Gamma_{u_{0}})$. Thus, $X = (X \cap C) \cup (2^{\omega} \setminus B) \setminus C$. Since $2^{\omega} \setminus B \in \mathring{\Gamma}_{u_{1}} < \Gamma_{u_{0}}$, we can argue as above. $Case 5: u = 1^{5} \cap \langle u_{0}, u_{1} \rangle$, where $\Gamma_{u_{1}} < \Gamma_{u_{0}}$, $u_{1}(0) \ge 1$, and $\Gamma_{u_{0}} = SU(\Sigma_{1}^{0}, U_{n=0})$ $U_{n=0}^{\infty} \Gamma_{v_{n}}$ for some strictly increasing sequence $\langle \Gamma_{v_{n}}: n < \omega \rangle$, where $\langle v_{n}(0): n < \omega \rangle$ is non-decreasing, and $\sup_{n < \omega} v_{n}(0) > 1$. Then $\Gamma_{u} = SD_{\eta}(\langle \Sigma_{1}^{0}, \Gamma_{u_{0}} \rangle, \Gamma_{u_{1}})$. If $X \in \Gamma_{u}$, say $X = U_{\zeta < \eta}(A_{\zeta} \setminus (U_{\beta < \zeta} C_{\beta})) \cup B \setminus (U_{\zeta < \eta} C_{\zeta})$, and if we put $C = U_{\zeta < \eta} C_{\zeta}$, then $X = (X \cap C) \cup B \setminus C$, with $C \in \Sigma_{1}^{0}$, $B \in \Gamma_{u_{1}}$, and $X \cap C \in SD_{\eta}(\langle \Sigma_{1}^{0}, \Gamma_{u_{0}} \rangle) = \Gamma_{v}$, where $v = 1^{5} \cap (\langle u_{0}, 0 \rangle)$. As in case 4, we will be done if we can show that $\Gamma_{v} < \Gamma_{u}$. Clearly, $\Gamma_{v} \subset \Gamma_{u}$, so it suffices to prove: $CLAIM: \mathring{\Gamma}_{v} \subset \Gamma_{u}$.

Indeed, let
$$Y \in I_{V}^{V}$$
, say $2^{\omega} \setminus Y = \bigcup_{\zeta < \eta} (B_{\zeta} \setminus (\bigcup_{\beta < \zeta} D_{\beta}))$. Then
 $X = \bigcup_{\zeta < \eta} (\widetilde{B} \setminus (\bigcup_{\beta < \zeta} D_{\beta})) \cup (2^{\omega} \setminus (\bigcup_{\beta < \zeta} D_{\beta}))$

$$Y = \bigcup_{\zeta < \eta} (B_{\zeta} \setminus (\bigcup_{\beta < \zeta} D_{\beta})) \cup 2^{\infty} \setminus (\bigcup_{\zeta < \eta} D_{\zeta})$$

where $\widetilde{B}_{\zeta} = (D_{\zeta} \setminus B_{\zeta}) \cup U_{\beta < \zeta} D_{\beta}$. Now since $B_{\zeta} \in \Gamma_{u_0}$ with envelop D_{ζ} , say $B_{\zeta} = \bigcup_{n=0}^{\infty} (B_n^{\zeta} \cap E_n^{\zeta})$, with $D_{\zeta} = \bigcup_{n=0}^{\infty} E_n^{\zeta}$, we can write $\widetilde{B}_{\zeta} = \bigcup_{n=0}^{\infty} ((2^{\omega} \setminus B_n^{\zeta}) \cap E_n^{\zeta}) \cup U_{\beta < \zeta} D_{\beta}$.

Since each B_n^{ζ} is in some $\Gamma_{v_m} < \Gamma_{v_{m+1}}$, we have $2^{\omega} \setminus B_n^{\zeta} \in \Gamma_{v_{m+1}}$, and of course we can assume that $v_{m+1}(0) \ge 2$. By lemma 4.2.11(b), also

 $(2^{\omega} \setminus B_n^{\zeta}) \cup \bigcup_{\beta < \zeta} D_{\zeta} \in \Gamma_{v_m+1}$, and if we let $\langle F_n^{\zeta}: n < \omega \rangle$ reduce $\langle E_n^{\zeta} \cup \bigcup_{\beta < \zeta} D_{\beta}: n < \omega \rangle$, then

$$\widetilde{B}_{\zeta} = U_{n=0}^{\infty} (((2^{\omega} \setminus B_{n}^{\zeta}) \cup U_{\beta < \zeta} D_{\beta}) \cap F_{n}^{\zeta})$$

 $\epsilon \Gamma_{u_0}$, with envelop $\bigcup_{n=0}^{\infty} E_n^{\zeta} \cup \bigcup_{\beta < \zeta} D_{\beta} = \bigcup_{\beta \leq \zeta} D_{\beta} = D_{\zeta}$. Since clearly $\widetilde{B}_{\zeta} \subset D_{\zeta} \subset \widetilde{B}_{\zeta+1}$, and since $2^{\omega} \epsilon \Gamma_{u_1}$ because $u_1(0) \ge 1$, we have $\Upsilon \epsilon \Gamma_{u_0}$. This proves the claim.

Finally, let $X \in \Upsilon_{u}$, say $2^{\omega} \setminus X = ((2^{\omega} \setminus X) \cap C) \cup B \setminus C$, with $C \in \Sigma_{1}^{0}$, $B \in \Gamma_{u_{1}}$, and $Y = (2^{\omega} \setminus X) \cap C \in SD_{\eta}((\Sigma_{1}^{0},\Gamma_{u_{0}})) = \Gamma_{v}$, as above. Then $X = (X \cap C) \cup (2^{\omega} \setminus B) \setminus C$. Now if $X \cap C \neq \emptyset$, let U be a clopen subset of 2^{ω} such that $U \cap X \neq \emptyset$ and $U \subset C$; then $U \cap X = U \cap C \cap X = U \cap C \cap (2^{\omega} \setminus Y) = U \cap (2^{\omega} \setminus Y)$ $\in \Upsilon_{v}$ since v(0) = 1, by lemma 4.2.11(b). Otherwise, $X = (2^{\omega} \setminus B) \setminus C \in \Gamma_{u_{0}} \leq \Gamma_{u}$.

4.4.2 LEMMA: Let $u \in D$, $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 2$. If X is a homogeneous subset of 2^{ω} , and A is a non-empty open subset of X such that A $\in \Gamma_{u}$, then $X \in \Gamma_{u}$; similarly for $\stackrel{V}{\Gamma}_{u}$, and hence for $\Delta(\Gamma_{u})$.

Proof: let $x \in A$, and for each $y \in X$, let $h_y: X \approx X$ be such that $h_y(x) = y$. Then $\{h_y[A]: y \in X\}$ is an open covering of X, which has a countable subcovering $\{U_n: n < \omega\}$. For each $n < \omega$, let V_n be open in 2^{ω} such that $V_n \cap X = U_n$, and let $\langle W_n: n < \omega \rangle$ reduce $\langle V_n: n < \omega \rangle$. Since $W_n \cap X = W_n \cap V_n \cap X = W_n \cap V_n \cap X = W_n \cap U_n$, and since $U_n \in \Gamma_u$ by lemma 4.2.16, we have $X = \bigcup_{n=0}^{\infty} (V_n \cap U_n) \in SU(\Sigma_1^0, \Gamma_u) = \Gamma_u$ by lemma 4.2.11(a). The proof for $\stackrel{\vee}{\Gamma}_u$ is identical.

4.4.3 LEMMA: Let X be a homogeneous Borel subset of 2^{ω} such that X $\notin \Delta(D_{\omega}(\Sigma_{2}^{0}))$. Then $[X] \in {\Gamma_{u}, \overset{\vee}{\Gamma}_{u}}$ for some $u \in D$ with $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$ and $u(0) \geq 2$.

Proof: Let $\{\Gamma_{\mathbf{u}}, \overset{\mathsf{Y}}{\Gamma}_{\mathbf{u}}\}$ be a minimal non-self-dual pair in the Borel Wadge hierarchy such that $A \in \Gamma_{\mathbf{u}} \cup \overset{\mathsf{Y}}{\Gamma}_{\mathbf{u}}$ for some non-empty open subset A of X. By lemma 4.4.2, if $A \in \Delta(\mathsf{D}_{\omega}(\Sigma_2^0))$, then $X \in \Delta(\mathsf{D}_{\omega}(\Sigma_2^0))$ (for $\mathsf{D}_{\omega}(\Sigma_2^0) = \Gamma_{\mathbf{v}}$, where $\mathbf{v} = 2^{1} \widehat{\mathsf{w}} \underbrace{0}$), a contradiction; thus, $\Delta(\mathsf{D}_{\omega}(\Sigma_2^0)) \subset \Gamma_{\mathbf{u}}$. So if u(0) = 1, then we can apply lemma 4.4.1 to obtain a non-empty open subset B of A such that $B \in \Gamma_{\mathbf{v}} \cup \overset{\mathsf{Y}}{\Gamma}_{\mathbf{v}}$ for some described class $\Gamma_{\mathbf{v}} < \Gamma_{\mathbf{u}}$, contradicting minimality of $\{\Gamma_{\mathbf{u}}, \overset{\mathsf{Y}}{\Gamma}_{\mathbf{u}}\}$. So we must have $u(0) \ge 2$, and thus we can apply lemma 4.4.2 again to obtain that $X \in \Gamma_{\mathbf{u}}$ (if $A \in \Gamma_{\mathbf{u}}$) or $X \in \overset{\mathsf{Y}}{\Gamma}_{\mathbf{u}}$ (if $A \in \overset{\mathsf{Y}}{\Gamma}_{\mathbf{u}}$), and hence clearly $[X] \in \{\Gamma_{\mathbf{u}}, \overset{\mathsf{Y}}{\Gamma}_{\mathbf{u}}\}$, again by minimality of $\{\Gamma_{\mathbf{u}}, \overset{\mathsf{Y}}{\Gamma}_{\mathbf{u}}\}$.

We can now state the main theorem of this chapter.

Let $D_0 = \{u \in D: \Delta(D_{\omega}(\Sigma_2^0)) \subset \Gamma_u, u(0) \ge 2\}$, and $D_1 = \{u \in D_0: u(0) \ge 3 \text{ or } t(u) = 3\}$. By lemmas 4.3.5 and 4.3.6, there are unique, homogeneous spaces $Y_u^0 \in Y_u^0, Z_u^1 \in Z_u^1$ for each $u \in D_0$, and $Y_u^1 \in Y_u^1, Z_u^0 \in Z_u^0$ for each $u \in D_1$. Let Δ denote the class of all spaces that are in $\Delta(D_{\omega}(\Sigma_2^0))$ in the sense of definition 3.1.1(c), i.e. in the absolute sense.

4.4.4 THEOREM: The topological types of homogeneous zero-dimensional absolute Borel sets that are not in Δ are precisely those of the spaces Y_{u}^{0} , Z_{u}^{1} for $u \in D_{0}$, and the spaces Y_{u}^{1}, Z_{u}^{0} for $u \in D_{1}$.

Proof: From theorem 4.3.8, we see that these spaces are homogeneous. So let X be a homogeneous zero-dimensional absolute Borel set which is not in Δ , and let Y be a copy of X in 2^{ω} ; then Y $\notin \Delta(D_{\omega}(\Sigma_{2}^{0}))$, so by lemma 4.4.3, $[Y] \in {\Gamma_{u}, \mathring{\Gamma}_{u}}$ for some $u \in D$ with $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$ and $u(0) \ge 2$. Consider the case $[Y] = \Gamma_{u}$ (the other case is similar). By the remark following definition 4.3.1, X is P_{u} ; and if A is a non-empty open subset of Y which is in $\mathring{\Gamma}_{u}$, then $Y \in \mathring{\Gamma}_{u}$ by lemma 4.4.2, a contradiction, so X is nowhere \check{P}_{u} . By 1.12.1, X is either first category or Baire, so $X \in \mathscr{Y}_{u}^{0} \cup \mathscr{Y}_{u}^{1} \cup Z_{u}^{0} \cup Z_{u}^{1}$.

With theorems 3.4.13, 3.5.9, 3.6.2, 4.3.8, and 4.4.4, we have now obtained topological characterizations of all homogeneous zero-dimensional absolute Borel sets.

Remark: The homogeneous zero-dimensional elements of $\Delta_3^0 \setminus \Delta$ have been described twice, by theorems 3.6.2 and 4.4.4. In section 4.6, we will relate the two classifications.

Remark: Since Steel's theorem always yields autohomeomorphisms of 2^{ω} , the methods used here can only produce spaces with respect to which the Cantor set is dense homogeneous (see theorem 4.3.9); thus it is clear that we can not extend theorem 4.4.4 to include the elements of Δ as well, since the Cantor set is not homogeneous with respect to dense copies of the spaces X_{4k} and X_{4k+1} of Chapter 3, where $k < \omega$ (see theorem 4.4.8).

Since there are only ω_1 Borel Wadge classes (see e.g. the second remark following definition 4.2.2), the above theorem, together with corollary 3.6.3, yields:
4.4.5 COROLLARY: There are pecisely ω_1 homogeneous zero-dimensional absolute Borel sets. $\hfill \Box$

In fact, by analyzing Louveau's description of the Wadge classes, it is not hard to see that there are precisely ω_1 homogeneous zero-dimensional absolute Borel sets of exactly ambiguous class α , for each $\alpha \ge 2$. In section 4.5, we will examine the homogeneous zero-dimensional absolute Borel sets that are exactly of additive or multiplicative class α , for $\alpha \ge 2$.

Another corollary to theorems 3.6.2 and 4.4.4 is the following strange characterization of $C \setminus \{p\}$.

4.4.6 COROLLARY: Up to homeomorphism, $C \setminus \{p\}$ is the only non-discrete, zero-dimensional absolute Borel set which is homogeneous, but not strongly homogeneous.

As an analogue to theorem 3.6.4, we have:

4.4.7 THEOREM: Let A be a zero-dimensional space. (a) If $u \in D_0$, and A is P_u , then A can be embedded as a closed subspace of Y_u^0 . (b) If $u \in D_0$, and A is P_u , then A can be embedded as a closed subspace of Z_u^1 . (c) If $u \in D_1$, and A is P_u , then A can be embedded as a closed subspace of Y_u^1 .

(d) If $u \in D_1$, and A is \not{P}_u , then A can be embedded as a closed subspace of Z_u^0 .

Proof:(a) For each $k \in N$, let X_k be a dense copy of Y_u^0 in $\{1/k\} \times 2^{\omega}$, and consider A as a subset of $\{0\} \times 2^{\omega}$. We claim that, as a subset of $[0,1] \times 2^{\omega}$, $\widetilde{A} = A \cup \bigcup_{k=1}^{\infty} X_k \approx Y_u^0$. Indeed, since A is nowhere dense in \widetilde{A} , \widetilde{A} is nowhere $\overset{P}{P}_u$, and clearly \widetilde{A} is first category. Also, if we identify ($\{0\} \cup \{1/k: k \in N\}) \times 2^{\omega}$ with 2^{ω} , then by lemma 4.2.16, $\widetilde{A} \in SU(\prod_1^0, \Gamma_u) \subset SU(\Sigma_2^0, \Gamma_u)$ = Γ_u by lemma 4.2.11(a), so \widetilde{A} is P_u . Clearly, A is closed in \widetilde{A} . (b) Let X_k be as in (a), and put $B = (\{0\} \times 2^{\omega}) \setminus A \cup \bigcup_{k=1}^{\infty} X_k$; then $B \approx Y_u^0$ as in (a). Hence, since B is dense in $X = (\{0\} \cup \{1/k: k \in N\}) \times 2^{\omega}$, we have $\widetilde{A} = X \setminus B \approx Z_u^1$ as in the proof of lemma 4.3.4; again, A is closed in \widetilde{A} . (d) is proved as (a), using corollary 4.2.14 instead of lemma 4.2.11(a), and (c) follows from (d) exactly as we proved (b) from (a). Finally, from corollaries 3.2.7 and 3.2.9, and theorems 3.4.14, 3.5.10, and 4.3.9, we can deduce:

4.4.8 THEOREM: Let X be a homogeneous zero-dimensional absolute Borel set. Then the Cantor set is homogeneous with respect to dense copies of X if and only if $C \setminus X_0 \approx C \setminus X_1$ for all dense copies X_0, X_1 of X in C.

4.5 Some applications

The results of the preceding sections are rather abstract; in particular, the construction of the elements of y_u^0 , y_u^1 , Z_u^0 , and Z_u^1 in lemmas 4.3.5 and 4.3.6 does not make it very clear as to what kind of space we end up with. Therefore, we will begin this section by showing that if $u = \alpha^{1} 1^{1} 0$ for some $\alpha \ge 3$, then there is a very elegant way of describing elements of y_u^0 , y_u^1 , Z_u^0 , and Z_u^1 (each of these classes is non-empty since $u(0) \ge 3$). The spaces that we want are obtained from the construction process for Borel sets of exact class, due to Sikorski [54] (see also Engelking, Holsztyński, and Sikorski [16], and Freiwald, McDowell, and McHugh [17]), which for 2^{ω} is the following:

Let $p \in 2^{\omega}$, and put $M_1 = \{p\}$, $A_1 = 2^{\omega} \setminus M_1$. If $\alpha \in [2, \omega_1)$, and A_{β}, M_{β} have been defined for $\beta < \alpha$, then $M_{\alpha} = \prod_{i=0}^{\infty} A_{\gamma} \subset \prod_{i=0}^{\infty} 2^{\omega} \approx 2^{\omega}$ if $\alpha = \gamma+1$, $M_{\alpha} = \prod_{\beta < \alpha} A_{\beta} \subset \prod_{\beta < \alpha} 2^{\omega} \approx 2^{\omega}$ if α is a limit.

In both cases, put $A_{\alpha} = 2^{\omega} \setminus M_{\alpha}$.

Sikorski showed that A_{α} is Σ_{α}^{0} but not Π_{α}^{0} in 2^{ω} , and M_{α} is Π_{α}^{0} but not Σ_{α}^{0} in 2^{ω} .

For $\alpha \in [2, \omega_1)$, put $u(\alpha) = \alpha^1 1^{\underline{0}}$; note that $y_{u(\alpha)}^0$ consists of all zerodimensional spaces that are Σ_{α}^0 , nowhere Π_{α}^0 , and first category, $y_{u(\alpha)}^1$ consists of all zero-dimensional spaces that are Σ_{α}^0 , nowhere Π_{α}^0 , and Baire, and similarly for Z_{u}^0, Z_{u}^1 .

4.5.1 THEOREM: Let $\alpha \in [2, \omega_1)$. (a) If $1+\alpha$ is even, then $M_{\alpha} \in Z_{u(\alpha)}^{0}$, $A_{\alpha} \in Y_{u(\alpha)}^{1}$, $\widetilde{A}_{\alpha} = \mathbf{Q} \times A_{\alpha} \in Y_{u(\alpha)}^{0}$, and $\widetilde{M}_{\alpha} = 2^{\omega} \setminus \widetilde{A}_{\alpha} \in Z_{u(\alpha)}^{1}$, where \widetilde{A}_{α} is densely embedded in 2^{ω} . (b) If $1+\alpha$ is odd, then $M_{\alpha} \in Z_{u(\alpha)}^{1}$, $A_{\alpha} \in Y_{u(\alpha)}^{0}$, and if $\alpha \geq 3$, then $\widetilde{M}_{\alpha} = \mathbf{Q} \times M_{\alpha} \in Z_{u(\alpha)}^{0}$, and $\widetilde{A}_{\alpha} = 2^{\omega} \setminus \widetilde{M}_{\alpha} \in Y_{u(\alpha)}^{1}$, where \widetilde{M}_{α} is densely embedded in 2^{ω} . Proof: The proof is by induction on α . If $\alpha = 2$, then the theorem is clear, since it is not hard to see that $M_2 \approx P$ and $A_2 \approx Q \times C$. So suppose that the theorem holds for all $\beta < \alpha$. Assume for example that α is a limit (the other cases are entirely similar); then $1+\alpha = \alpha$ is even, and $M_{\alpha} =$ $\Pi_{\beta < \alpha} A_{\beta}$. Now $A_1 \times A_2 \approx C \setminus \{p\} \times Q \times C \approx Q \times C$ is strongly homogeneous, and if $\beta \geq 3$, then $A_\beta \in \mathcal{Y}_{u(\beta)}^0 \cup \mathcal{Y}_{u(\beta)}^1$ is strongly homogeneous by theorem 4.3.8. So if U is a non-empty basic clopen subset of M_{α} , then $U \approx M_{\alpha} \notin \Sigma_{\alpha}^0$; hence M_{α} is nowhere Σ_{α}^0 . Since A_{β} is first category if $1+\beta$ is odd, M_{α} is first category, so $M_{\alpha} \in Z_{u(\alpha)}^0$. It is now easily deduced that $A_{\alpha} \in \mathcal{Y}_{u(\alpha)}^1$, $Q \times A_{\alpha} \in$ $\mathcal{Y}_{u(\alpha)}^0$, and $2^{u} \setminus (Q \times A_{\alpha}) \in Z_{u(\alpha)}^1$.

Thus, the homogeneous Borel sets of exact class 2 are precisely Q^{ω} , $2^{\omega} \setminus Q^{\omega}$, $Q \times (2^{\omega} \setminus Q^{\omega})$, and $2^{\omega} \setminus (Q \times (2^{\omega} \setminus Q^{\omega}))$; those of exact class 3 are $(2^{\omega} \setminus Q^{\omega})^{\omega} = Z$, $2^{\omega} \setminus Z$, $Q \times Z$, and $2^{\omega} \setminus (Q \times Z)$, etc..

Note that if $u = u(3) = 3^{1}_{0}$, then theorem 4.4.7(d) says that any zerodimensional absolute $F_{\sigma\delta}$ can be embedded as a closed subspace of Q^{ω} ; this was proved independently also by van Douwen [9] and Junnila [24]. In fact, if we let $Y_{\alpha} = \tilde{A}_{\alpha}$ (resp. A_{α}), and $Z_{\alpha} = M_{\alpha}$ (resp. \tilde{M}_{α}) if 1+ α is even (resp. odd), for $\alpha \ge 3$ (so actually $Y_{\alpha} \in Y_{u(\alpha)}^{0}$, $Z_{\alpha} \in Z_{u(\alpha)}^{0}$, both first category), then we have the following analogue to corollaries 2.1.3, 2.2.4, 2.3.3, 2.4.3, and 2.4.7:

4.5.2 THEOREM: Let X be a non-empty zero-dimensional space, and let $\alpha \ge 3$. (a) If X is Σ^{0}_{α} , then $X \times Y_{\alpha} \approx Y_{\alpha}$. (b) If X is Π^{0}_{α} , then $X \times Z_{\alpha} \approx Z_{\alpha}$.

Proof: Immediate from the characterizations.

Thus, if X is a non-empty zero-dimensional absolute $F_{\sigma\delta}$, then $X \times Q^{\omega} \approx Q^{\omega}$; this was first proved in van Engelen [12].

Remark: We could not have chosen $Y_{\alpha} \in Y_{u(\alpha)}^{1}$, $Z_{\alpha} \in Z_{u(\alpha)}^{1}$ in the above theorem since $X \times Y_{\alpha}, X \times Z_{\alpha}$ are not necessarily Baire.

For the spaces A_{α} , \widetilde{A}_{α} , M_{α} , and \widetilde{M}_{α} , we now derive Hurewicz-type theorems. We need three lemmas; the first is due to Harrington (see the proof of lemma 3 in Steel [56]).

4.5.3 LEMMA: Let Γ be a reasonably closed class of Borel sets in 2^{ω} ; let $A, B \subset 2^{\omega}$ be such that $A \in \Gamma$, $B \notin \overset{V}{\Gamma}$. Then there is a one-to-one continuous f: $2^{\omega} \rightarrow 2^{\omega}$ such that $A = f^{-1}[B]$.

4.5.4 LEMMA (Kunen and Miller [26], theorem 4): Suppose A is a Borel subset of a complete space X. Then for any $\alpha < \omega_1$, A is Π^0_{α} in X if and only if for all compact zero-dimensional $P \subset X$, $P \cap X$ is Π^0_{α} in P.

The final lemma is a Hurewicz-type result itself, but only for zero-dimensional spaces.

4.5.5 LEMMA: Let B be a zero-dimensional absolute Borel set, and let u ϵ D be such that $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u}$, and $u(0) \geq 2$. Let Y be P_{u} , nowhere P_{u} , and let Z be P_{u} , nowhere P_{u} . (a) If B is not P_{u} , then B contains a closed copy of Y. (b) If B is not P_{u} , then B contains a closed copy of Z. Proof:(a) Embed B and Y in 2^{ω} ; then $B \notin \Gamma_{u}$, $Y \in \Gamma_{u}$, so by lemma 4.2.17 we can apply lemma 4.5.3 to obtain a continuous one-to-one mapping f: $2^{\omega} \rightarrow 2^{\omega}$ such that $Y = f^{-1}[B]$. Then f: $2^{\omega} \rightarrow f[2^{\omega}]$ is a homeomorphism, so $Y \approx f[Y]$, and $f[Y] = f[2^{\omega}] \cap B$ is closed in B. The proof of (b) is analogous.

Remark: The above proof, and the proof of the next theorem were motivated by a suggestion of A.W. Miller.

4.5.6 THEOREM: Let A be a Borel subset of a compact space X, and $\alpha \ge 3$. (a) If A is not Π^0_{α} , then X contains Cantor sets K_0 and K_1 such that $K_0 \cap A \approx A_{\alpha}$, $K_0 \setminus A \approx M_{\alpha}$, and $K_1 \cap A \approx \widetilde{A}_{\alpha}$, $K_1 \setminus A \approx \widetilde{M}_{\alpha}$. (b) If A is not Σ^0_{α} , then X contains Cantor sets K_0 and K_1 such that $K_0 \cap A \approx M_{\alpha}$, $K_0 \setminus A \approx A_{\alpha}$, and $K_1 \cap A \approx \widetilde{M}_{\alpha}$, $K_1 \setminus A \approx \widetilde{A}_{\alpha}$.

Proof: Clearly, (b) follows from (a). So suppose that A is not Π_{α}^{0} . By lemma 4.5.4, X contains a compact zero-dimensional space K such that $B = K \cap A$ is not Π_{α}^{0} . Now apply lemma 4.5.5 to B and $u = u(\alpha) = \alpha^{-1} 1^{-1} 0$ to obtain closed subsets $A_{0} \approx A_{\alpha}$, $A_{1} \approx \widetilde{A}_{\alpha}$ of B; then $K_{0} = \overline{A}_{0}$, $K_{1} = \overline{A}_{1}$ are as required.

The construction of the sets M_{α} and A_{α} naturally led Sikorski, in [54], to the following question (Coll. Math. problem P.215):

QUESTION: Let B_n be a Σ_{α}^0 -set in a space X_n , but not a Π_{α}^0 -set in X_n (n = 0,1,2,...). Prove or disprove that the set $B = \Pi_{n=0}^{\infty} B_n$ (which is, of course, a $\Pi_{\alpha+1}^0$ -set) is not a $\Sigma_{\alpha+1}^0$ -set in $X = \Pi_{n=0}^{\infty} X_n$.

We give a partial answer to this question:

4.5.7 THEOREM: Let $\alpha \in [1, \omega_1)$. In the above question, $\prod_{n=0}^{\infty} B_n$ is not a Σ_{n+1}^0 -set absolutely.

Proof: If $\alpha = 1$, then \mathbb{B}_n is not compact, so it contains a closed copy of ω ; hence B contains a closed copy of $\omega^{\omega} \approx P$, which is not σ -compact, so B is not Σ_2^0 . If $\alpha = 2$, then by theorem 3.3.1, \mathbb{B}_n contains a closed copy of Q, so B contains a closed copy of $Q^{\omega} \approx \mathbb{M}_3$; and if $\alpha \ge 3$, then by theorem 4.5.6, \mathbb{B}_n contains a closed copy of \mathbb{A}_{α} , so B contains a closed copy of $\mathbb{G}_{\alpha+1}^{\infty} = \mathbb{M}_{\alpha+1}$. In both cases, since $\mathbb{M}_{\alpha+1}$ is not a $\Sigma_{\alpha+1}^0$ -set, neither is B.

Remark: Sikorski took X_n to be a metric space, not mentioning separability; thus, the question is also open for the non-separable case.

4.6 The ambiguous class 2 revisited

Recall the definitions of the classes X_{β}^{i} , $i \in \{1,2\}$, $\beta \in [\omega, \omega_{1})$ from definition 3.5.7: if X is zero-dimensional, $\alpha < \omega_{1}$ is a limit, and $n \in \mathbb{N}$, then:

In theorem 3.6.2 it was proved that the unique elements of these classes are precisely the homogeneous zero-dimensional absolute Borel sets in $\Delta_3^0 \setminus \Delta$, where Δ consists of all spaces that are $\Delta(D_{\omega}(\Sigma_2^0))$ absolutely. From theorem 4.4.4 it follows that each of the above classes is some y_u^0 , y_u^1 , Z_u^0 , or Z_u^1 ; in this section, we establish the exact correspondence.

We first determine which $u \in D$ determine a relevant Wadge class.

4.6.1 LEMMA: Let $u \in D$ be such that $\Delta(D_{\omega}(\Sigma_{2}^{0})) \subset \Gamma_{u} \subset \Delta_{3}^{0}$, and $u(0) \geq 2$. Then $u = 2^{1} n^{0}$ for some $n \in [\omega, \omega_{1})$, i.e. $\Gamma_{u} = D_{n}(\Sigma_{2}^{0})$.

Proof: It is easily seen from definition 4.2.2 that if $u(0) \ge 3$, then $\Gamma_u \ge \Sigma_3^0$; so u(0) = 2. If u(1) = 2 (resp. 3, resp. 4), then Γ_u contains Γ_{u*} (resp. Γ_{u_0} , resp. some Γ_{u_n}), where $u*(0) \ge 3$ (resp. $u_0(0) \ge 3$, resp. $u_n(0) \ge 3$), and thus $\Gamma_u \ge \Sigma_3^0$; if u(1) = 5, then Γ_u contains $\Gamma_{u_0} = SU(\Sigma_2^0, U_{n=0}^{\infty} \Gamma_{u_n})$, which contains Σ_3^0 by the previous case u(1) = 4. So u(1) = 1, and clearly $u(2) \ge \omega$.

Since for each $u = 2^1 n 0$ we have $t(u) \neq 3$ and $u(0) \not\geq 3$, we see from theorem 4.4.4 that the homogeneous zero-dimensional absolute Borel sets in $\Delta_3^0 \setminus \Delta$ are precisely

{ Y_{u}^{0}, Z_{u}^{1} : $u = 2^{1} \eta_{0}^{0}$ for some $\eta \in [\omega, \omega_{1})$ };

and by the remark following theorem 4.3.8, these spaces are characterized by being P_u and nowhere P_u , respectively P_u and nowhere P_u . Let $u(n) = 2^1^n^0$.

4.6.2 THEOREM: Let $\alpha < \omega_1$ be a limit ordinal. (a) For each $n < \omega$, $\chi^2_{\alpha+2n} = y^0_{u(\alpha+2n)}$ and $\chi^1_{\alpha+2n+1} = Z^1_{u(\alpha+2n)}$. (b) For each $n \in \mathbb{N}$, $\chi^1_{\alpha+2n} = y^0_{u(\alpha+2n-1)}$ and $\chi^2_{\alpha+2n-1} = Z^1_{u(\alpha+2n-1)}$. Proof: If we consider all classes as consisting of dense subsets of 2^{ω} , then $\chi^2_{\alpha+2n} = \chi^1_{\alpha+2n+1}$ and $\chi^1_{\alpha+2n} = \chi^2_{\alpha+2n-1}$ (inductive hypotheses (3) and (9) for the proof of theorems 3.5.9 - 3.5.11), and $y^0_{u(\beta)} = \chi^1_{u(\beta)}$ (proof of lemma 4.3.4). So it suffices to prove only the first statement in (a) and (b). Also, we need only prove one inclusion since each class contains exactly one topological type. (a) Let $X \in \chi^2_{\alpha+2n}$; then X is $P_{\alpha+2n}$ is $P_{u(\alpha+2n)}$. And if X is densely embedded in 2^{ω} , then $2^{\omega} \setminus X \in \chi^1_{\alpha+2n+1}$ is nowhere $P_{\alpha+2n}$, so $2^{\omega} \setminus X$ is nowhere $P_{\alpha+2n} = S_{\alpha} \times is$ powhere $\overset{V}{P}$ whence $X \in V^0_{\alpha}$

 $P_{u(\alpha+2n)}, \text{ so } X \text{ is nowhere } P_{u(\alpha+2n)}^{\alpha+2n+1}, \text{ whence } X \in y_{u(\alpha+2n)}^{\alpha+2n+1}.$ (b) Embed $X \in X_{\alpha+2n}^{1}$ densely in 2^{ω} ; then $2^{\omega} \setminus X \in X_{\alpha+2n-1}^{2}$ is $P_{\alpha+2(n-1)} \cup C$ complete, say $2^{\omega} \setminus X = D_{\beta}(\langle A_{\zeta}; \zeta < \beta \rangle) \cup G$, where G is complete, and $\beta = \alpha+2(n-1)$. Then

 $X = (U\{A_{\zeta} \setminus (U_{\gamma < \zeta} A_{\gamma}): \zeta \text{ even } < \beta\} \cup 2^{\omega} \setminus (U_{\zeta < \beta} A_{\zeta})) \cap 2^{\omega} \setminus G;$

thus, if we put $A_{\beta} = 2^{\omega}$, then $X = D_{\beta+1}(\langle A_{\zeta} \cap 2^{\omega} \backslash G: \zeta < \beta+1 \rangle) \in D_{\alpha+2n-1}(\Sigma_{2}^{0})$, i.e. X is $P_{u(\alpha+2n-1)}$. If U is a non-empty clopen subset of 2^{ω} , and

 $U \cap X$ is $\not P_{u(\alpha+2n-1)}$, then $U \setminus X$ is $P_{u(\alpha+2n-1)}$, say $U \setminus X = D_{\alpha+2n-1} (\langle A_{\zeta} \rangle$: $\zeta < \alpha + 2n - 1$): hence,

$$U \cap X = \{A_{\zeta} \setminus (U_{\gamma < \zeta} A_{\gamma}) : \zeta \text{ odd } < \alpha + 2(n-1)\} \cup 2^{\omega} \setminus A_{\alpha + 2(n-1)} = D_{\alpha}(\langle A_{\gamma} : \zeta < \beta \rangle) \cup 2^{\omega} \setminus A_{\alpha},$$

where $\beta = \alpha + 2(n-1)$. So $U \cap X$ is $P_{\beta} \cup \text{complete is } P_{\alpha+2n-1}$, a contradiction. Thus, X is nowhere $\mathcal{P}_{u(\alpha+2n-1)}$, i.e. $X \in \mathcal{Y}_{u(\alpha+2n-1)}^{0}$.

We finish this chapter by computing the Wadge classes of the other homogeneous absolute Borel sets, i.e. the spaces which were described and characterized in section 3.4 and Chapter 2. For this, recall the last remark following theorem 4.1.9, that if $\Gamma \subset \mathcal{P}(2^{\omega})$ is a continuously closed class of Borel sets, and $A \in \Gamma \setminus \Gamma$, then $\Gamma = [A]$.

4.6.3 LEMMA: Let
$$A \subseteq 2^{\omega}$$
, and $n \in \mathbb{N}$.
(a) $A \in \overset{\vee}{D}_{2n}(\Sigma_2^0)$ if and only if A is $P_{4(n-1)+2}^2$.
(b) $A \in D_{2n+1}(\Sigma_2^0)$ if and only if A is $P_{4(n-1)+3}^2$.
(c) $A \in \overset{\vee}{D}_{2n+1}(\Sigma_2^0)$ if and only if A is P_{4n}^2 .

Proof: Since $A \in D_{2n}(\Sigma_2^0)$ if and only if A is $P_{4(n-1)+1}$, (a) is merely a reformulation of lemma 3.4.4(b). Similarly, (c) follows from (b) and lemma 3.4.4(a). To prove (b), let $A \in D_{2n+1}(\Sigma_2^0)$, say $A = D_{2n+1}((A_m: m < 2n+1));$ then $A = A_0 \cup D_{2n}((A_{m+1}: m < 2n)))$ which is σ -compact $\cup S_n$ by lemma 3.1.4, so A is $P_{4(n-1)+3}^2$. Conversely, if A is $P_{4(n-1)+3}^2$ is $S_n \cup \sigma$ -compact, then by lemma 3.1.4 we can write $A = A_0 \cup D_{2n}(\langle A_m^{\prime}: m < 2n \rangle)$, where A_0 and each A'_m are σ -compact; if we put $A'_{m+1} = A'_m \cup A'_0$ for each m < 2n, then $(A_m: m < 2n+1)$ is increasing, and $A = D_{2n+1}((A_m: m < 2n+1)).$

4.6.4 THEOREM: Let
$$i \in \{1,2\}, k < \omega, and X \in 2^{\omega}$$
.
(a) If $X \in X_{4k+1}$, then $[X] = D_{2(k+1)}(\Sigma_{2}^{0})$.
(b) If $X \in X_{4k+2}^{i}$, then $[X] = D_{2(k+1)}(\Sigma_{2}^{0})$.
(c) If $X \in X_{4k+3}^{i}$, then $[X] = D_{2(k+1)+1}(\Sigma_{2}^{0})$.
(d) If $X \in X_{4(k+1)}$, then $[X] = D_{2(k+1)+1}(\Sigma_{2}^{0})$.
Proof: (a) Clearly, $X \in D_{2(k+1)}(\Sigma_{2}^{0})$. If $X \in \overset{\vee}{D}_{2(k+1)}(\Sigma_{2}^{0})$.

Proof: (a) Clearly, $X \in D_{2(k+1)}(\Sigma_2^0)$. If $X \in \overset{V}{D}_{2(k+1)}(\Sigma_2^0)$, then by lemma 4.6.3, X is P_{4k+2}^2 is $P_{4(k-1)+3}^2 \cup$ complete; since X is nowhere $P_{4(k-1)+3}^2$, X is Baire, contradicting lemma 3.4.7. Thus, $X \in D_{2(k+1)}(\Sigma_{2}^{0}) \setminus D_{2(k+1)}(\Sigma_{2}^{0})$, whence $[X] = D_{2(k+1)}(\Sigma_2^0).$

(b) By lemma 4.6.3, $X \in \overset{\vee}{D}_{2(k+1)}(\Sigma_{2}^{0})$. If $X \in D_{2(k+1)}(\Sigma_{2}^{0})$, then X is strong-

ly $\sigma - P_{4k}$, whence first category since X is nowhere P_{4k} . This contradicts lemma 3.4.7, so X $\notin D_{2(k+1)}(\Sigma_2^0)$, and $[X] = \bigvee_{2(k+1)}^{V}(\Sigma_2^0)$. The proof of (c) and (d) is similar.

The only thing left is to determine the Wadge classes of $C, C \setminus \{p\}, Q, Q \times C$, and P. The fact that P is Π_2^0 but not Σ_2^0 , and Q and $Q \times C$ are Σ_2^0 but not Π_2^0 , yields the following theorem:

4.6.5 THEOREM: Let
$$X \in 2^{\omega}$$
.
(a) If $X \approx P$, then $[X] = \Pi_2^0$.
(b) If $X \approx Q$ or $X \approx Q \times C$, then $[X] = \Sigma_2^0$.

For C and $C \setminus \{p\}$, the Wadge class turns out to depend upon the embedding in 2^{ω} . Note that from theorem 4.2.7 it is easily deduced that the Wadge hierarchy starts with



4.6.6 THEOREM: Let $C \approx X \subset 2^{\omega}$. (a) If $X = 2^{\omega}$, then $[X] = \{2^{\omega}\}$. (b) If $2^{\omega} \neq X$ is clopen in 2^{ω} , then $[X] = \Delta_1^0$. (c) If X is not open in 2^{ω} , then $[X] = \Pi_1^0$. Proof: (a) and (b) are trivial. If X is not open, then $X \in \Pi_1^0 \setminus \Sigma_1^0$, so $[X] = \Pi_1^0$.

4.6.7 LEMMA: Let $X \subset 2^{\omega}$. Then $X \in \overset{\vee}{D}_2(\Sigma_1^0)$ if and only if $X \setminus Int(X)$ is compact.

Proof: Since $D_2(\Sigma_1^0)$ consists of all subsets of 2^{ω} that are the intersection of an open subset and a closed subset of 2^{ω} , $\overset{V}{D}_2(\Sigma_1^0)$ consists of all subsets of 2^{ω} that are the union of an open set and a closed set. Now if

 $X = 0 \cup F$, with 0 open and F compact, then $0 \subset Int(X)$, so $X \setminus Int(X)$ is closed in F, whence compact. Conversely, if $X \setminus Int(X)$ is compact, then clearly $X = Int(X) \cup (X \setminus Int(X)) \in \overset{V}{D}_{2}(\Sigma_{1}^{0})$.

4.6.8 THEOREM: Let $\mathbb{C} \setminus \{p\} \approx x \in 2^{\omega}$. (a) If X is open, then $[X] = \Sigma_1^0$. (b) If X\Int(X) is non-empty and compact, then $[X] = \Delta(D_2(\Sigma_1^0))$. (c) If X\Int(X) is non-compact, then $[X] = D_2(\Sigma_1^0)$. Proof:(a) Since $X \in \Sigma_1^0 \setminus \Pi_1^0$, $[X] = \Sigma_1^0$. (b) $X \in \overset{V}{D}_2(\Sigma_1^0)$ by lemma 4.6.7; also, $X \in D_2(\Sigma_1^0)$ since X is locally compact. Thus, $[X] \subset \Delta(D_2(\Sigma_1^0))$, and hence $[X] = \Delta(D_2(\Sigma_1^0))$ since X $\notin \Sigma_1^0 \cup \Pi_1^0$. (c) X $\notin \overset{V}{D}_2(\Sigma_1^0)$ by lemma 4.6.7, so $\Delta(D_2(\Sigma_1^0)) \notin [X] \subset D_2(\Sigma_1^0)$ since X is locally compact; hence $[X] = D_2(\Sigma_1^0)$.

CHAPTER 5: RIGID BOREL SETS

The "propeller space" defined by de Groot and Wille in [19] is an example of a non-trivial compact rigid subset of \mathbb{R}^2 . A space with these properties cannot be constructed in \mathbb{R} : indeed, note that such a space would have to be zero-dimensional (since one can move points around in an interval), that the space could contain at most one isolated point (since interchanging two distinct isolated points and leaving the other points in place yields a nontrivial autohomeomorphism), and that removing this isolated point if necessary, we would still have a compact rigid space (since a clopen subset of a rigid space is rigid). Thus, we would have a rigid compact zero-dimensional space without isolated points, i.e. a rigid Cantor set (theorem 2.1.1); this is impossible.

On the other hand, a method is known for constructing non-trivial zero-dimensional rigid spaces (see e.g. Kuratowski [27]), but this method only yields spaces which are not even analytic. Thus, the question arises whether it is possible at all to construct "nice" zero-dimensional rigid spaces, say rigid zero-dimensional absolute Borel sets; this question was asked by van Douwen in [8].

In this chapter, we answer van Douwen's question in the negative; also, we briefly comment on the question of whether rigid zero-dimensional analytic spaces exist.

5.1 THEOREM: There are no non-trivial rigid zero-dimensional absolute Borel sets.

Proof: If there is a non-trivial, rigid, zero-dimensional absolute Borel set, then by an argument as above, there is also one without isolated points.

$$A_0 = \{A \subset 2^{\omega} : A \in \Delta(D_{\omega}(\Sigma_2^0)), A \text{ rigid, dense in itself}\},\$$

and

Put

 $A_{1} = \{A \in 2^{\omega}: A \notin \Delta(D_{\omega}(\Sigma_{2}^{0})), A \text{ Borel, rigid, dense in itself}\}.$ We first show that $A_{0} = \emptyset$; recall the definitions of $P_{n}^{(i)}, X_{n}^{(i)}$ from section 3.4, and of $P_{\omega}, X_{\omega}^{2}$ from section 3.5. Assume that for each $j \in \{1,2\}, m < \omega$, no element of A_{0} is $P_{m}^{(j)}$. If $A \in A_{0}$, then each non-empty clopen subset of A is in A_{0} , so A is nowhere $P_{m}^{(j)}$ for all $j \in \{1,2\}, m < \omega$; on the other hand, $A \in \Delta(D_{\omega}(\Sigma_{2}^{0}))$, so A is P_{ω} . Thus $A \in X_{\omega}$, whence A is homogeneous, a contradiction. So now let $i \in \{1,2\}, n \ge -1$ be such that some $X \in A_{0}$ is $P_{n}^{(i)}$, but no $A \in A_{0}$ is $P_{m}^{(j)} < P_{n}^{(i)}$, in the ordering of theorem 3.4.24; then X is nowhere $P_{m}^{(j)}$ for each $P_{m}^{(j)} < P_{n}^{(i)}$. If $n \ge 0$, then by theorem 3.4.24, $X \in X_{n}^{(i)}$, so X is homogeneous by theorem 3.4.13, a contradiction, so n = -1, i.e. X is σ -compact. By the argument in the introduction to this chapter, X is nowhere compact; if X is nowhere countable, then $X \approx Q \times C$ by theorem 2.4.5, so again X is homogeneous, a contradiction. The last possibility is that X contains a non-empty clopen subset which is not rigid either, and we are done. We now show that $A_{1} = \emptyset$. If not, then there is a minimal pair $\{\Gamma_{n}, \tilde{\Gamma}_{n}\}$ of

We now show that $A_1 = \emptyset$. If not, then there is a minimal pair $\{\Gamma_u, \Gamma_u\}$ of described Wadge classes such that $X \in \Gamma_u \cup \Gamma_u$ for some $X \in A_1$. Since $A_0 = \emptyset$, $V \in A_1$ for each non-empty clopen V in X, and thus $V \notin \Gamma_v \cup \Gamma_v$ for each $\Gamma_v < \Gamma_u$; because $X \notin \Delta(D_\omega(\Sigma_2^0))$, we have $\Delta(D_\omega(\Sigma_2^0)) \subset \Gamma_u$, so we can apply lemma 4.4.1 and obtain that $u(0) \ge 2$.

Suppose that $X \in \Gamma_u$ (the case $X \in \stackrel{Y}{\Gamma}_u$ is similar). If V is a non-empty clopen subset of X, then $V \notin \stackrel{Y}{\Gamma}_u$ since by theorem 4.2.6, $\Delta(\Gamma_u)$ is the union of described Wadge classes smaller than Γ_u ; thus, X is P_u , and nowhere $\stackrel{Y}{P}_u$ (definition 4.3.1 and the remark following it). Now if X is Baire, put Y = X; then $Y \in \stackrel{Y^1}{U}$. Otherwise, there are dense open subsets U_n of X, for $n < \omega$, such that $\bigcap_{n=0}^{\infty} U_n$ is not dense, i.e. $Y \subset \bigcup_{n=0}^{\infty} (X \setminus U_n)$ for some non-empty clopen subset Y of X; then $Y = \bigcup_{n=0}^{\infty} (Y \setminus U_n)$ is first category, and also Y is P_u , nowhere $\stackrel{Y}{P}_u$, so $Y \in \stackrel{Y^0}{U}_u$. In both cases, Y is homogeneous by theorem 4.3.8, a contradiction.

Remark: In van Engelen, Miller, and Steel [13], using a different method, a generalization of the above theorem will be proved. This generalization has as a consequence that if all analytic games (i.e. games $G_X(A)$ with A analytic) are determined, then there are no rigid zero-dimensional analytic spaces; determinacy of analytic sets can be proved granting MC (the hypothesis that there exists at least one measurable cardinal). On the other hand, under the axiom of constructibility V = L, an analytic subset of the Cantor set will be constructed such that both it and its complement are rigid.

APPENDIX: THE COUNTABLE INFINITE PRODUCT OF RATIONALS

As we noted in section 4.5, the case Z_u^0 , $u = 3^1 1^0$, of theorem 4.3.8 yields that the countable infinite product of rationals Q^{ω} is the only zero-dimensional absolute $F_{\sigma\delta}$ which is first category and nowhere an absolute $G_{\delta\sigma}$. In this appendix, we will give an elementary proof of this characterization. Our first step is to provide a characterization of Q^{ω} which is inspired by Sierpiński's internal topological characterization of the absolute $F_{\sigma\delta}$ spaces ([51]). Then we use techniques similar to those of section 3.3 to get the intended result. We will also show how to deduce the characterizations of the three other zero-dimensional homogeneous Borel sets of exact class 2 from that of Q^{ω} in an elementary way.

For obvious reasons, we will use the term " $\sigma\text{-complete space"}$ for "absolute $G_{\delta\sigma}$ space".

A.1 The first characterization of ${oldsymbol Q}^\omega$

We start with an "estimated homeomorphism extension theorem" for the Cantor set, which is essentially due to van Mill [39]; we include a proof.

A.1.1 THEOREM: Let $g: C \rightarrow C$ be a homeomorphism, let $\varepsilon > 0$, and let A be closed and nowhere dense in C. If $h_0: A \rightarrow g[A]$ is a homeomorphism such that $d(g|A,h_0) < \varepsilon$, then there exists an autohomeomorphism h of C such that $h|A = h_0$, and $d(g,h) < \varepsilon$.

Proof: Put $\delta = \varepsilon - d(g|A,h_0)$, and let $\eta > 0$ be such that

(1) for all $x, y \in C$, if d(x, y) < n, then $d(g(x), g(y)) < \frac{1}{2}\delta$, and (2) for all $x, y \in A$, if d(x, y) < n, then $d(h_0(x), h_0(y)) < \frac{1}{2}\delta$.

Let $\{U_i: i = 0, ..., n\}$ be a disjoint family of clopen subsets of C of

diameter less than n, with $A \subset \bigcup_{i=0}^{n} \bigcup_{i}$, and $\bigcup_{i} \cap A \neq \emptyset$ for each i. By (2), we can find a family $\{V_i: i = 0, ..., n\}$ of pairwise disjoint clopen subsets of C of diameter less than $\frac{1}{2}\delta$ such that $V_i \cap g[A] = h_0[\bigcup_i \cap A]$. Since $g[A] \subset \bigcup_{i=0}^{n} V_i$, we can find a clopen subset U of C such that $A \subset U \subset \bigcup_{i=0}^{n} \bigcup_{i}$, and $g[U] \subset \bigcup_{i=0}^{n} V_i$. Now put

 $W_i = U_i \cap U, Z_i = V_i \cap g[U].$

Then $\bigcup_{i=0}^{n} W_{i} = U$, $\bigcup_{i=0}^{n} Z_{i} = g[U]$, $W_{i} \cap A = U_{i} \cap A$, and $Z_{i} \cap g[A] = V_{i} \cap g[A]$. For each i, define $h_{i} \colon W_{i} \to Z_{i}$ to be a homeomorphism extending $h_{0} \mid (W_{i} \cap A)$ (use corollary 3.2.3), and put

 $\mathbf{h} = \bigcup_{i=0}^{n} \mathbf{h}_{i} \cup \mathbf{g} | (C \setminus U).$

Then $d(g(x),h(x)) = 0 < \varepsilon$ if $x \notin U$. If $x \in W_i$, then we can find a point $a \in W_i \cap A$. Since $diam(W_i) < n$, we have $d(g(x),g(a)) < \frac{1}{2}\delta$ by (1), and since $\{h(x),h(a)\} \subset Z_i$, also $d(h(x),h(a)) < \frac{1}{2}\delta$. Hence, $d(g(x),h(x)) \leq d(g(x),g(a)) + d(g(a),h(a)) + d(h(a),h(x)) < \frac{1}{2}\delta + d(g|A,h_0) + \frac{1}{2}\delta = \varepsilon$.

Recall the definition of M from 1.4.

A.1.2 DEFINITION: X is the class of all non-empty zero-dimensional spaces X for which there exist closed subspaces X_{s} , for each $s \in M$, satisfying

- (i) $X = X_{q}$, and $X_{s} = \bigcup_{i=0}^{\infty} X_{si}$ for each $s \in M$;
- (ii) for each $i < \omega$, and each $s \in M$, $X_{s^{i}}$ is nowhere dense in X_{s} ; (iii) if $\sigma \in \omega^{\omega}$, and $p_{k} \in X_{\sigma|k}$ for each $k < \omega$, then the sequence $(p_{k})_{k < \omega}$ converges in X.

A.1.3 LEMMA: $Q^{\omega} \in X$.

Proof: Enumerate \boldsymbol{Q} as $\{q_n: n < \omega\}$, and put $X_{\boldsymbol{Q}} = \boldsymbol{Q}^{\omega}$, $X_{i_0} \dots i_k = (q_{i_0}, \dots, q_{i_k}) \times \boldsymbol{Q} \times \boldsymbol{Q} \times \dots$

Our aim is to prove that, up to homeomorphism, Q^{ω} is the only element of X. We first show that the sets X_s of definition A.1.2 can be chosen to be disjoint. In fact, we have:

A.1.4 LEMMA: Let $X \in X$ be embedded in C. Then the sets X_s from definition A.1.2 can be chosen to satisfy the additional property

(iv)
$$X_s \cap X_t = \emptyset$$
 if $s, t \in M$, $|s| = |t|$, $s \neq t$.

Proof: For each $s \in M \setminus \{\emptyset\}$, let U_s be a disjoint clopen covering of $\overline{X}_s \setminus U_i < f(s) \xrightarrow{X}_{s^{i}}$; enumerate U_s as $\{U_j: j \in E_s\}$, using pairwise disjoint indexing sets. If $j_{n-1} \in E_{s|n}$ for each $1 \le n \le |s| = k$, then define a closed set $X(s, j_0 \dots j_{k-1}) = X_s \cap \bigcap_{n=0}^{k-1} U_{j_n}$. The reader can easily verify that if |s| = k, then

- (1) $\overline{X}(s, j_0 \dots j_{k-1}) \cap \overline{X}(t, \ell_0 \dots \ell_{k-1}) = \emptyset$ if $(s, j_0 \dots j_{k-1}) \neq (t, \ell_0 \dots \ell_{k-1})$ and |t| = |s|; (2) $X = \bigcup \{X(s, i): |s| = 1, i \in E\}$ and X(s, i) is prover d
- (2) $X = U{X(s,j): |s| = 1, j \in E_s}$, and X(s,j) is nowhere dense in X;
- (3) $X(s,j_0...j_{k-1}) = U\{X(t,j_0...j_{k-1}j): t = s, j \in E_t\}$, and $X(t, j_0...j_{k-1}j)$ is nowhere dense in $X(s,j_0...j_{k-1})$.

Furthermore, if $\sigma \in \omega^{\omega}$, $j_{k-1} \in E_{\sigma|k}$ for each $k \in N$, $p_k \in X(\sigma|k, j_0 \dots j_{k-1})$ for each $k \in N$, and $p_0 \in X$, then $p_k \in X_{\sigma|k}$ for each $k < \omega$, and hence the sequence $(p_k)_{k < \omega}$ converges to a point of X. It is now obvious that we can reindex the sets $X(s, j_0 \dots j_{k-1})$ to obtain the required representation of X.

A.1.5 LEMMA: Let $X, Y \in X$ be densely embedded in C. Then there exists a homeomorphism h: $C \rightarrow C$ such that h[X] = Y.

Proof: Since X,Y \in X, there exist closed non-empty subsets X_S of X, Y_S of Y, for each s \in M, staisfying properties (i), (ii), and (iii) of definition A.1.2, and (iv) of lemma A.1.4. Throughout this proof, (i), (ii), (iii), and (iv) will always refer to those properties. We will construct, for each n \in N, a homeomorphism h_n: $C \rightarrow C$, such that

 $\begin{array}{l} (\star) \ d(h_n, h_{n+1}) < \varepsilon_n = \min\{2^{-n}, 3^{-n}, \min\{\min\{d(h_1(x), h_1(y)): d(x, y) \\ \geq 1/n\}: 1 \leq i \leq n\}\}; \\ (\star\star) \ \forall s \ \epsilon \ M: \ \forall x \ \epsilon \ X_s: \ \exists t \ \epsilon \ M: \ |t| = |s|, \ \text{and} \ \forall n \geq \nu(s): \ h_n(x) \ \epsilon \ \overline{Y}_t; \\ (\star\star\star) \ \forall s \ \epsilon \ M: \ \forall y \ \epsilon \ Y_s: \ \exists t \ \epsilon \ M: \ |t| = |s|, \ \text{and} \ \forall n \geq \nu(s): \ h_n^{-1}(y) \ \epsilon \ \overline{X}_t. \end{array}$

Suppose this has been done; then $\lim_{n\to\infty} h_n$ is an autohomeomorphism of C by (*) and lemma 3.2.5. We claim that h[X] = Y. Indeed, let $x \in X$, say $x \in \bigcap_{S < \sigma} X_S$. By (**), for each $k < \omega$ we can find $t(k) \in M$ such that |t(k)| = k, and such that $h_n(x) \in \overline{Y}_{t(k)}$ for each $n \ge \nu(\sigma|k)$; thus, if $k < \ell$, then $h_{\nu(\sigma|\ell)}(x) \in \overline{Y}_{t(k)} \cap \overline{Y}_{t(\ell)} \subset \overline{Y}_{t(k)} \cap \overline{Y}_{t(\ell)|k}$, and hence $t(\ell)|k = t(k)$ by (iv). So in fact, we can find $\tau \in \omega^{\omega}$ such that $\tau|k = t(k)$. Then $h_{\nu(\sigma|k)}(x) \in \overline{Y}_{\tau|k}$; and if $p_k \in Y_{\tau|k}$ satisfies $d(h_{\nu(\sigma|k)}(x), p_k) < 1/(k+1)$, then $(p_k)_{k<\omega}$ converges in Y by (iii), and hence $h(x) = \lim_{k\to\infty} h_{\nu(\sigma|k)}(x)$ = $\lim_{k\to\infty} p_k \in Y$. A similar argument shows that (***) implies $h^{-1}[Y] \subset X$.

Thus, roughly speaking, if $x \in X_s$, and v(s) = n, then h_n determines the Y_t with |t| = |s| which, at the end of the process, will contain h(x); and if $y \in Y_s$, and v(s) = n, then h_n^{-1} determines the X_t with |t| = |s| which, at the end of the process, will contain $h^{-1}(x)$. We will construct the homeomorphisms h_n inductively, together with finite collections $A_s = \{A_{\alpha}: \alpha \in E_s\}, B_s = \{B_{\alpha}: \alpha \in E_s\}$ (using pairwise disjoint indexing sets), each consisting of pairwise disjoint Cantor sets in C, such

- that the following hold for each $n < \omega$, and each $s \in M$ with v(s) = n:
 - (1) if $n \ge 2$, then $d(h_n, h_{n-1}) < \varepsilon_{n-1}$;
 - (2) if $v(t) \leq n$, and $\alpha \in E_t$, then $h_n[A_\alpha] = B_\alpha$;
 - (3) if |t| = |s|, $t \neq s$, and $v(t) \leq n$, then $UA_s \cap UA_t = \emptyset = UB_s \cap UB_t$;

 - (5) if $\alpha \in E_s$, then there exist $t_0, t_1 \in M$ with $|t_0| = |s| = |t_1|$, such that A_α is a clopen subset of \overline{X}_{t_0} , and B_α is a clopen subset of \overline{Y}_{t_1} ;
 - (6) if $\alpha \in E_{\hat{s}}$, $|t| = |\hat{s}|$, and $A_{\alpha} \subset \overline{X}_{t}$, then $A_{\alpha} \cap \overline{X}_{t}\hat{r}_{f}(s) \subset U\{UA_{\hat{s}\hat{j}}; j \leq f(s)\};$ if $\alpha \in E_{\hat{s}}$, $|t| = |\hat{s}|$, and $B_{\alpha} \subset \overline{Y}_{t}$, then $B_{\alpha} \cap \overline{Y}_{t}\hat{r}_{f}(s) \subset U\{UB_{\hat{s}\hat{r}_{i}}; j \leq f(s)\}$
 - (7) if $\alpha \in E_s$, then for some $\beta \in E_{\hat{s}}$, A_{α} is a nowhere dense subset of A_{β} , and B_{α} is a nowhere dense subset of B_{β} .

First note that from (ii) it follows that no X_s or Y_s can contain isolated points (in the relative topology), so that $\overline{A} \approx C$ for any non-empty clopen subset A of X_s or Y_s .

Put $A_{\emptyset} = \{C\} = B_{\emptyset}$. Since $\overline{X}_{0}, \overline{Y}_{0}$ are nowhere dense in C, we can define a homeomorphism $h_{1}: C \to C$ such that $h_{1}[\overline{X}_{0}] = \overline{Y}_{0}$; if we put $A_{0} = \{\overline{X}_{0}\}, B_{0} = \{\overline{Y}_{0}\}$, then (1) - (7) are satisfied ((5) is satisfied for $s = \emptyset$ since X, Y are dense in C). So suppose that h_{m}, A_{s} , and B_{s} , satisfying (1) - (7), have been constructed for $m \le n, v(s) \le n (\ge 1)$.

Fix $s \in M$ with v(s) = n+1, and fix $\alpha \in E_{\widehat{S}}$. By (5), there exist $t_0, t_1 \in M$ with $|t_0| = |\widehat{s}| = |t_1|$, such that $A_\alpha \subset \overline{X}_{t_0}$, $B_\alpha \subset \overline{Y}_{t_1}$. Put $s_0 = t_0^{-}f(s)$, $s_1 = t_1^{-}f(s)$. By (5), $\overline{Y}_{s_1} \setminus U\{UB_{\widehat{S}^{-}i}: i < f(s)\}$ is closed in \overline{Y}_{s_1} , so we can find a clopen V' in B_α , satisfying

$$B_{\alpha} \cap U\{UB_{\hat{s}_{1}}: i < f(s)\} \subset V' \subset B_{\alpha} \setminus (\overline{Y}_{s_{1}} \setminus U\{UB_{\hat{s}_{1}}: i < f(s)\})$$

Since $v(\hat{s}) \leq n$, $h_n[A_\alpha] = B_\alpha$ by (2), and also by (2), $h_n[U[UA_{\hat{s}\hat{i}}: i < f(s)]]$ = $U\{UB_{s^{i}}: i < f(s)\}$ since $v(\hat{s}) \leq n$ for each i < f(s). Thus,

$$h_n[A_{\alpha} \cap U\{UA_{\hat{s}^i}: i < f(s)\}] = B_{\alpha} \cap U\{UB_{\hat{s}^i}: i < f(s)\},$$

and since $\overline{X}_{s_0} \setminus U\{UA_{s_1}: i < f(s)\}$ is closed in \overline{X}_{s_0} by (5), we can find a clopen $U_{s,\alpha}$ in A_{α} such that

$$A_{\alpha} \cap U\{UA_{\widehat{s}\widehat{i}}: i < f(s)\} \subset U_{s,\alpha} \subset A_{\alpha} \setminus (\overline{X}_{s_0} \setminus U\{UA_{\widehat{s}\widehat{i}}: i < f(s)\}),$$

while moreover

$$h_n[U_{s,\alpha}] = V_{s,\alpha} \subset V'.$$

Since $A_{\hat{s}}$ is disjoint, $A_{\alpha} \cap U\{UA_{\hat{s}\hat{s}\hat{i}}: i < f(s)\}$ is nowhere dense in A_{α} by (7), so we may assume that $A_{\alpha} \setminus U_{s,\alpha} \neq \emptyset$, and hence $B_{\alpha} \setminus V_{s,\alpha} \neq \emptyset$. Let $V_{s,\alpha}$ be a clopen disjoint covering of $B_{\alpha} \setminus V_{s,\alpha}$ by non-empty sets of diameter less than ε_n . For each $W \in V_{s,\alpha}$, put

$$\mathbf{p}_{W} = \min\{\mathbf{p}: \mathbf{h}_{n}^{-1}[W] \cap \overline{X}_{t_{0}} \neq \emptyset\}, \ \mathbf{q}_{W} = \min\{\mathbf{q}: W \cap \overline{Y}_{t_{1}} \neq \emptyset\},$$

and

S

ν ŝ S

$$A(W,s,\alpha) = h_n^{-1}[W] \cap \overline{X}_{t_0^{p_W}}, B(W,s,\alpha) = W \cap \overline{Y}_{t_1^{q_W}}$$

Note that p_W (resp. q_W) is well-defined since A_α (resp. B_α) is clopen in \overline{X}_{t_0} (resp. \overline{Y}_{t_1}), and $\bigcup_{p < \omega} \overline{X}_{t_0} \widehat{p}$ (resp. $\bigcup_{q < \omega} \overline{Y}_{t_1} \widehat{q}$) is dense in \overline{X}_{t_0} (resp. \overline{Y}_{t_1}). Now define

$$A_{s} = \{A(W,s,\alpha): W \in V_{s,\alpha}, \alpha \in E_{\widehat{S}}\}, B_{s} = \{B(W,s,\alpha): W \in V_{s,\alpha}, \alpha \in E_{\widehat{S}}\}, and put A_{s} = \{A_{\beta}: \beta \in E_{s}\}, B_{s} = \{B_{\beta}: \beta \in E_{s}\}, such that if A_{\beta} = A(W,s,\alpha), then B_{\beta} = B(W,s,\alpha).$$
 Before we define h_{n+1} , we will show that (3) - (7) are satisfied for each $s \in M$ with $v(s) = n+1$. Fix $s \in M$ with $v(s) = n+1$. To prove (3), let $|t| = |s|, t \neq s$, and $v(t) \leq n+1$. If $\widehat{s} \neq \widehat{t}$, then since $v(\widehat{s}), v(\widehat{t}) \leq n$, we have $UA_{\widehat{s}} \cap UA_{\widehat{t}} = \emptyset$, and hence by (7), $UA_{s} \cap UA_{t} = \emptyset$; if $\widehat{s} = \widehat{t}$, then $UA_{t} \subset U\{UA_{\widehat{s}}: i < f(s)\} \subset C \setminus UA_{s}$ by the construction of A_{s} . Similarly, $UB_{s} \cap UB_{t} = \emptyset$.
For (4), fix $x \in \overline{X}_{s}$; note that $x \in \overline{X}_{\widehat{s}}$. We must show that $x \in U\{UA_{r}: |r| = V|A_{s} \in V|A_{r}: |r| = V|A_{s} \in V|A_{s} \in V|A_{s} = V|A_{s} =$

|s|, v(r) < n+1 U UA_s. First note that if $x \notin U\{UA_r: |r| = |s|, v(r) < n+1\}$, then $x \notin U\{UA_t: |t| = |\hat{s}|, v(t) < v(\hat{s})\}$. Indeed, if $x \in A$ for some $A \in A$ A_t for some $t \in M$ with $|t| = |\hat{s}|, v(t) < v(\hat{s}), \text{ then by (5), } A \subset \overline{X}_{t_0}$ for some $t_0 \in M$ with $|t_0| = |t|$, and since $|\hat{s}| = |t|$ and $A \cap \overline{X}_{\hat{s}} \neq \emptyset$, we must have $t_0 = \hat{s}$ by (iv). Put $q = t^f(s)$. Since v(q) = v(t)+f(s)+1 < v(q) $v(\hat{s})+f(s)+1 = v(s) = n+1$, we can apply (6), and obtain that $x \in A \cap \overline{X}_s =$ $A \cap \overline{X}_{s^{f}(q)} \subset U\{UA_{q^{i}}: j \leq f(q)\} \subset U\{UA_{r}: |r| = |s|, v(r) < n+1\}, a \text{ contra-}$

diction. So if $x \notin U\{UA_r: |r| = |s|, v(r) < n+1\}$, then by (4) applied to \hat{s} , we find that $x \in A_{\alpha}$ for some $\alpha \in E_{\hat{s}}$. Using notation as in the construction of A_s , we find that $t_0 = \hat{s}$, and hence $t_0 \cap f(s) = s_0 = s$. Since $U\{UA_r: |r| = |s|, v(r) < n+1\} \supset U\{UA_{\hat{s}\cap i}: i < f(s)\}$, we have $x \in A_{\alpha} \cap (\overline{X}_{s_0} \setminus U\{UA_{\hat{s}\cap i}: i < f(s)\})$, whence $x \in h_n^{-1}[W]$ for some $W \in V_{s,\alpha}$. We claim that $x \in A(W, s, \alpha)$; since $x \in h_n^{-1}[W] \cap \overline{X}_{t_0} \cap f(s)$, it suffices to show that $h_n^{-1}[W] \cap \overline{X}_{t_0} \cap p = \emptyset$ if p < f(s). So take p < f(s); then $v(t_0 \cap p) \le n$, so by (4) and (7), $\overline{X}_{t_0} \cap p \subset U\{UA_t: |t| = |t_0 \cap p|, v(t) < n\} \cup UA_{t_0} \cap p \subset U\{UA_t: |t| = |\hat{s}|, t \neq \hat{s}\} \cup U\{UA_{\hat{s}\cap i}: i < f(s)\}$. Now $UA_t \cap A_\alpha = \emptyset$ if $|t| = |\hat{s}|, t \neq \hat{s}$, by (3); and $h_n^{-1}[W] \subset A_\alpha \setminus U\{UA_{\hat{s}\cap i}: i < f(s)\}$. This proves the claim. The proof that $\overline{Y}_s \subset U\{UB_r: |r| = |s|, v(r) < n+1\} \cup UB_s$ is similar, so (4) holds. (5) is trivial, and so is (7). It remains so check (6); we will only do so for the first part.

Let $\alpha \in E_{\hat{s}}$, and suppose that $t \in M$ is such that $|t| = |\hat{s}|$, and $A_{\alpha} \subset \overline{X}_{t}$. Then $A_{\alpha} \cap \bigcup \{\bigcup A_{\hat{s} \uparrow j} : j < f(s)\} \subset \bigcup_{s,\alpha} = A_{\alpha} \setminus \bigcup \{h_{n}^{-1}[\mathbb{W}] : \mathbb{W} \in V_{s,\alpha}\} \subset A_{\alpha} \setminus (\overline{X}_{t\uparrow f(s)} \setminus \bigcup \{\bigcup A_{\hat{s} \uparrow j} : j < f(s)\}$, where $\bigcup_{s,\alpha}$ and $V_{s,\alpha}$ are as in the construction of A_{s} . If p < f(s), then $\cup (\hat{s} \uparrow p) \leq n$, so by (6), $A_{\alpha} \cap \overline{X}_{t\uparrow p} \subset \bigcup \{\bigcup A_{\hat{s} \uparrow j} : j < f(s)\}$, whence $h_{n}^{-1}[\mathbb{W}] \cap \overline{X}_{t\uparrow p} = h_{n}^{-1}[\mathbb{W}] \cap A_{\alpha} \cap \overline{X}_{t\uparrow p} \subset h_{n}^{-1}[\mathbb{W}] \cap A_{\alpha} \cap (\overline{X}_{t\uparrow p} \subset h_{n}^{-1}[\mathbb{W}] \cap A_{\alpha} \cap (\mathbb{W}, s, \alpha) = h_{n}^{-1}[\mathbb{W}] \cap \overline{X}_{t\uparrow f(s)} \neq \emptyset$, then $A(\mathbb{W}, s, \alpha) = h_{n}^{-1}[\mathbb{W}] \cap \overline{X}_{t\uparrow f(s)}$. Now let $x \in A_{\alpha} \cap (\overline{X}_{t\uparrow f(s)} \setminus \bigcup \{\bigcup A_{\hat{s} \uparrow j} : j < f(s)\})$; then $x \in h_{n}^{-1}[\mathbb{W}] \cap \overline{X}_{t\uparrow f(s)}$ for some $\mathbb{W} \in V_{s,\alpha}$, and hence $x \in A(\mathbb{W}, s, \alpha) \in A_{s}$. This completes the proof of (6).

We will now define h_{n+1} satisfying (1) and (2).

Since $A(W,s,\alpha) \approx B(W,s,\alpha) \approx h_n^{-1}[W] \approx W \approx C$ for each $W \in V_{s,\alpha}$, each $s \in M$ with v(s) = n+1, and each $\alpha \in E_s$, and since $A(W,s,\alpha)$ (resp. $B(W,s,\alpha)$) is closed and nowhere dense in $h_n^{-1}[W]$ (resp. W), there exist homeomorphisms $g(W,s,\alpha):h_n^{-1}[W] \rightarrow W$ such that $g(W,s,\alpha)[A(W,s,\alpha)] = B(W,s,\alpha)$. Since $V_{s,\alpha}$ is a disjoint clopen covering of $B_{\alpha} \setminus V_{s,\alpha}$, we can define a homeomorphism $g_{s,\alpha}:A_{\alpha} \setminus U_{s,\alpha} \rightarrow B_{\alpha} \setminus V_{s,\alpha}$ by

$$g_{s,\alpha} = \bigcup \{ g(W,s,\alpha) : W \in V_{s,\alpha} \}.$$

Note that $d(g_{s,\alpha},h_n|(A_{\alpha} \setminus U_{s,\alpha})) < \epsilon_n$ since $diam(W) < \epsilon_n$ for each $W \in V_{s,\alpha}$. Now put $I_j = \{s \in M: v(s) = n+1, f(s) = j\}$, for each $j \le n$. Using induction on j, we will define for each $s \in I_j$, and each $\alpha \in E_{\widehat{s}}$, a homeomorphism $h_{s,\alpha}: A_{\alpha} \rightarrow B_{\alpha}$, such that

> (I) $h_{s,\alpha} | (A_{\alpha} \setminus U_{s,\alpha}) = g_{s,\alpha};$ (II) if $\ell \leq j$, $t \in I_{\ell}$, $\beta \in E_{f}$, and $A_{\beta} \subset A_{\alpha}$, then $h_{s,\alpha} | A_{\beta} = h_{t,\beta};$ (III) $d(h_{s,\alpha}, h_{\alpha} | A_{\alpha}) < \varepsilon_{\alpha}.$

Suppose that the $h_{s,\alpha}$ can be constructed. Let $s_0 \in M$ be the sequence (n+1), and let α_0 be the unique element of $E_{\hat{s}} = E_{\emptyset}$; then $A_{\alpha_0} = C = B_{\alpha_0}$, so h_{s_0,α_0} is an autohomeomorphism of C. We claim that $h_{n+1} = h_{s_0,\alpha_0}$ is as required. Indeed, by (III), h_{n+1} clearly satisfies (1). To prove (2), let $t \in M$ with $v(t) \leq n+1$, and let $\gamma \in E_t$. If v(t) = n+1, then $A_{\gamma} = A(W,t,\beta)$ for some $\beta \in E_t$, and some $W \in V_{t,\beta}$. Hence $A_{\gamma} \subset A_{\beta} \setminus U_{t,\beta} \subset A_{\beta}$, so applying (II) (for $\ell = f(t)$, j = n, $\alpha = \alpha_0$, and $s = s_0$), we find that $h_{n+1}[A_{\gamma}] = (h_{s_0,\alpha_0}|A_{\beta})[A_{\gamma}] = h_{t,\beta}[A_{\gamma}]$, and by (I), $h_{t,\beta}[A_{\gamma}] = g_{t,\beta}[A_{\gamma}] = B(W,t,\beta) = B_{\gamma}$. If $v(t) \leq n$, then $t = \hat{s}$ for some $s \in M$ with v(s) = n+1; hence by (II) (for $\ell = f(s)$, j = n, t = s, $\beta = \gamma$, $\alpha = \alpha_0$, $s = s_0$), we find that $h_{n+1}[A_{\gamma}] = h_{s,\gamma}[A_{\gamma}] = B_{\gamma}$.

The homeomorphisms $h_{s,\alpha}$ are constructed as follows.

For $s \in I_0$, $\alpha \in E_{\hat{s}}$, define h_{s, \alpha} by

$$\begin{split} & h_{s,\alpha} | U_{s,\alpha} = h_n | U_{s,\alpha}; \\ & h_{s,\alpha} | (A_{\alpha} \setminus U_{s,\alpha}) = g_{s,\alpha}. \end{split}$$

Since $g_{s,\alpha}[A_{\alpha} \setminus U_{s,\alpha}] = h_n[A_{\alpha} \setminus U_{s,\alpha}]$, and $h_n[A_{\alpha}] = B_{\alpha}$, $h_{s,\alpha}$ maps A_{α} onto B_{α} , and $h_{s,\alpha}$ is a homeomorphism since $U_{s,\alpha}$ is clopen in A_{α} . Clearly, (I) and (III) are satisfied. For (II), note that from (3) and (7) it follows that $A_{\beta} \subset A_{\alpha}$ for some $\beta \in E_{\mathfrak{k}}$, $t \in I_0$, can only occur if $\mathfrak{s} \leq \mathfrak{k}$; since $\nu(\mathfrak{s}) = \nu(\mathfrak{t})$ and $\mathfrak{f}(\mathfrak{s}) = \mathfrak{f}(\mathfrak{t})$, we have $\nu(\mathfrak{s}) = \nu(\mathfrak{t})$, and hence $\mathfrak{s} = \mathfrak{k}$, so $\mathfrak{s} = \mathfrak{t}$. Then $\alpha = \beta$, and we are done.

Now suppose that $h_{t,\beta}$ has been defined for $t \in \bigcup_{\ell=0}^{j} I_{\ell}$, $\beta \in E_{t}$, such that (I), (II), and (III) are satisfied, and fix $s \in I_{j+1}$, $\alpha \in E_{s}$. For $\ell < f(s) = j+1$, put $s_{\ell} = \hat{s} \cdot \ell \cdot j - \ell$. Then $s_{\ell} \in I_{j-\ell}$, and $j-\ell \leq j$, so $h_{s_{\ell},\gamma} \colon A_{\gamma} \to B_{\gamma}$ has been defined for each $\gamma \in E_{s_{\ell}}$. Let $E_{s_{\ell}} = \{\gamma \in E_{s_{\ell}} \colon A_{\gamma} \subset A_{\alpha}\} = \{\gamma \in E_{s_{\ell}} \colon B_{\gamma} \subset B_{\alpha}\}$. Then

$$g_{\ell} = U\{h_{s_{\ell},\gamma}: \gamma \in \widetilde{E}_{\widehat{s}_{\ell}}\}: U\{A_{\gamma}: \gamma \in \widetilde{E}_{\widehat{s}_{\ell}}\} \rightarrow U\{B_{\gamma}: \gamma \in \widetilde{E}_{\widehat{s}_{\ell}}\}$$

is a well-defined homeomorphism since $A_{\widehat{S}\ell}$ and $B_{\widehat{S}\ell}$ consist of pairwise disjoint sets; and

$$g = \bigcup_{\ell=0}^{j} g_{\ell} \colon \bigcup\{A_{\gamma} \colon \gamma \in \widetilde{E}_{\widehat{S}\ell}, \ \ell \leq j\} \rightarrow \bigcup\{B_{\gamma} \colon \gamma \in \widetilde{E}_{\widehat{S}\ell}, \ \ell \leq j\}$$

is a well-defined homeomorphism since by (3), $UA_{\hat{s}\ell} \cap UA_{\hat{s}\ell'} = \emptyset = UB_{\hat{s}\ell} \cap UB_{\hat{s}\ell'}$ if $\ell \neq \ell'$. Let D_0 denote the domain, and D_1 the range of g. Then $D_0 \subset A_{\alpha} \cap U\{UA_{\hat{s}^*i}: i < f(s)\} \subset U_{s,\alpha}, D_1 \subset B_{\alpha} \cap U\{UB_{\hat{s}^*i}: i < f(s)\} \subset V_{s,\alpha}, D_0 \approx C$ (resp. $D_1 \approx C$) is nowhere dense in $U_{s,\alpha} \approx C$ (resp. $V_{s,\alpha} \approx C$) by (7), and $d(g,h_n|D_0) < \epsilon_n$ by (III). So by theorem A.1.1, there exists a homeomorphism $\tilde{g}: U_{s,\alpha} \rightarrow V_{s,\alpha}$ such that $\tilde{g}|D_0 = g$, and $d(\tilde{g},h_n|U_{s,\alpha}) < \epsilon_n$. Define $h_{s,\alpha}$: $\begin{array}{ll} A_{\alpha} \rightarrow B_{\alpha} & by \\ & & h_{s,\alpha} \mid U_{s,\alpha} = \tilde{g}; \\ & & h_{s,\alpha} \mid (A_{\alpha} \setminus U_{s,\alpha}) = g_{s,\alpha}. \end{array}$

Then $h_{s,\alpha}$ satisfies (I) and (III). If $\ell \leq j+1$, $t \in I_{\ell}$, $\beta \in E_{t}$, and $A_{\beta} \subset A_{\alpha}$, then by (3) and (7), $\hat{s} \leq \hat{t}$. If $\hat{s} = \hat{t}$, then s = t, $\alpha = \beta$, and we are done. If $\hat{s} < \hat{t}$, then for some $k \leq j$, we have $\hat{s} < \hat{s}_{k} \leq \hat{t}$. By (7), there exist $\gamma \in E_{s,k}^{\circ}$, $\delta \in E_{s}^{\circ}$, such that $A_{\beta} \subset A_{\gamma} \subset A_{\delta}^{\circ}$. Since A_{s}° consists of pairwise disjoint sets, we have $\delta = \alpha$. Hence, $h_{s,\alpha}|A_{\beta} = (h_{s,\alpha}|A_{\gamma})|A_{\beta} = (\tilde{g}|A_{\gamma})|A_{\beta}$ = $(g_{k}|A_{\gamma})|A_{\beta} = h_{s,k}^{\circ}, \gamma|A_{\beta}^{\circ}$. Since (II) holds for $j = f(s_{k})$, $h_{s,k}^{\circ}, \gamma|A_{\beta} = h_{t,\beta}^{\circ}$, and we are done.

This completes the inductive construction of the homeomorphisms $h_{s,\alpha}$, and hence of the autohomeomorphisms h_n of *C*. To complete the proof of the lemma, we must show that the conditions (*), (**), and (***) at the begin of this proof, follow from (1) - (7). Now (*) is clear from (I), and since (***) is similar to (**), we will only prove (**). So let $s \in M$, and $x \in \overline{X}_s$. By (4), $x \in A_\alpha$ for some $\alpha \in E_t$, for some $t' \in M$ with $|t'| = |s|, v(t') \leq$ v(s). Hence by (2), $h_n(x) \in B_\alpha$ for each $n \geq v(t')$, in particular for each $n \geq v(s)$. By (5), $B_\alpha \subset \overline{Y}_t$ for some $t \in M$ with |t| = |t'|, so |t| = |s|, and $h_n(x) \in \overline{Y}_t$ for each $n \geq v(s)$.

A.1.6 THEOREM: Up to homeomorphism, Q^{ω} is the unique element of X. Proof: $Q^{\omega} \in X$ by lemma A.1.3; and if $X \in X$, then X contains no isolated points, so X can be densely embedded in C. Now apply lemma A.1.5.

A.1.7 COROLLARY: The Cantor set is homogeneous with respect to dense copies of Q^{ω} .

In [34], Luzin "effectively" described an absolute $F_{\sigma\delta}$ which is not an absolute $G_{\delta\sigma}$, viz. the subspace of ω^{ω} consisting of all sequences of elements of ω which converge to infinity. As a corollary to our first characterization, we show that in fact this space is homeomorphic to Q^{ω} .

A.1.8 THEOREM: Let $X = \{(x_i)_{i < \omega} \in \omega^{\omega} : \lim_{i \to \infty} x_i = \infty\}$. Then $X \approx Q^{\omega}$. Proof: Note that X consists of those sequences which, for each $n < \omega$, take the value n at only finitely many coordinates. Let $\{E_i : i < \omega\}$ be an enumeration of the collection of finite subsets of ω .

For s,t $\in M$, if $|s| = |t| \ge 1$, s = (i_0, \dots, i_k) , put

$$X(s,t) = \{\sigma = (x_m)_m \in X: t < \sigma, \text{ and for each } n \le k, x_m = n \text{ if } and only if m \in E_{in} \}$$

Then X(s,t) is closed in X. If we also put $X(\emptyset, \emptyset) = X$, then it is easily seen that, for each $s_0, t_0 \in M$ with $|s_0| = |t_0|$,

$$X(s_0, t_0) = U[X(s, t): s, t \in M, \hat{s} = s_0, \hat{t} = t_0],$$

and that X(s,t) is nowhere dense in $X(s_0,t_0)$ if $\hat{s} = s_0$, $\hat{t} = t_0$. Finally, if $\sigma, \tau \in \omega^{\omega}$, and $p_k = (p_i^k)_{i < \omega} \in X(\sigma|k,\tau|k)$ for each $k < \omega$, then $p_i^k = p_i^{k+1}$ if i < k, so $(p_k)_k$ converges to a point of X.

A.2 The second characterization of ${oldsymbol{\varrho}}^\omega$

Throughout this section, X_0 denotes the class of all zero-dimensional absolute $F_{\sigma\delta}$ spaces which are nowhere σ -complete, and of the first category. Using theorem A.1.6, we will show that, up to homeomorphism, Q^{ω} is the unique element of X_0 , yielding the result we desire.

A.2.1 LEMMA: If X is an analytic space which is not σ -complete, then X contains a closed nowhere σ -complete subspace Y which is nowhere dense in X.

Proof: First note that any non- σ -complete space A contains a nowhere σ complete closed subspace B, viz. B = A\U{U: U is an open σ -complete subset
of A}. So we may assume that X is nowhere σ -complete. If X is Baire,
then by 1.12.2, X contains a dense complete subset G. Since G is an absolute G_{δ} , we can write X\G = $\bigcup_{i=0}^{\infty} F_i$, with F_i closed in X. Then for some
j, F_j is not σ -complete. By the above remark, F_j contains a closed nowhere σ -complete subspace Y; then Y is as required. If X is not Baire, then
there exist a non-empty open set U, and closed nowhere dense sets A_i in
X, such that $U \subset \bigcup_{i=0}^{\infty} A_i$. Since U is an F_{σ} in X, and since U is not σ -complete, U contains a subset F which is not σ -complete, and closed in
X. Then $F = \bigcup_{i=0}^{\infty} (A_i \cap F)$, and hence some $A_j \cap F$ is not σ -complete; again,
if Y is nowhere σ -complete and closed in $A_i \cap F$, then Y is as required. \Box

The following lemma is the key to our second characterization; the proof is similar to that of lemmas 3.3.4 and 3.3.5.

A.2.2 LEMMA: Let A be a Borel set in C which is not σ -complete, and let F be a σ -compact space such that A \subset F \subset C. Then A contains a closed no-

where dense subset Y which is nowhere o-complete and first category, such that $Cl_C(Y) \subset F$.

Proof: We let $\overline{}$ denote closure in C. Since F\A is Borel in C, there exists a continuous surjection $\phi: P \rightarrow F \setminus A$. Let

> W = {x $\in P$: there exists a neighborhood V of x in P, and a σ -compact subset E_x of F, such that $\phi[V_x] \subset E_x$, and $E_x \cap A$ is σ -complete}.

Then W is open in P, so there exist countably many open V_i in P, and $\sigma\text{-compact } \mathbf{E}_{i} \text{ in } \mathbf{F}, \text{ such that } \mathbf{W} = \bigcup_{i=0}^{\infty} \mathbf{V}_{i}, \phi[\mathbf{V}_{i}] \subset \mathbf{E}_{i}, \text{ and } \mathbf{E}_{i} \cap \mathbf{A} \text{ is } \sigma\text{-complete. Suppose that } \mathbf{F} \setminus \mathbf{A} \subset \mathbf{E} = \bigcup_{i=0}^{\infty} \mathbf{E}_{i}; \text{ then } \mathbf{A} = (\mathbf{E} \cap \mathbf{A}) \cup (\mathbf{F} \setminus \mathbf{E}) \text{ is } \sigma\text{-complete, a contradiction. So } \mathbf{G} = \mathbf{P} \setminus \phi^{-1}[\mathbf{E} \setminus \mathbf{A}] \text{ is non-empty, and a } \mathbf{G}_{\delta} \text{ in } \mathbf{G}_{\delta}$ P, whence complete. If U is a non-empty open subset of G, say U = U' \cap G, with U' open in P, then $\phi[U'] = \phi[U] \cup \phi[U' \setminus U] \subset (\overline{\phi[U]} \cap F) \cup E$, which is a σ -compact subset of F. Since U' ¢ W, (($\overline{\phi[U]} \cap F$) $\cup E$) $\cap A$ is not σ complete; but $E \cap A$ is σ -complete, so $\overline{\phi[U]} \cap A$ is not σ -complete. Now write $F = \bigcup_{i=0}^{\infty} F_i$, with F_i compact, and let $\{B_i: i < \omega\}$ be a basis for the topology of A. We will construct compact sets K_s , open subsets U_s of C, open subsets W_s of G, and points $x_i \in B_i$, for each $s \in M$ and each i < ω , such that the following hold:

- (1) $K_s = \overline{K_s \cap A} \subset \overline{\phi[W_s]} \subset U_s;$ (2) for each $n < \omega: \overline{U}_{s^n} \cap K_s = \emptyset;$ (3) for each $n,m < \omega: \overline{U}_{s^n} \cap \overline{U}_{s^m} = \emptyset$ if $n \neq m;$
- (4) for each $n < \omega$: $Cl_G(W_{s^n}) \subset W_s;$
- (5) for each $n < \omega$: $\overline{U}_{sn} \subset U_{s}$; (6) diam(W_{s}) $\leq 2^{-|s|}$ (here, the diameter is taken with respect to a complete metric for G);
- (7) diam(U_{s}) $\leq 2^{-v(s)}$;
- (8) for each $n < \omega$: $d(K_s, K_{s^n}) \le 2^{1-\nu(s^n)}$;
- (9) $K_s \cap A$ is nowhere σ -complete, and nowhere dense in $\overline{\phi[W_{\sigma}]} \cap A$;
- (10) K_s is nowhere dense in $K_s \cup \bigcup_{n=0}^{\infty} K_s n;$
- (11) K is contained in some F_i ;
- (12) $Z_i = U\{K_s: |s| \le i\}$ is compact, and $Z_i \cap A$ is nowhere dense in A;
- (13) for each $j \leq i: x_i \notin Z_i$.

We use induction on |s| and i. First, put $W_{\phi} = G$, $U_{\phi} = C$. Then $\overline{\phi[W_{\phi}]} \cap A$ $= \bigcup_{i=0}^{\infty} (\overline{\phi[W_{0}]} \cap A \cap F_{i}) \text{ is not } \sigma\text{-complete, so some } \overline{\phi[W_{0}]} \cap A \cap F_{i} \text{ is not}$

 σ -complete. By lemma A.2.1, $\overline{\phi[W_d]} \cap A \cap F_i$ contains a nowhere σ -complete, closed nowhere dense subset H_d ; put $K_d = \overline{H}_d$. Since H_d is nowhere dense in A, $B_0 \notin H_0$, say $x_0 \in B_0 \setminus H_0$. Then (1), (9), and (11) - (13) are satisfied, and so are (6) and (7) since all metrics are assumed to be bounded by 1. Next, suppose that K_s , W_s , W_s , and x_i have been defined for $|s| \le k$, $i \le k$, in accordance with conditions (1) - (13). Fix $s \in M$ with |s| = k. From (1) and (9), it is not hard to see that K_s is nowhere dense in $K_s \cup$ $\phi[W_s]$, so by lemma 3.3.3 there exists a countable discrete subset $D_s =$ $\{y_{s^n}: n < \omega\}$ of $\phi[W_s] \setminus K_s$, such that $\overline{D}_s = D_s \cup K_s$, and $d(y_{s^n}, K_s) \le C_s$ $2^{-\nu(s^n)}$ for each $n < \omega$. Now let U_{s^n} be an open neighborhood of y_{s^n} such that $\overline{U}_{s^n} \subset U_s$, $\overline{U}_{s^n} \cap K_s = \emptyset$, $\overline{U}_{s^n} \cap \overline{U}_{s^m} = \emptyset$ if $n \neq m$, and diam $(U_{s^n}) \leq 2^{-\nu(s^n)}$, for each $n,m < \omega$. Since $y_{s^n} \in \phi[W_s]$, $y_{s^n} = \phi(x_{s^n})$ for some $x_{sn} \in W_s$; hence there is an open neighborhood W_{sn} of x_{sn} in G such that $\operatorname{Cl}_{G}(W_{s^{n}}) \subset W_{s}$, diam $(W_{s^{n}}) \leq 2^{-|s|-1}$, and $\overline{\phi[W_{s^{n}}]} \subset U_{s^{n}}$. Then $\overline{\phi[W_{s^n}]} \cap A$ is not σ -complete, so as above, $\overline{\phi[W_{s^n}]} \cap A$ contains a nowhere σ -complete, closed nowhere dense subset H'_{sn} which is contained in some F_i ; let H_{sn} be a non-empty clopen subset of H'_{sn} which is disjoint from $\{x_i: j \le k\}$, and put $K_{s^n} = \overline{H}_{s^n}$. Then (1) - (7), (9), and (11) are satisfied; furthermore, (8), (10), and the first part of (12) are proved as in lemma 3.3.4. To prove the second part of (12), suppose that V is a non-empty open subset of A which is contained in Z_{k+1} . Since $Z_k \cap A$ is closed and nowhere dense in A, $\mathbb{V}\backslash\mathbb{Z}_k^{}$ is a non-empty open subset of A, contained in $U_{|s|=k+1} K_s$. So for some $s \in M$ with |s| = k+1, $(V \setminus Z_k) \cap K_s \neq 0$ \emptyset ; however, by (1), (3), and (5), $(\mathbb{V}\setminus\mathbb{Z}_k) \cap \mathbb{K}_s = (\mathbb{V}\setminus\mathbb{Z}_k) \cap \mathbb{U}_s$, contradicting the fact that $K_s \cap A = H_s$ is nowhere dense in A. Hence (12) holds. In particular, $B_{k+1} \notin Z_{k+1} \cap A$, so we can find a point $x_{k+1} \in B_{k+1} \setminus Z_{k+1}$; then (13) is also satisfied. This completes the induction. Now put $Y = \bigcup_{i=0}^{\infty} (Z_i \cap A)$; we claim that Y is as required. First, that $\overline{Y} \setminus (\bigcup_{i=0}^{\infty} Z_i) \subset \phi[P] = F \setminus A$ is shown exactly as in the proof of lemma 3.3.4; so $\frac{1}{\overline{Y}} \subset \bigcup_{i=0}^{\infty} Z_i \cup (F \setminus A) = \bigcup_{s \in M} K_s \cup (F \setminus A) \subset F$ by (11), and $\overline{Y} \cap A = Y$. By (13), $B_j \notin Y$ for each $j < \omega$, so Y is closed and nowhere dense in A. Now Y is clearly nowhere σ -complete, and each K $_{S}$ \cap A is nowhere dense in Y by (1) and (10), so Y is first category.

A.2.3 LEMMA: Let $X \in X_0$, let F be a σ -compact space such that $X \subset F \subset C$, and let $\varepsilon > 0$. Then there exist closed nowhere dense subsets X_i of X such that

(i)
$$X = \bigcup_{i=0}^{\omega} X_i;$$

(ii) $X_i \in X_0$ for each $i < \omega;$
(iii) $Cl_C(X_i) \subset F;$
(iv) diam(X_i) < ε .

Proof: Again, let denote closure in C. If $F = \bigcup_{i=0}^{\infty} F_i$, with F_i compact, and $X = \bigcup_{i=0}^{\infty} Y_i$, with Y_i closed and nowhere dense in X, then $X = \bigcup_{i=0}^{\infty} Y_i$ $U_{j=0}^{\infty}(Y_i \cap F_j)$, i.e. we can write $X = U_{i=0}^{\infty} A_i$, where A_i is closed and nowhere dense in X, and $\overline{A}_i \subset F$; of course, we may assume that each A_i is non-empty. Fix $i < \omega$, and let \mathcal{D} be a covering of $\overline{X} \setminus \overline{A}_i$ by non-empty disjoint clopen subsets of \overline{X} , such that diam(D) < d(D, \overline{A} ,) for each D $\in \mathcal{D}$. Since $D \cap X$ is non-empty, it is not σ -complete, and since $D \cap X \subset F \subset C$, we can apply lemma A.2.2 to obtain, for each D ϵ D, a closed nowhere dense subset E(D) of D \cap X which is nowhere $\sigma\text{-complete}$ and first category, such that $\overline{E(D)} \subset F$. Put $B_i = A_i \cup U_{D \in \mathcal{D}} E(D)$. Since $\overline{X} \setminus (\overline{A_i} \cup U_{D \in \mathcal{D}} \overline{E(D)}) = U_{D \in \mathcal{D}} (D \setminus \overline{A_i})$ $\overline{E(D)}$) is open in \overline{X} , we have $\overline{B}_i = \overline{A}_i \cup \bigcup_{D \in D} \overline{E(D)} \subset F$, and B_i is closed in X. From the diameter condition on the elements of $\mathcal D$ it follows that A. is nowhere dense in B_i ; thus, since each E(D) is first category, B_i is first category. Also, if U is a non-empty open subset of B_i , then U \cap E(D) $\neq \emptyset$ for some $D \in \mathcal{D}$, so U is not σ -complete, i.e. B_i is nowhere σ complete, whence $B_i \in X_0$. Finally, B_i is nowhere dense in X: if V is non-empty and open in X, and $V \subset B_i$, then $V \cap D = V \cap E(D) \neq \emptyset$ for some D ϵ $\mathcal D$, contradicting the fact that E(D) is nowhere dense in X. Now let $u_i^{}$ be a clopen covering of B, by non-empty sets of diameter less than ϵ , and enumerate $\bigcup_{i=0}^{\infty} U_i$ as $\{X_i: i < \omega\}$; then the sets X_i are as required.

To keep everything elementary, we here also give a new proof of the fact that ${\it Q}^\omega$ is not σ -complete.

A.2.4 LEMMA: $Q^{\omega} \in X_{0}$.

Proof: Being a countable product of σ -compacta, \boldsymbol{Q}^{ω} is an absolute $F_{\sigma\delta}$, and clearly it is first category. Since every non-empty open subset of \boldsymbol{Q}^{ω} contains a closed copy of \boldsymbol{Q}^{ω} , it suffices to prove that \boldsymbol{Q}^{ω} is not σ -complete. Suppose that $\{A_i: i < \omega\}$ is a family of complete subsets of \boldsymbol{Q}^{ω} . Since \boldsymbol{Q}^{ω} is not Baire, A_0 is not dense in \boldsymbol{Q}^{ω} , so there exists a nonempty basic open subset U of \boldsymbol{Q}^{ω} such that $U \cap A_0 = \emptyset$. Let $n_0 \in \mathbb{N}$, $(q_0, \ldots, q_{n_0-1}) \in \boldsymbol{Q}^{n_0}$ be such that $X_0 = (q_0, \ldots, q_{n_0-1}) \times \boldsymbol{Q} \times \boldsymbol{Q} \times \cdots \subset U$. Since $A_1 \cap X_0$ is closed in A_1 , it is complete, and since $X_0 \approx \boldsymbol{Q}^{\omega}$ is not Baire,

as above we can find $n_0 < n_1 < \omega$, $(q_{n_0}, \dots, q_{n_1-1}) \in Q^{n_1-n_0}$, such that $X_1 = (q_0, \dots, q_{n_1-1}) \times Q \times Q \times \dots \subset Q^{\omega} \setminus (A_0 \cup A_1)$. Proceeding in this way, we find a point $(q_i)_{i < \omega} \in Q^{\omega} \setminus (\bigcup_{i=0}^{\infty} A_i)$; so Q^{ω} is not σ -complete.

We are now ready to prove the main theorem of this appendix.

A.2.5 THEOREM: Up to homeomorphism, Q^{ω} is the unique element of X_0 .

Proof: By lemma A.2.4, $Q^{\omega} \in X_0$. So suppose $X \in X_0$; embed X in C, and let $\{F_k : k < \omega\}$ be a family of σ -compact subsets of C such that $X = \bigcap_{k=0}^{\infty} F_k$, and $F_0 = C$. We will construct closed subspaces X_s of X, for each $s \in M$, satisfying conditions (i) and (ii) of definition A.1.2, as well as

(*) for each $s \in M$, $X_s \in X_0$; (**) for each $s \in M$, diam $(X_s) < (|s|+1)^{-1}$; (***) for each $s \in M$, $Cl_C(X_s) \subset F_{|s|}$.

The construction is a triviality: put $X_{\emptyset} = X$, and if X_s has been defined for all $s \in M$ with $|s| \le k$, then we obtain the sets X_{s^i} by applying lemma A.2.3 to $X_s \in F_{|s|+1} \in C$, $\varepsilon = (|s|+2)^{-1}$. We claim that the sets X_s also satisfy condition (iii) of definition A.1.2. Indeed, let $\sigma \in \omega^{\omega}$. Since $\overline{X}_{\sigma|0} \supset \overline{X}_{\sigma|1} \supset \overline{X}_{\sigma|2} \supset \cdots$ is a decreasing sequence of compacta, $\bigcap_{k=0}^{\infty} \overline{X}_{\sigma|k} \neq \emptyset$, say $x \in \bigcap_{k=0}^{\infty} \overline{X}_{\sigma|k}$. By (***), $x \in \bigcap_{k=0}^{\infty} F_k = X$. Thus, $x \in \bigcap_{k=0}^{\infty} X_{\sigma|k}$, and if U is any open neighborhood of x in X, then by (**), $X_{\sigma|n} \in U$ for some $n < \omega$. Hence, if $P_k \in X_{\sigma|k}$ for each $k < \omega$, then $P_k \in U$ for each $k \ge n$, so $(P_k)_k$ converges to x.

From this characterization of Q^{ω} we can, by elementary methods, obtain characterizations of the other homogeneous zero-dimensional absolute Borel sets of exact class two. Let X_1 be the class of all zero-dimensional nowhere σ -complete absolute $F_{\sigma\delta}$ spaces that are Baire, and let W_0 (resp. W_1) denote the class of all zero-dimensional σ -complete spaces that are first category (resp. Baire), and nowhere an absolute $F_{\sigma\delta}$. By 1.12.3, it is clear that if X is dense and co-dense in C, then X ϵX_0 (resp. X_1) if and only if $C \setminus X \epsilon W_1$ (resp. W_0).

A.2.6 THEOREM: Let Q^{ω} be densely embedded in C. Then up to homeomorphism, $C \setminus Q^{\omega}$ is the only element of W_1 ; furthermore, C is homogeneous with respect to dense copies of $C \setminus Q^{\omega}$. Proof: By the above remark, $C \setminus Q^{\omega} \in W_1$; and if A, B $\in W_1$ are densely embedded in *C*, then by the same remark and theorem A.2.5, $C \setminus A \approx Q^{\omega} \approx C \setminus B$, so by corollary A.1.7, there is an autohomeomorphism h of *C* such that $h[C \setminus A] = C \setminus B$, whence h[A] = B.

We will just write $C \setminus Q^{\omega}$ for the unique element of W_1 .

A.2.7 THEOREM:(a) Up to homeomorphism, $Q \times (C \setminus Q^{\omega})$ is the unique element of W_0 . (b) Let $Q \times (C \setminus Q^{\omega})$ be densely embedded in C. Then up to homeomorphism, $C \setminus (Q \times (C \setminus Q^{\omega}))$ is the unique element of X_1 .

Proof:(a) It is clear that $Q \times (C \setminus Q^{\omega}) \in W_0$. So suppose that $X \in W_0$, say $X = \bigcup_{i=0}^{\infty} X_i$, with X_i closed and nowhere dense in X. Fix $i < \omega$, and let D be a covering of $X \setminus X_i$ by non-empty clopen disjoint subsets of X such that diam(D) $< d(D, X_i)$ for each $D \in D$. If we embed D densely in C, then since D is σ -complete and not an absolute $F_{\sigma\delta}$, $C \setminus D$ is an absolute $F_{\sigma\delta}$ which is not σ -complete. By lemma A.2.2, $C \setminus D$ contains a closed nowhere dense subset Y such that $Y \in X_0$, i.e. $Y \approx Q^{\omega}$; then $E(D) = \overline{Y} \setminus Y \approx C \setminus Q^{\omega}$. Note that E(D) is closed and nowhere dense in D. Put $A_i = X_i \cup \bigcup_{D \in D} E(D)$; then A_i is closed and nowhere dense in X, and it is easily seen that $A_i \in W_1$, i.e. $A_i \approx C \setminus Q^{\omega}$. So by theorem 3.2.4, $\bigcup_{i=0}^{\infty} A_i = X \approx Q \times (C \setminus Q^{\omega})$. (b) By theorems A.2.6 and 3.2.8, C is homogeneous with respect to dense copies of $Q \times (C \setminus Q^{\omega})$; now proceed as in the proof of theorem A.2.6.

REFERENCES

[1] P. Alexandroff and P. Urysohn, Über nulldimensionale Punktmengen, Math. Ann. 98 (1928), 89-106.

[2] R.D. Anderson, On topological infinite deficiency, Mich. Math. J. 14 (1967), 365-383.

[3] R. Bennett, Countable dense homogeneous spaces, Fund. Math. 74 (1972), 189-194.

[4] L.E.J. Brouwer, On the structure of perfect sets of points, Proc. Akad. Amsterdam 12 (1910), 785-794.

[5] G. Cantor, Über unendliche, lineare Punktmannichfaltigkeiten, Math. Ann.
21 (1883), 545-591.

[6] M. Davis, Infinite games of perfect information, Ann. Math. Studies 52 (1964), 85-101.

[7] E.K. van Douwen, unpublished.

[8] E.K. van Douwen, Top. Proc. 5 (1980), Problem section.

[9] E.K. van Douwen, *The integers and topology*, Handbook of Set-theoretic topology, North-Holland (1984), 111-167.

[10] F. van Engelen, Homogeneous Borel sets of ambiguous class two, Trans.Am. Math. Soc. 290 (1985), 1-39.

[11] F. van Engelen, Homogeneous Borel sets, Proc. Am. Math. Soc. (to appear).

[12] F. van Engelen, Countable products of zero-dimensional absolute $~F_{\sigma\delta}$

spaces, Indag. Math. 46 (1984), 391-399.

[13] F. van Engelen, A.W. Miller, and J.R. Steel, *Rigid Borel sets and better quasi-order theory*, preprint.

[14] F. van Engelen and J. van Mill, Borel sets in compact spaces: some Hurewicz-type theorems, Fund. Math. 124 (1984), 271-286.

[15] R. Engelking, General Topology, PWN Warszawa (1977).

[16] R. Engelking, W. Holsztyński, and R. Sikorski, Some examples of Borel sets, Coll. Math. 15 (1966), 271-274. [17] R.C. Freiwald, R. McDowell, and E.F. McHugh, Jr., Borel sets of exact class, Coll. Math. 41 (1979), 187-191. [18] D. Gale and F.M. Stewart, Infinite games with perfect information, Ann. Math. Studies 28 (1953), 245-266. [19] J. de Groot and R.J. Wille, Rigid continua and topological group-pictures Archiv der Math. 9 (1958), 441-446. [20] A. Gutek, On extending homeomorphisms on the Cantor set, Topological Structures II, Math. Centre Tracts 115 (1979), 105-116. [21] F. Hausdorff, Grundzilge der Mengenlehre, Leipzig (1914). [22] F. Hausdorff, Die schlichten stetigen Bilder des Nullraums, Fund. Math. 29 (1937), 151-158. [23] W. Hurewicz, Relativ perfekte Teile von Punktmengen und Mengen (A), Fund. Math. 12 (1928), 78-109. [24] H.J.K. Junnila, On strongly zero-dimensional F_{α} -metrizable stratifiable spaces, Topology, Proc. Int. Conf., Leningrad 1982, Lect. Notes in Math. 1060 (1984), 67-75. [25] B. Knaster and M. Reichbach, Notion d'homogénéité et prolongements des homéomorphies, Fund. Math. 40 (1953), 180-193. [26] K. Kunen and A.W. Miller, Borel and projective sets from the point of view of compact sets, Math. Proc. Camb. Phil. Soc. 94 (1983), 399-409. [27] K. Kuratowski, Sur la puissance de l'ensemble des "nombres de dimensions" de M. Fréchet, Fund. Math. 8 (1925), 201-208. [28] K. Kuratowski, Topologie I, Warszawa (1948). [29] M. Lavrentieff, Contribution à la théorie des ensembles homéomorphes, Fund. Math. 6 (1924), 149-160. [30] M. Lavrentieff, Sur les sous-classes de la classification de M. Baire, Comptes Rendus Acad. Sci. Paris 180 (1925), 111-114. [31] S. Levi, On Baire cosmic spaces, Gen. Top. and its relations to Modern Anal. and Alg. V, Prague 1981, Heldermann, Berlin (1983), 450-451. [32] A. Louveau, Some results in the Wadge hierarchy of Borel sets, Cabal Seminar 79-81, Lect. Notes in Math. 1019 (1983), 28-55. [33] N. Luzin, Sur la classification de M. Baire, Comptes Rendus Acad. Sci. Paris 164 (1917), 91-94. [34] N. Luzin, Leçons sur les ensembles analytiques et leurs applications, Gauthier-Villars, Paris (1930).

[35] D.A. Martin, Borel determinacy, Ann. of Math. 102 (1975), 363-371. [36] D.A. Martin and A.S. Kechris, Infinite games and effective descriptive set theory, Analytic Sets, Instr. Conf. on Anal. Sets, Univ. Coll., London 1978, Academic Press, London (1980). [37] E. Michael, Local properties of topological spaces, Duke Math. J. 21 (1954), 163-171. [38] J. van Mill, Characterization of some zero-dimensional separable metric spaces, Trans. Am. Math. Soc. 264 (1981), 205-215. [39] J. van Mill, Characterization of a certain subset of the Cantor set, Fund. Math. 118 (1983), 81-91. [40] J. van Mill, Closed images of topological groups, Coll. Math. Soc. János Bolyai, Topology, Eger (Hungary) 1983 (to appear). [41] J. van Mill, Topological equivalence of certain function spaces (to appear). [42] D. Montgomery, Non-separable metric spaces, Fund. Math. 25 (1935), 527-533. [43] Y. Moschovakis, Descriptive set theory, North-Holland, Amsterdam (1980). [44] A.V. Ostrovskii, Continuous images of the Cantor product $C \times Q$ of a perfect set C and the rational numbers Q (Russian), Sem. on Gen. Top., Moskov. Gos. Univ., Moscow (1981), 78-85. [45] J.B. Paris, $ZF \vdash \Sigma_{L}^{0}$ determinateness, J. Symb. Logic 37 (1972), 661-667. [46] J. Pollard, On extending homeomorphisms on zero-dimensional spaces, Fund. Math. 67 (1970), 39-48. [47] D. Ravdin, On extensions of homeomorphisms to homeomorphisms, Pac. J. Math. 37 (1971), 481-495. [48] J.E. Rubin, Set theory for the Mathematician, Holden Day, San Francisco (1967). [49] J. Saint-Raymond, La structure Borélienne d'Effros est-elle standard?, Fund. Math. 100 (1978), 201-210. [50] W. Sierpiński, Sur une propriété topologique des ensembles dénombrables denses en soi, Fund. Math. 1 (1920), 11-16. [51] W. Sierpiński, Sur une définition topologique des ensembles $F_{\sigma\delta}$, Fund. Math. 6 1924), 24-29. [52] W. Sierpiński, Sur l'existence de diverses classes d'ensembles, Fund. Math. 14 (1929), 82-91. [53] W. Sierpiński, Cardinal and ordinal numbers, PWN Warszawa (1958). [54] R. Sikorski, Some examples of Borel sets, Coll. Math. 5 (1958), 170-171.

[55] J.R. Steel, Determinateness and subsystems of analysis, Thesis, Univ. of Calif. Berkeley (1977).

[56] J.R. Steel, Analytic sets and Borel isomorphisms, Fund. Math. 108 (1980), 83-88.

[57] A.H. Stone, Kernel constructions and Borel sets, Trans. Am. Math. Soc. 107 (1963), 58-70.

[58] F. Topsøe and J. Hoffman-Jørgensen, Analytic spaces and their applications, Analytic Sets, Instr. Conf. on Anal. Sets, Univ. Coll., London 1978, Academic Press, London (1980).

[59] W.W. Wadge, Degrees of complexity of subsets of the Baire space, Notices Am. Math. Soc. 1972, A-714.

[60] W.W. Wadge, Reducibility and determinateness on the Baire space, Thesis, Univ. of Calif. Berkeley (1984).

[61] R. Van Wesep, Wadge degrees and descriptive set theory, Cabal Seminar 76-77, Lect. Notes in Math. 689 (1978), 151-170.

[62] R. Van Wesep, Subsystems of second-order arithmetic, and descriptive

set theory under the axiom of determinateness, Thesis, Univ. of Calif. Berkeley (1978).

[63] P. Wolfe, The strict determinateness of certain games, Pac. J. Math. 5 (1955), 841-847.

C	1,11	ř	77
Q	1,17	$\Delta(\Gamma)$	77
Р	1,16	[A]	78
S	3,28	< (Wadge classes)	78
Τ	3,28	Sep $(D_n(\Sigma_{\mathcal{E}}^0),\Gamma)$	81
М	6	Bisep $(D_{\xi}, \Sigma_{\xi}^{0}), \Gamma, \Gamma')$	81
s	6	$SU(\Sigma_{\mathcal{F}}^{0},\Gamma)$	81
ν(s)	6	ΡU(Σ ⁰ _ε ,Γ)	81
s^i	6	$SD_{p}(\langle \Sigma_{F}^{0}, SU(\Sigma_{F}^{0}, \Gamma) \rangle, \Gamma')$	81
f(s)	6	$Bisep(\tilde{D}_{r}(\Sigma_{\xi}^{0}),\Gamma)$	82
s k	6	$SD_{r} (\langle \Sigma_{\mathcal{E}}^{0}, SU(\Sigma_{\mathcal{E}}^{0}, \Gamma) \rangle)$	82
ŝ	6	D	82
≤ (sequences)	6	Г	82
Σ_{α}^{0}	8	t(u)	83
πõ	8	ū	83
Δ_{α}^{0}	8	г _и (X)	91
D _η ((A _ζ : ζ < η))	23	Pu	93
$D_{\eta}^{X}(\Sigma_{\xi}^{0})$	23	۴ _u	93
$D_{\eta}(\Sigma_{\xi}^{0})$	23	yi u	93
S _n	27	Zu	93
$P_{n}^{(i)}$	28		
$\chi_n^{(i)}$	28		
P_{α}	29		
X ⁱ _α	29		

SPECIAL SYMBOLS

AUTHOR INDEX

P. Alexandroff, 1,2,3,11,15,16,19. R.D. Anderson, 33. R. Bennett, 36. L.E.J. Brouwer, 1,11. G. Cantor, 11. M. Davis, 75. E.K. van Douwen, 3,38,51,72,103,110. F. van Engelen, 23,37,51,74,103,112. R. Engelking, 5,9,13,41,102. R.C. Freiwald, 102. D. Gale, 74,75. J. de Groot, 110. A. Gutek, 30. L. Harrington, 103. F. Hausdorff, 16,23. J. Hoffman-Jørgensen, 38. W. Holsztyński, 102. W. Hurewicz, 37. H.J.K. Junnila, 103. A.S. Kechris, 74,78. B. Knaster, 30. K. Kunen, 104. K. Kuratowski, 3,8,9,10,14,23,110. M. Lavrentieff, 9,24. S. Levi, 10. A. Louveau, 3,23,73,81,82,83,84,86,89.

```
N. Luzin, 24,37,120.
```

D.A. Martin, 73,74,75,78. R. McDowell, 102. E.F. McHugh, Jr., 102. E. Michael, 14. J. van Mill, 3,4,21,22,32,33,37,51,72. A.W. Miller, 104,112. D. Montgomery, 14. Y. Moschovakis, 8,73,74,75,78. A.V. Ostrovskii, 30,32. J.B. Paris, 75. J. Pollard, 30. D. Ravdin, 30. M. Reichbach, 30. J.E. Rubin, 79. J. Saint-Raymond, 37,38. W. Sierpiński, 1,11,17,19,24,79,113. R. Sikorski, 74,102,104. M. Suslin, 13,37,43. J.R. Steel, 3,73,74,78,80,103,112. F.M. Stewart, 74,75. A.H. Stone, 14. F. Topsøe, 38. P. Urysohn, 1,2,3,11,15,16,19. W.W. Wadge, 73,74,77,78. R. Van Wesep, 74,78,79. R.J. Wille, 110. P. Wolfe, 75.

SUBJECT INDEX

Borel set		nowher
- absolute	8	
- exactly of additive class α	9	ξ
– exactly of ambiguous class α	9	reason
- exactly of multiplicative class	α 9	reduct
- irreducible	1	rigid
– of exact class α	9	Σ ⁰ ⊱set
Borel Wadge class	78	self-o
Cantor set	1,11	smal1
clopen	6	strate
complete	9	- winn
continuously closed	77	strong
AQ-act (in X)	0	strong
$\Delta \xi$ -set (in x)	8	strong
description	82	tvpe (
determinacy		USPE (
- axiom of	75	Wadge
everywhere properly [80	Wadge
		Wadge
game	74	Wadge
homogeneous	7	Wadge
- with respect to dense copies of A	A 7	Wadge-
Hurewicz-type theorem	37	zero-d
Knaster-Reichbach covering	30	
KR-covering	30	
	50	
Lipschitz-game	75,76	

nowhere P	6
Π^0_{ξ} -set (in X)	8
reasonably closed	80
reduction property	9
rigid	7
Σ^0_{ξ} -set (in X)	8
self-dual	77
small Borel class	23
strategy	74,75
- winning	75
strongly homogeneous	7
strongly σ -complete	25
strongly $\sigma-P$	38
type (of a description)	83
Wadge class	78
Wadge degree	77
Wadge game	76
Wadge lemma	76
Wadge ordering	78
Wadge-reducible	76
zero-dimensional	6

MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points. 1963.

2 A.R. Bloemena. Sampling from a graph. 1964. 3 G. de Leve. Generalized Markovian decision processes, part 1: model and method. 1964.

4 G. de Leve. Generalized Markovian decision processes, part II: probabilistic background. 1964.

5 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.

6 M.A. Maurice. Compact ordered spaces. 1964.

7 W.R. van Zwet. Convex transformations of random variables. 1964.

8 J.A. Zonneveld. Automatic numerical integration. 1964.

9 P.C. Baayen. Universal morphisms. 1964.

10 E.M. de Jager. Applications of distributions in mathematical physics. 1964. 11 A.B. Paalman-de Miranda. Topological semigroups. 1964.

12 J.A.Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. Formal properties of newspaper Dutch. 1965

13 H.A. Lauwerier. *Asymptotic expansions.* 1966, out of print; replaced by MCT 54.

14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

15 R. Doornbos. Slippage tests. 1966.

16 J.W. de Bakker. Formal definition of programming languages with an application to the definition of ALGOL 60. 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2, 1968.

19 J. van der Slot. Some properties related to compactness. 1968

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and topology. 1968.

22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part 1, 1968.

23 T.J. Dekker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968.

24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paerl. Representations of the Lorentz group and projec-tive geometry. 1969. tive geometry

26 European Meeting 1968. Selected statistical papers, part 1. 1968

27 European Meeting 1968. Selected statistical papers, part II. 1968

28 J. Oosterhoff. Combination of one-sided statistical tests. 1969.

29 J. Verhoeff. Error detecting decimal codes. 1969.

30 H. Brandt Corstius. Exercises in computational linguistics. 1970.

1970.
31 W. Molenaar. Approximations to the Poisson, binomial and hypergeometric distribution functions. 1970.
32 L. de Haan. On regular variation and its application to the weak convergence of sample extremes. 1970.

33 F.W. Steutel. Preservation of infinite divisibility under mix-ing and related topics. 1970.

34 I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal func-tions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971.

36 J. Grasman. On the birth of boundary. layers. 1971. 37 J.W. de Bakker, G.A. Blaauw, A.J.W. Duijvestijn, E.W. Dijkstra, P.J. van der Houwen, G.A.M. Kamsteeg-Kemper, F.E.J. Kruseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G. Zoutendijk. MC-25 Informatica Symposium. 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972. 39 H. Bavinck. Jacobi series and approximation. 1972.

40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972.

42 W. Vervaat. Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence. 1973.

45 A.A. Balkema. Monotone transformations and limit laws. 1973. 44 H. Bart. Meromorphic operator valued functions. 1973.

46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.

Jorman manipulation systems, part 1: ungauge: tribust formula manipulation systems, part 2: the compiler. 1973. 48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973.

50 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A. Sotter, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G. Fisker (eds.), Revised report on the algorithmic language ALGOL 68, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory, 1974.

52 P.C. Baayen (ed.). Topological structures. 1974.

53 M.J. Faber. Metrizability in generalized ordered spaces. 1974.

54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.

55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974.

56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory, foundations, partitions and combinatorial geometry. 1974.

57 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.

58 W. Albers. Asymptotic expansions and the deficiency concept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1975.

60 F. Göbel. Queueing models involving buffers. 1975. 63 J.W. de Bakker (ed.). Foundations of computer science.

1975 64 W.J. de Schipper. Symmetric closed categories. 1975.

65 J. de Vries. Topological transformation groups, 1: a categor-ical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

68 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977. 70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

71 J.A.E.E. van Nunen. Contracting Markov decision processes. 1976.

72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveguides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979. 74 H.J.J. te Riele. A theoretical and computational study of generalized aliquot sequences. 1976.

75 A.E. Brouwer. Treelike spaces and related connected topological spaces. 1977.

76 M. Rem. Associons and the closure statement. 1976

77 W.C.M. Kallenberg. Asymptotic optimality of likelihood ratio tests in exponential families. 1978.

78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.
79 M.C.A. van Zuijlen. Emperical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary problems. 1977

81 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science 11, part 1. 1976.

82 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science 11, part 2, 1976.

83 L.S. van Benthem Jutting. Checking Landau "Grundlagen" in the AUTOMATH system. 1979

84 H.L.L. Busard. The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii-xii, 1977.

85 J. van Mill. Supercompactness and Wallman spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program-ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977. 89 J.W. de Roever. Complex Fourier transformation and analytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen. Characterization of optimal strategies in dynamic games. 1981.

91 J.M. Geysel. Transcendence in fields of positive characteristic. 1979

92 P.J. Weeda. Finite generalized Markov programming. 1979. 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977.

94 A. Bijlsma. Simultaneous approximations in transcend number theory. 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978. 96 P.M.B. Vitányi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; functional equations and algorithms. 1984.

98 R. Geel. Singular perturbations of hyperbolic type. 1978. 99 J.K. Lenstra, A.H.G. Rinnooy Kan, P. van Emde Boas (eds.). Interfaces between computer science and operations research. 1978.

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceedings bicentennial congress of the Wiskundig Genootschap, part 1. 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 2, 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces.

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978.

104 J.M. van Wouwe. Go-spaces and generalizations of metri-zability. 1979. zability.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982.

106 A. Schrijver (ed.). Packing and covering in combinatorics. 1979.

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979.

108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979.

109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 2, 1979. 110 J.C. van Vliet. ALGOL 68 transput, part 1: historical review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part II: an implemen-tation model. 1979.

112 H.C.P. Berbee. Random walks with stationary increments and renewal theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979.

, 114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979. 115 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 1. 1979.

116 P.C. Baayen, J. van Mill (eds.). Topological structures 11, part 2. 1979.

117 P.J.M. Kallenberg. Branching processes with continuous state space. 1979.

118 P. Groeneboom. Large deviations and asymptotic efficiencies. 1980.

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980.

120 W.P.M. de Ruyter. On the asymptotic analysis of large-scale ocean circulation. 1980.

121 W.H. Haemers. Eigenvalue techniques in design and graph theory. 1980.

122 J.C.P. Bus. Numerical solution of systems of nonlinear equations. 1980.

123 I. Yuhász. Cardinal functions in topology - ten years later. 1880.

124 R.D. Gill. Censoring and stochastic integrals. 1980.

125 R. Eising. 2-D systems, an algebraic approach. 1980.

126 G. van der Hoek. Reduction methods in nonlinear pro-gramming. 1980.

127 J.W. Klop. Combinatory reduction systems. 1980. 128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. *ILP: intermediate language for pictures*.

1980

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.
132 H.M. Mulder. The interval function of a graph. 1980.

133 C.A.J. Klaassen. Statistical performance of location esti-

mators. 1981. 134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68. 1981.

135 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part I. 1981.

J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part 11. 1981.
 J. Telgen. Redundancy and linear programs. 1981.

138 H.A. Lauwerier. Mathematical models of epidemics. 1981. 139 J. van der Wal. Stochastic dynamic programming, succes sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis. 1981.

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981.

142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. 1981.

145 A.J. Brentjes. Multi-dimensional continued fraction algorithms. 1981.
146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.
141 ULT Strategie and factorization and factorization for the second strategies.

147 H.H. Tigelaar. Identification and informative sample size. 1982

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems. 1983.

Kovian Control progens. 1963.
 H49 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg,
 W.K. Vietsch (eds.). From A to Z, proceedings of a symposium in honour of A.C. Zaanen. 1982.
 H50 M. Veldhorst. An analysis of sparse matrix storage schemes (1982)

schemes. 1982.

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequences. 1982

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired ser-vices. 1982.

Vices. 1962.
154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.
155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.
155 P.M.G. Apper, Organization and deta allocation in

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two-dimensional smooth commutative formal groups over an alge-braically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983.

160 A. Rezus. Abstract AUTOMATH. 1983.

161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.
162 J.J. Dik. Tests for preference. 1983.

163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983. 164 F.A. van der Duyn Schouten. Markov decision processes with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983.

166 H.B.M. Jonkers. Abstraction, specification and implemen-tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program-ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.
CWI TRACTS

1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984.

1784.
2 J.J. Dijkstra. Fake topological Hilbert spaces and characterizations of dimension in terms of negligibility. 1984.
3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984.

systems, 1964.
4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984.
5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.
6 A.P.W. Böhm. Dataflow computation. 1984.

7 A. Blokhuis. Few-distance sets. 1984. 8 M.H. van Hoorn. Algorithms and approximations for queue-ing systems. 1984.

ing systems. 1984. 9 C.P.J. Koymans. Models of the lambda calculus. 1984. 10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.

11 N.M. van Dijk. Controlled Markov processes; time-discretization. 1984.

12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.

13 D. Grune. On the design of ALEPH. 1985.

13 D. Grune. On the design of ALEPH. 1985.
14 J.G.F. Thiemann. Analytic spaces and dynamic programming: a measure theoretic approach. 1985.
15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.
16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial differential operators: a case-study in Fourier integral operators. 1985.

17 H.M.M. ten Eikelder. Symmetries for dynamical and Ham-iltonian systems. 1985.

18 A.D.M. Kester. Some large deviation results in statistics. 1985.

19 T.M.V. Janssen. Foundations and applications of Montague grammer, part 1: Philosophy, framework, computer science. 1986.

20 B.F. Schriever. Order dependence. 1986.

21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986.

22 J.C.S.P. van der Woude. Topological dynamix. 1986.

23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution. 1986.
 24 J.C.M. Baeten. Filters and ultrafilters over definable subsets of admissible ordinats. 1986.

25 A.W.J. Kolen. Tree network and planar rectilinear location theory, 1986.

Interfy, 1960.
26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.
27 A.J.M. van Engelen. Homogeneous zero-dimensional abso-lute Borel sets. 1986.

.